%%%@@@ input from paper.tex
\documentclass[12pt,reqno]{amsart}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Papersize and margins:                                                      %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\textheight23cm
\textwidth16cm
\hoffset-2truecm
\voffset-0.5truecm

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Macros for drawing:                                                         %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%%%@@@ input from texdraw
% TeXdraw macros

% $Id: texdraw.tex,v 2.3 1995/12/19 texdraw-V2R0 $

%   Copyright (C) 1991-1995  Peter Kabal

% The TeXdraw routines in this file are provided free of charge without
% warranty of any kind.  Note that the TeXdraw routines are copyrighted.
% They may be distributed freely provided that the recipients also
% acquire the right to distribute them freely.  The notices to this
% effect must be preserved when the files are distributed.

%  Peter Kabal
%  Department of Electrical Engineering
%  McGill University
%  3480 University
%  Montreal, Quebec
%  Canada  H3A 2A7

%  kabal@TSP.EE.McGill.CA
 
\def\setRevDate $#1 #2 #3${#2}
\def\TeXdrawId{\setRevDate $Date: 1995/12/19 16:40:42 $ TeXdraw V2R0}

% ===============================================================

% The TeXdraw macros allow PostScript line drawings and such to be
% generated from within TeX and LaTeX.
% (1) TeXdraw allows TeX text (either horizontal or rotated) to be
%     superimposed on the figure.
% (2) TeXdraw implements a \bsegment-\esegment environment which allows
%     parameter changes and coordinate changes to be kept local.  In
%     effect, these segments are self-contained relocatable procedures.
% (3) TeX's macro facility can be used to modularize drawing units and
%     produce more complex entities from simple elements.
% (4) The drawing can be positioned on the page like any other TeX box.

% TeXdraw coordinate units have positive X to the right and positive Y up.
% The drawing units can be selected (initially inches).  In addition, two
% scaling parameters - unit scale and segment scale - are available.  Their
% effects are multiplicative.

% Segments allow for relocatable drawing units.  Inside each segment the
% coordinates are relative to the initial point, which becomes (0 0).
% Scaling is local to segments.  Each segment inherits the unit scale
% scaling from outside the segment, but any changes apply to that segment
% and inferior segments.  The segment scale factor is reset to unity on
% entry to each segment.

% The coordinates given as command arguments are used to determine
% the size of the drawing.  The width of the plot line, sizes of
% arrowheads, arcs or text do not affect the size of the drawing.

% TeXdraw writes PostScript commands to an intermediate file.  After the
% drawing is finished, the PostScript file is included in the document as a
% PostSript include file.

%  This file is divided into 4 parts,
%  - TeXdraw user interface
%  - Utility definitions
%  - Low level definitions
%  - PostScript file macros


\chardef\catamp=\the\catcode`\@
\catcode`\@=11

% ===============================================================
% ===== TeXdraw user interface ==================================
\long                              % \centertexdraw needs to be very \long
\def\centertexdraw #1{\hbox to \hsize{\hss
                                      \btexdraw #1\etexdraw
                                      \hss}}

% ====== Begin TeXdraw
% Inside the texdraw box:
%   The \vbox should be of zero size; none of the TeXdraw drawing commands
%   generate text, the TeXdraw text commands generate zero size boxes.

\def\btexdraw {\x@pix=0             \y@pix=0
               \x@segoffpix=\x@pix  \y@segoffpix=\y@pix
% Set the default values (define outside of the group so that \etexdraw can
% see the scaling parameters)
               \t@exdrawdef
               \setbox\t@xdbox=\vbox\bgroup\offinterlineskip
                   \global\d@bs=0           % pending segments
                   \global\t@extonlytrue    % no PS commands yet
                   \global\p@osinitfalse
                   \s@avemove \x@pix \y@pix % capture the initial position
                   \m@pendingfalse
                   \global\p@osinitfalse    % capture the next move
                   \p@athfalse
                   \the\everytexdraw}

% ====== End TeXdraw
% Write out a trailer, close the file, bring in the PostScript code as
% a \special include file.  The \special is offset to position on the page
% to be (llx,ury) in PostScript coordinates.

% The drawing is placed in a \vbox of appropriate size (zero depth).  The
% temporary PostScript file is superimposed with offsets such that the
% lower lefthand corner of the drawing is aligned with the lower lefthand of
% the box.  Then, the position is temporarily offset to that corresponding
% to (0 0) to place the TeX text.  Note that TeX text goes on top of the
% PostScript drawing.
\def\etexdraw {\ift@extonly \else
                 \t@drclose      % close the PostScript file
               \fi
               \egroup           % ends the \vbox \bgroup
               \ifdim \wd\t@xdbox>0pt
                 \t@xderror {TeXdraw box non-zero size,
                             possible extraneous text}%
               \fi
               \vbox {\offinterlineskip
                      \pixtobp \xminpix \l@lxbp  \pixtobp \yminpix \l@lybp
                      \pixtobp \xmaxpix \u@rxbp  \pixtobp \ymaxpix \u@rybp
                      \hbox{\t@xdinclude 
                        [{\l@lxbp},{\l@lybp}][{\u@rxbp},{\u@rybp}]{\p@sfile}}%
                      \pixtodim \xminpix \t@xpos  \pixtodim \yminpix \t@ypos
                      \kern \t@ypos
                      \hbox {\kern -\t@xpos
                             \box\t@xdbox       % TeX text
                             \kern \t@xpos}%
                      \kern -\t@ypos\relax}}

% ===== Drawing scaling
% The units in any segment may be scaled arbitrarily.  A unit scale is
% local to a segment but affects enclosed segments unless specifically
% overridden in that segment.  In addition there is a segment scale.  The
% overall scale is the product of the two scaling factors.
%
% Scaling is handled entirely on the TeX side, the PostScript side gets
% absolute pixel coordinates.

% Drawing units, e.g. "in" or "cm"
\def\drawdim #1 {\def\d@dim{#1\relax}}

% \u@nitsc - unit scale
% \s@egsc  - segment scale
% \d@sc    - drawing scale, product of the unit scale and segment scale

% Note that successive application of relative scale factors can lead
% to poor accuracy of the final scale factor.  Each scale factor is
% represented to about 5 decimal digits after the decimal point.
\def\setunitscale #1 {\edef\u@nitsc{#1}%
                      \realmult \u@nitsc \s@egsc \d@sc}
\def\relunitscale #1 {\realmult {#1}\u@nitsc \u@nitsc
                      \realmult \u@nitsc \s@egsc \d@sc}
\def\setsegscale #1 {\edef\s@egsc {#1}%
                     \realmult \u@nitsc \s@egsc \d@sc}
\def\relsegscale #1 {\realmult {#1}\s@egsc \s@egsc
                     \realmult \u@nitsc \s@egsc \d@sc}

% ===== Drawing segments
% The position is restored after a segment.
% Segments use TeX grouping on the TeX side and gsave/grestore on the
% PostScript side to keep changes local.  On the TeX side segments have
% (0 0) as the initial point, while the PostScript side sees no scale
% changes or translations.
\def\bsegment {\ifp@ath
                 \f@lushbs
                 \f@lushmove
               \fi
               \begingroup
               \x@segoffpix=\x@pix
               \y@segoffpix=\y@pix
               \setsegscale 1
               \global\advance \d@bs by 1\relax}
\def\esegment {\endgroup
               \ifnum \d@bs=0
                 \writetx {es}%
               \else
                 \global\advance \d@bs by -1
               \fi}

% Save a position
% Save each coordinate as the macro "*name".  The macro is defined to
% be the pixel coordinate value.
\def\savecurrpos (#1 #2){\getsympos (#1 #2)\a@rgx\a@rgy
                         \s@etcsn \a@rgx {\the\x@pix}%
                         \s@etcsn \a@rgy {\the\y@pix}}
\def\savepos (#1 #2)(#3 #4){\getpos (#1 #2)\a@rgx\a@rgy
                            \coordtopix \a@rgx \t@pixa
                            \advance \t@pixa by \x@segoffpix
                            \coordtopix \a@rgy \t@pixb
                            \advance \t@pixb by \y@segoffpix
                            \getsympos (#3 #4)\a@rgx\a@rgy
                            \s@etcsn \a@rgx {\the\t@pixa}%
                            \s@etcsn \a@rgy {\the\t@pixb}}

% ===== Line parameters
% The following parameters apply to subsequent lines.  Each of these
% commands invokes a stroke to draw the previous line segments,
% establishes the current point and then changes the line parameter.
% The parameters are kept local by the PostScript gsave/grestore
% mechanism.  We use \writetx here, instead of \writeps, since we
% do not want to flush any moves.
\def\linewd #1 {\coordtopix {#1}\t@pixa
                \f@lushbs
                \writetx {\the\t@pixa\space sl}}
\def\setgray #1 {\f@lushbs
                 \writetx {#1 sg}}
\def\lpatt (#1){\listtopix (#1)\p@ixlist
                \f@lushbs
                \writetx {[\p@ixlist] sd}}

% ===== Line drawing
% PostScript uses the concept of a path consisting of line segments.
% In this interface to PostScript, paths are continuous across the
% beginning of segments.  Paths terminate at the end of a segment with
% an implicit move.  In addition, paths are both terminated and started
% with a move.  There is a current point at all times, starting with
% initial position (0,0).
\def\lvec (#1 #2){\getpos (#1 #2)\a@rgx\a@rgy
                  \s@etpospix \a@rgx \a@rgy
                  \writeps {\the\x@pix\space \the\y@pix\space lv}}
\def\rlvec (#1 #2){\getpos (#1 #2)\a@rgx\a@rgy
                   \r@elpospix \a@rgx \a@rgy
                   \writeps {\the\x@pix\space \the\y@pix\space lv}}
\def\move (#1 #2){\getpos (#1 #2)\a@rgx\a@rgy
                  \s@etpospix \a@rgx \a@rgy
                  \s@avemove \x@pix \y@pix}
\def\rmove (#1 #2){\getpos (#1 #2)\a@rgx\a@rgy
                   \r@elpospix \a@rgx \a@rgy
                   \s@avemove \x@pix \y@pix}

% ===== Circles, ellipses and arcs
% Note that arcs do not update the size of the drawing.
% \lcir, stroked circle
%   r:#1 -  radius
% \fcir, filled circle
%   f:#1 - fill gray level, 0 is black, 1 is white
%   r:#2 - radius
% \ellip, stroked ellipse
%   rx:#1 - x radius
%   ry:#2 - y radius
% \fellip, filled ellipse
%   f:#1 - fill gray level, 0 is black, 1 is white
%   rx:#1 - x radius
%   ry:#2 - y radius
% \larc, stroked counterclockwise arc, with the present position being
%        the center of the arc.  Only the arc is drawn (not the line
%        joining the center to the beginning of the arc)
%   r:#1  - radius
%   sd:#2 - start angle (degrees)
%   ed:#3 - end angle (degrees)
\def\lcir r:#1 {\coordtopix {#1}\t@pixa
                \writeps {\the\t@pixa\space cr}%
                \r@elupd \t@pixa \t@pixa
                \r@elupd {-\t@pixa}{-\t@pixa}}
\def\fcir f:#1 r:#2 {\coordtopix {#2}\t@pixa
                     \writeps {\the\t@pixa\space #1 fc}%
                     \r@elupd \t@pixa \t@pixa
                     \r@elupd {-\t@pixa}{-\t@pixa}}
\def\lellip rx:#1 ry:#2 {\coordtopix {#1}\t@pixa
                     \coordtopix {#2}\t@pixb
                     \writeps {\the\t@pixa\space \the\t@pixb\space el}%
                     \r@elupd \t@pixa \t@pixb
                     \r@elupd {-\t@pixa}{-\t@pixb}}
\def\fellip f:#1 rx:#2 ry:#3 {\coordtopix {#2}\t@pixa
                     \coordtopix {#3}\t@pixb
                     \writeps {\the\t@pixa\space \the\t@pixb\space #1 fe}%
                     \r@elupd \t@pixa \t@pixb
                     \r@elupd {-\t@pixa}{-\t@pixb}}
\def\larc r:#1 sd:#2 ed:#3 {\coordtopix {#1}\t@pixa
                            \writeps {\the\t@pixa\space #2 #3 ar}}

% ===== Fill commands
% The form here completes a path with a closepath, applies the fill,
% starts a newpath and moves to the current point.  The gray level has
% 0 as black and 1 as white.  The current path is terminated.

\def\ifill f:#1 {\writeps {#1 fl}}     % Fill only
\def\lfill f:#1 {\writeps {#1 fp}}     % Stroke and fill

% ===== Text
% TeX text superimposed on the drawing
%  \htext (x y){text} or \htext {text}
%  \vtext (x y){text} or \vtext {text}
%  \rtext td:angle (x y){text} or \rtext td:angle {text}
%  \textref h:#1 v:#2

% The TeX text (or whatever) is placed in an \hbox.  The box is rotated
% for vertical text.  The text is placed on the drawing at the specified
% location (coordinates specified) or the current location (coordinates
% not specified).  The text reference point is placed at that location.
% For the purposes of determining the drawing size, the text box is of
% zero size.

% Horizontal text
% Check if the argument starts with a left parenthesis
\def\htext #1{\def\testit {#1}%
              \ifx \testit\l@paren
                \let\next=\h@move
              \else
                \let\next=\h@text
              \fi
              \next {#1}}

% Rotated text
\def\rtext td:#1 #2{\def\testit {#2}%
                    \ifx \testit\l@paren
                      \let\next=\r@move
                    \else
                      \let\next=\r@text
                    \fi
                    \next td:#1 {#2}}

% Vertical text
\def\vtext {\rtext td:90 }

% Text reference point
%  h:#1  text horizontal reference point - L, C or R
%  v:#2  text vertical reference point - T, C or B
\def\textref h:#1 v:#2 {\ifx #1R%
                          \edef\l@stuff {\hss}\edef\r@stuff {}%
                        \else
                          \ifx #1C%
                            \edef\l@stuff {\hss}\edef\r@stuff {\hss}%
                          \else  % default L
                            \edef\l@stuff {}\edef\r@stuff {\hss}%
                          \fi
                        \fi
                        \ifx #2T%
                          \edef\t@stuff {}\edef\b@stuff {\vss}%
                        \else
                          \ifx #2C%
                            \edef\t@stuff {\vss}\edef\b@stuff {\vss}%
                          \else  % default B
                            \edef\t@stuff {\vss}\edef\b@stuff {}%
                          \fi
                        \fi}

% ===== Arrow vectors
\def\avec (#1 #2){\getpos (#1 #2)\a@rgx\a@rgy
                  \s@etpospix \a@rgx \a@rgy
                  \writeps {\the\x@pix\space \the\y@pix\space (\a@type) %
                            \the\a@lenpix\space \the\a@widpix\space av}}

\def\ravec (#1 #2){\getpos (#1 #2)\a@rgx\a@rgy
                   \r@elpospix \a@rgx \a@rgy
                   \writeps {\the\x@pix\space \the\y@pix\space (\a@type) %
                             \the\a@lenpix\space \the\a@widpix\space av}}

% Arrowhead size
%  l:#1 - length of the arrowhead
%  w:#2 - width of the base of the arrowhead
\def\arrowheadsize l:#1 w:#2 {\coordtopix{#1}\a@lenpix
                              \coordtopix{#2}\a@widpix}
% Arrowhead type
%  t:#1 - arrowhead type, F  filled triangle (using current gray level)
%                         T  empty closed triangle
%                         W  white filled triangle
%                         V  Vee shape, at the end of the vector
%                         H  (or other character) Vee shape, vector stops
%                            short of the Vee
\def\arrowheadtype t:#1 {\edef\a@type{#1}}

% ===== Bezier curve
% The initial point is assumed to be the current point.  Only the last
% coordinate affects the size of the plot.
\def\clvec (#1 #2)(#3 #4)(#5 #6)%
           {\getpos (#1 #2)\a@rgx\a@rgy
            \coordtopix \a@rgx\t@pixa
            \advance \t@pixa by \x@segoffpix
            \coordtopix \a@rgy\t@pixb
            \advance \t@pixb by \y@segoffpix
            \getpos (#3 #4)\a@rgx\a@rgy
            \coordtopix \a@rgx\t@pixc
            \advance \t@pixc by \x@segoffpix
            \coordtopix \a@rgy\t@pixd
            \advance \t@pixd by \y@segoffpix
            \getpos (#5 #6)\a@rgx\a@rgy
            \s@etpospix \a@rgx \a@rgy
            \writeps {\the\t@pixa\space \the\t@pixb\space
                      \the\t@pixc\space \the\t@pixd\space
                      \the\x@pix\space \the\y@pix\space cv}}

% ===== Draw the bounding box
\def\drawbb {\bsegment
               \drawdim bp
               \linewd 0.24       % line width 1/300 inch = 0.24 bp
               \setunitscale {\p@sfactor}
               \writeps {\the\xminpix\space \the\yminpix\space mv}%
               \writeps {\the\xminpix\space \the\ymaxpix\space lv}%
               \writeps {\the\xmaxpix\space \the\ymaxpix\space lv}%
               \writeps {\the\xmaxpix\space \the\yminpix\space lv}%
               \writeps {\the\xminpix\space \the\yminpix\space lv}%
             \esegment}


% ===============================================================
% ===== Utility macros used by TeXdraw ==========================

% ===== Decode coordinates
% Get coordinates
% This macro is used to get two arguments separated by a blank, with
% possible leading and trailing blanks.  Symbolic coordinates are
% converted to user coordinates.
%  (#1 #2) - coordinates
%  #3 - macro name to receive the x-coordinate value
%  #4 - macro name to receive the y-coordinate value
\def\getpos (#1 #2)#3#4{\g@etargxy #1 #2 {} \\#3#4%
                        \c@heckast #3%
                        \ifa@st
                          \g@etsympix #3\t@pixa
                          \advance \t@pixa by -\x@segoffpix
                          \pixtocoord \t@pixa #3%
                        \fi
                        \c@heckast #4%
                        \ifa@st
                          \g@etsympix #4\t@pixa
                          \advance \t@pixa by -\y@segoffpix
                          \pixtocoord \t@pixa #4%
                        \fi}

% Get symbolic coordinate names
%  (#1 #2) - symbolic coordinates
%  #3 - macro name to receive the symbolic x coordinate name
%  #4 - macro name to receive the symbolic y coordinate name
\def\getsympos (#1 #2)#3#4{\g@etargxy #1 #2 {} \\#3#4%
                     \c@heckast #3%
                     \ifa@st \else
                       \t@xderror {TeXdraw: invalid symbolic coordinate}%
                     \fi
                     \c@heckast #4%
                     \ifa@st \else
                       \t@xderror {TeXdraw: invalid symbolic coordinate}%
                     \fi}

% ===== Convert a list of values to pixel values
%  (#1) - blank separated list of values in user coordinates
%  #2 - macro name to receive the blank separated list of pixel values
\def\listtopix (#1)#2{\def #2{}%
                      \edef\l@ist {#1 }%    % append a blank to the string
                      \m@oretrue
                      \loop
                        \expandafter\g@etitem \l@ist \\\a@rgx\l@ist
                        \a@pppix \a@rgx #2%
                        \ifx \l@ist\empty
                          \m@orefalse
                        \fi
                      \ifm@ore
                      \repeat}

% ===== Real multiplication
% This function uses the property that a box dimension may be scaled by
% a real value.  The values are converted to dimensions in units of pt.
% This choice gives us a reasonable dynamic range.  The final step is to
% clean off the "pt" on the resulting dimension.  Note that these are fixed
% point operations with each operand represented to an accuracy of about 5
% decimal places.

% Note we must use magnified points not "true" points, since the answer is
% expressed in magnified points.  The result will be calculated in the same
% manner no matter what the magnification is.
%  #1 and #2 are multiplicands
%  #3 macro name to capture the real result
\def\realmult #1#2#3{\dimen0=#1pt
                     \dimen2=#2\dimen0
                     \edef #3{\expandafter\c@lean\the\dimen2}}

% ===== Divide integers, real result
%  #1 integer numerator value
%  #2 integer denominator (divisor) value
%  #3 macro name to capture the real result
\def\intdiv #1#2#3{\t@counta=#1
                   \t@countb=#2
%  Limitations: #1 must be negatable, i.e. it must not be the largest
%                  magnitude negative number
%               #2 must be able to be multiplied by 2 without overflow
%  Calculate a*65536/b  where the factor 65536 converts from pt to sp.
%  This operation can also be interpretated as an extended precision
%  numerator divided by the denominator.  The scheme used is basically a
%  long division, except that it is bootstrapped by an integer divide.
%  The computations are carried out with positive numerator and
%  denominator, with the appropriate restoration of sign at the end.
%    \t@counta == remainder, r, initially set to a
%    \t@countb == denominator, b
%    \t@countc == quotient, q
%    \t@countd == +1, a and b have the same sign
%                 -1, a and b have opposite signs
%    \t@counte == temporary register
                   \ifnum \t@countb<0
                      \t@counta=-\t@counta
                      \t@countb=-\t@countb
                   \fi
                   \t@countd=1                    % record the sign
                   \ifnum \t@counta<0
                      \t@counta=-\t@counta
                      \t@countd=-1
                   \fi
%                                                 % q=a/b, r=a-q*b
                   \t@countc=\t@counta  \divide \t@countc by \t@countb
                   \t@counte=\t@countc  \multiply \t@counte by \t@countb
                   \advance \t@counta by -\t@counte
                   \t@counte=-1
                   \loop
                     \advance \t@counte by 1
                   \ifnum \t@counte<16                  % loop 16 times
                       \multiply \t@countc by 2           % q=2q
                       \multiply \t@counta by 2           % r=2r
                       \ifnum \t@counta<\t@countb \else   % if ( r >= b )
                         \advance \t@countc by 1          %   q=q+1
                         \advance \t@counta by -\t@countb %   r=r-b
                       \fi
                   \repeat
                   \divide \t@countb by 2         % rounding
                   \ifnum \t@counta<\t@countb     % if ( r >= b/2 ) q=q+1
                     \advance \t@countc by 1
                   \fi
                   \ifnum \t@countd<0             % restore the sign
                     \t@countc=-\t@countc
                   \fi
                   \dimen0=\t@countc sp           % express as a dimension
                   \edef #3{\expandafter\c@lean\the\dimen0}}

% ===============================================================
% ===== Internal TeXdraw macros =================================

% ===== Macros for converting between dimensions and units
% Convert drawing units (coordinate value, scaled by the unit scale and
% segment scale) to pixels.  We use rounding to get more accurate results.
%  #1 dimension in drawing units
%  #2 count in pixels (returned into a count)
\def\coordtopix #1#2{\dimen0=#1\d@dim
                     \dimen2=\d@sc\dimen0
                     \t@counta=\dimen2          % scaled dimension in sp
                     \t@countb=\s@ppix
                     \divide \t@countb by 2
                     \ifnum \t@counta<0         % rounding
                       \advance \t@counta by -\t@countb
                     \else
                       \advance \t@counta by \t@countb
                     \fi
                     \divide \t@counta by \s@ppix
                     #2=\t@counta}

% Convert from absolute pixels to relative scaled coordinates
%  #1 - input integer pixel value
%  #2 - macro name to receive the character string corresponding to the
%       floating point coordinate value
\def\pixtocoord #1#2{\t@counta=#1%
                     \multiply \t@counta by \s@ppix
                     \dimen0=\d@sc\d@dim
                     \t@countb=\dimen0
                     \intdiv \t@counta \t@countb #2}

% Convert pixels to TeX dimensions.
%  #1 - input integer pixel value
%  #2 - returned dimension (returned into a dimension register)
\def\pixtodim #1#2{\t@countb=#1%
                   \multiply \t@countb by \s@ppix
                   #2=\t@countb sp\relax}

% Convert pixels to (integer) bp units
%  #1 - input pixel value
%  #2 - integer value, returned as a macro definition
\def\pixtobp #1#2{\dimen0=\p@sfactor pt
                  \t@counta=\dimen0
                  \multiply \t@counta by #1%
                  \ifnum \t@counta < 0             % rounding
                    \advance \t@counta by -32768
                  \else
                    \advance \t@counta by 32768
                  \fi
                  \divide \t@counta by 65536
                  \edef #2{\the\t@counta}}
                  
% ===== Allocations for registers and counts
% == Temporary count registers
\newcount\t@counta    \newcount\t@countb   % Use at lowest levels
\newcount\t@countc    \newcount\t@countd
\newcount\t@counte
\newcount\t@pixa      \newcount\t@pixb     % Use for pixel values
\newcount\t@pixc      \newcount\t@pixd

% == Temporary dimension registers
\newdimen\t@xpos      \newdimen\t@ypos

% == Position and parameter registers
% The minimum and maximum extent in the X and Y direction in pixel units
% (updated globally to reach outside segments)
\newcount\xminpix      \newcount\xmaxpix
\newcount\yminpix      \newcount\ymaxpix

% == Arrowhead parameters
\newcount\a@lenpix     \newcount\a@widpix

% == Absolute pixel positions
\newcount\x@pix        \newcount\y@pix
\newcount\x@segoffpix  \newcount\y@segoffpix
\newcount\x@savepix    \newcount\y@savepix

% == Conversion factor
\newcount\s@ppix       % sp/pixel 

% == Pending segments count
\newcount\d@bs

% == Counter to form unique file names
\newcount\t@xdnum
\global\t@xdnum=0

% == TeXdraw box
\newbox\t@xdbox

% == Output stream number for the PostScript file
\newwrite\drawfile

% == \newif
\newif\ifm@pending
\newif\ifp@ath
\newif\ifa@st
\newif\ifm@ore
\newif \ift@extonly
\newif\ifp@osinit

% == \newtoks
\newtoks\everytexdraw

% ===== Character definitions
\def\l@paren{(}
\def\a@st{*}

% ===== Special character macros
% Need to be able to insert "%", "{" and "}" characters into the
% PostScript file.
% Define macros which have these characters with category "other".
% We will assume, that these characters have the standard meanings --
% after all, we use comments and braces in this code.
\catcode`\%=12
  \def\p@b {%!}  \def\p@p {%%}
\catcode`\%=14
\catcode`\{=12  \catcode`\}=12  \catcode`\u=1 \catcode`\v=2
  \def\l@br u{v  \def\r@br u}v
\catcode `\{=1  \catcode`\}=2   \catcode`\u=11 \catcode`\v=11

% ===== Pixel conversion factors
% The position is kept as an integer value (count).  It is set to a
% resolution corresponding to 300 units/inch.  We refer to them as pixels,
% but in fact the resolution is just that: movements are quantized to lie
% on a grid with that resolution.

% Using pixel units which correspond to the actual resolution of the device
% has advantages in that all horizontal and vertical lines then will be
% drawn with the same line thickness.  In addition the coordinates are
% then integer values (no decimal point or leading zeros) which leads
% to a more compact PostScript file.

% The following macro sets the conversion from PostScript units (bp) to the
% integer units (pixels).  The file inclusion \special environment in the
% PostScript driver restores the context to default PostScript values
% (bp or 72/in and origin in the lower lefthand corner).  A scaling value
% of 0.24 converts to 300/inch.  Note that the PostScript commands written
% to the temporary PostScript file do not depend on the TeX magnification
% in effect.  Magnification should be handled by the dvi to PostScript
% driver at the time that the file is included in the output.

% Calculate the conversion factors
% Let s@ppix = sp/pixel = u / p, where u = sp/unit and p = pix/unit (both
% integer values).
% We calculate s@ppix as
%         s@ppix = [ (u+0.5p)/p ]
% We also calculate the PostScript scale factor bp/pixel
% Let b = sp/bp.  We want p@sfactor = s@ppix/b.  For 300 pixels/inch, this
% gives p@sfactor=0.24.  Using rounding
%         p@sfactor = [ (s@ppix+0.5b)/b ] .
% To carry out the arithmetic, we will operate in sp units (integers) and
% generate the answer in pt units (multiplying by sp/pt).  This result will
% expressed as a character string representing a real number after the "pt"
% designator is stripped off.

{\catcode`\p=12 \catcode`\t=12
 \gdef\c@lean #1pt{#1}}

\def\sppix#1/#2 {\dimen0=1#2 \s@ppix=\dimen0
                 \t@counta=#1%
                 \divide \t@counta by 2
                 \advance \s@ppix by \t@counta
                 \divide \s@ppix by #1%             % \s@ppix available
                 \t@counta=\s@ppix
                 \multiply \t@counta by 65536       % 1 pt = 65536 sp
                 \advance \t@counta by 32891        % 0.5 bp = 32890.88 sp
                 \divide \t@counta by 65782         % 1 bp = 65781.76 sp
                 \dimen0=\t@counta sp
                 \edef\p@sfactor {\expandafter\c@lean\the\dimen0}}

% ===== Low level coordinate decoding macros
% Get two values, separated by a blank
% Invoke as \g@etargxy <stuff> {} \\\ma\mb
\def\g@etargxy #1 #2 #3 #4\\#5#6{\def #5{#1}%
                           \ifx #5\empty
                             \g@etargxy #2 #3 #4 \\#5#6%  leading blank
                           \else
                             \def #6{#2}%
                             \def\next {#3}%
                             \ifx \next\empty \else
                               \t@xderror {TeXdraw: invalid coordinate}%
                             \fi
                           \fi}

% Check for a leading asterisk
% Sets \a@stfalse or \a@sttrue, test with \ifa@st
\def\c@heckast #1{\expandafter
                  \c@heckastll #1\\}
\def\c@heckastll #1#2\\{\def\testit {#1}%
                        \ifx \testit\a@st
                          \a@sttrue
                        \else
                          \a@stfalse
                        \fi}

% Decode a symbolic coordinate
% Pixel value returned to a count
\def\g@etsympix #1#2{\expandafter
               \ifx \csname #1\endcsname \relax
                 \t@xderror {TeXdraw: undefined symbolic coordinate}%
               \fi
               #2=\csname #1\endcsname}

% Set a macro named #1 to have value #2
\def\s@etcsn #1#2{\expandafter
                  \xdef\csname#1\endcsname {#2}}

% ===== Low level list decoding macros
% Pick off the first item -> #3, rest of string -> #4
\def\g@etitem #1 #2\\#3#4{\edef #4{#2}\edef #3{#1}}
\def\a@pppix #1#2{\edef\next {#1}%
                  \ifx \next\empty \else
                    \coordtopix {#1}\t@pixa
                    \ifx #2\empty
                      \edef #2{\the\t@pixa}%
                    \else
                      \edef #2{#2 \the\t@pixa}%
                    \fi
                  \fi}

% ===== Macros for updating the position
% Calculate the position in pixels and update the maximum excursions
\def\s@etpospix #1#2{\coordtopix {#1}\x@pix
                     \advance \x@pix by \x@segoffpix
                     \coordtopix {#2}\y@pix
                     \advance \y@pix by \y@segoffpix
                     \u@pdateminmax \x@pix \y@pix}

\def\r@elpospix #1#2{\coordtopix {#1}\t@pixa
                     \advance \x@pix by \t@pixa
                     \coordtopix {#2}\t@pixa
                     \advance \y@pix by \t@pixa
                     \u@pdateminmax \x@pix \y@pix}

\def\r@elupd #1#2{\t@counta=\x@pix
                  \advance\t@counta by #1%
                  \t@countb=\y@pix
                  \advance\t@countb by #2%
                  \u@pdateminmax \t@counta \t@countb}

\def\u@pdateminmax #1#2{\ifnum #1>\xmaxpix
                          \global\xmaxpix=#1%
                        \fi
                        \ifnum #1<\xminpix
                          \global\xminpix=#1%
                        \fi
                        \ifnum #2>\ymaxpix
                          \global\ymaxpix=#2%
                        \fi
                        \ifnum #2<\yminpix
                          \global\yminpix=#2%
                        \fi}

% =====  Save moves / flush moves
% A TeXdraw segment which generates only TeX text uses only move, begin
% segment and end segment commands.  The goal is to avoid writing out
% empty segments for such cases.  To this end, moves are held back and
% only written out if necessary to set the position or terminate a path.
% Also in this way, a TeXdraw drawing which generates only TeX text will
% not generate a PostScript file.

% Two flags are used.  Both flags are local to a segment.
% - move pending:  Set when a move has been invoked but the move command
%   has not been written out to the PostScript file
% - path in progress:  Set when a PostScript path has been started but the
%   path has not been terminated and stroked.
% (1) Moves are kept back.  Using TeX's groups, a local flag and local
%     position registers are used to keep track of whether the latest
%     move applicable to a given segment has been written out or not.  In
%     effect there is a stack of pending moves, one for each level of
%     segment nesting.
% (2) At the beginning of a segment, if a PS path is in progress and a
%     a move is pending, the move is written out, terminating the path
%     and stroking the path.  This is done to ensure that the path is
%     stroked before lines and/or fills are executed in the segment.
% (3) At the beginning of a segment, if a PS path is not in progress,
%     any pending moves are kept back.  Effectively, the move will be
%     transferred into the segment.  It will be written out only when
%     the position needs to be updated for some other command.  Such
%     moves which are transferred into segments may have to be repeated
%     outside the segment.  The move pending flag will be restored to
%     the value outside the segment on exit from the segment.
% (4) A begin segment command is not written out, but instead a global
%     segment backlog counter is incremented.  The backlog of begin
%     segment commands is written out when a drawing command is
%     encountered.
% The effect of the above on the TeXdraw commands is as follows.
% (a) move:
%     - set the current position
%     - record the position of the saved move
%     - set the move pending flag
% (b) begin segment:
%     - if a path is in progress
%         - if a move is pending
%           - if there is a backlog of segments
%             - write out enough begin segments to clear the backlog
%           - write out the pending move
%           - reset the move pending flag (local to the containing segment,
%             but affects inferior segments)
%           - reset the path in progress flag
%     - increment the segment backlog counter
%     - begin a group
% (c) end segment:
%     - end a group
%     - if there is no backlog of segments
%       - write an end segment command
%     - if there is a backlog of segments
%       - decrement the backlog counter, thereby omitting an empty
%         empty segment.
%     - the move pending flag and path in progress flag are automatically
%       restored on leaving the TeX group
% (d) text:
%     - create a text box
% (e) line parameters:
%     - if there is a backlog of segments
%        - write out enough begin segments to clear the backlog
%     - clear the path in progress flag
%     - write the PS command changing the line parameter
% (f) other drawing commands:
%     - update the current position
%     - if there is a backlog of segments
%        - write out enough begin segments to clear the backlog
%     - if there is a pending move
%        - write out the pending move
%        - reset the move pending flag (local to this segment, but affects
%          inferior segments)
%     - set the path in progress flag
%     - write the drawing command
% Notes:
% (1) The es PS command strokes the path at the end of a segment to
%     ensure that the correct line parameters are used for the segment.
%     The path before the corresponding bs command is restored and
%     continued.
% (2) The \f@lushbs and \f@lushmove commands must be invoked before each
%     drawing command written to the PS file.  The macro \writeps includes
%     these operations.

% Another awkward business has to do with initialization.  We want a
% default (0 0) initial position so that the user can draw vectors
% immediately.  However, if the user specifies another move before
% beginning to draw, that position should be the initial position.  The
% importance of this initial position is that the determination of the
% maximum excursion must take this value into account.  We handle the
% initialization in the \s@avemove and \f@lushmove macros.  The macro
% \ifp@osinit indicates whether the next move should be captured as the
% initial values for \xminpix, \yminpix, \xmaxpix, and \ymaxpix.  However,
% if a \f@lushmove is invoked, then we assume that the appropriate initial
% values has already been set.  The "mv" command in PostScript is defined
% to stroke the current path (if any) and move to the pixel coordinates
% specified.

% Note that \m@pendingtrue and \m@pendingfalse define the flag locally.
% In addition, \x@savepix and \y@savepix are local variables.  We make
% use of the fact that the values of the flag and positions propagate
% down to inferior segments but not up to superior segments.  This
% behaviour Is consistent with the gsave/grestore operation on the
% PostScript side.
\def\s@avemove #1#2{\x@savepix=#1\y@savepix=#2%
                    \m@pendingtrue
                    \ifp@osinit \else
                      \global\p@osinittrue
                      \global\xminpix=\x@savepix \global\yminpix=\y@savepix
                      \global\xmaxpix=\x@savepix \global\ymaxpix=\y@savepix
                    \fi}

\def\f@lushmove {\global\p@osinittrue
                 \ifm@pending
                   \writetx {\the\x@savepix\space \the\y@savepix\space mv}%
                   \m@pendingfalse
                   \p@athfalse
                 \fi}

% =====  Flush begin segment
% \f@lushbs flushes any saved up \bsegments.  Some of these may be
% redundant, but we cannot know without looking ahead beyond the
% \esegment.
\def\f@lushbs {\loop
                 \ifnum \d@bs>0
                   \writetx {bs}%
                   \global\advance \d@bs by -1
               \repeat}
               
% ===== Internal text macros
% Horizontal text, use only 3 levels of box nesting here
\def\h@move #1#2 #3)#4{\move (#2 #3)%
                       \h@text {#4}}
\def\h@text #1{\pixtodim \x@pix \t@xpos
               \pixtodim \y@pix \t@ypos
               \vbox to 0pt{\normalbaselines
                            \t@stuff
                            \kern -\t@ypos
                            \hbox to 0pt{\l@stuff
                                         \kern \t@xpos
                                         \hbox {#1}%
                                         \kern -\t@xpos
                                         \r@stuff}%
                            \kern \t@ypos
                            \b@stuff\relax}}

% Rotated text
% Uses 5 levels of box nesting here (so that the text reference point
% is that <before> rotation).  This was done so that the reference point
% definition makes sense with arbitrary angle rotation.  The text is
% rotated with respect to the text reference point.  The result is zero
% sized.  These macros generate in-line PostScript.
% #1 - rotation angle in degrees
% #2 - text to be rotated
\def\r@move td:#1 #2#3 #4)#5{\move (#3 #4)%
                             \r@text td:#1 {#5}}
\def\r@text td:#1 #2{\vbox to 0pt{\pixtodim \x@pix \t@xpos
                                  \pixtodim \y@pix \t@ypos
                                  \kern -\t@ypos
                                  \hbox to 0pt{\kern \t@xpos
                                               \rottxt {#1}{\z@sb {#2}}%
                                               \hss}%
                                  \vss}}
\def\z@sb #1{\vbox to 0pt{\normalbaselines
                          \t@stuff
                          \hbox to 0pt{\l@stuff \hbox {#1}\r@stuff}%
                          \b@stuff}}

% ===== Rotate text, in-line PostScript code
\ifx \rotatebox\@undefined
  \def\rottxt #1#2{\bgroup
                     \special {ps: gsave currentpoint currentpoint translate
                               #1\space neg rotate
                              neg exch neg exch translate}%
                     #2%
                     \special {ps: currentpoint grestore moveto}%
                   \egroup}
\else
  \let\rottxt=\rotatebox
\fi

% ===== Error message
% If not defined, use the plain TeX errmessage macro
\ifx \t@xderror\@undefined
  \let\t@xderror=\errmessage
\fi

% ===== Default values
% These are reset each time TeXdraw is invoked
\def\t@exdrawdef {\sppix 300/in            % 300 pixels/inch
                  \drawdim in              % drawing units are inches
                  \edef\u@nitsc {1}%       % unit scale 1 (has to be set
                                           % before invoking \setsegscale)
                  \setsegscale 1           % segment scale 1
                  \arrowheadsize l:0.16 w:0.08
                  \arrowheadtype t:T
                  \textref h:L v:B }


% ===============================================================
% ===== PostScript file macros ==================================

% ===== Include the TeXdraw graphics
% The drawing in a box of appropriate size will be placed such that its
% lower left hand corner will be at the current TeX position.
\ifx \includegraphics\@undefined
  \def\t@xdinclude [#1,#2][#3,#4]#5{%
    \begingroup                           % keep definitions local
      \message {<#5>}%
      \leavevmode
      \t@counta=-#1%                      % integer bounding box coordinates
      \t@countb=-#2%
      \setbox0=\hbox{%
        \special {PSfile="#5"\space
                  hoffset=\the\t@counta\space voffset=\the\t@countb}}%
      \t@ypos=#4 bp%
        \advance \t@ypos by -#2 bp%
      \t@xpos=#3 bp%
        \advance \t@xpos by -#1 bp%
      \dp0=0pt \ht0=\t@ypos  \wd0=\t@xpos
      \box0%
    \endgroup}
\else
  \let\t@xdinclude=\includegraphics
\fi

% ===== Write to the PostScript file
% Macro to write PostScript commands to the temporary PostScript file
% To decrease the size of the PostScript file, moves are kept back to
% allow redundant multiple moves to be removed.  In addition empty gsave/
% grestore pairs are not written.  The PostScript file is not opened if
% TeXdraw has not generated any PostScript commands, i.e. it has produced
% only TeX text.
% \writeps : flushes the pending move to make sure things are positioned
%            correctly and flushes pending begin segments before calling
%            \writetx to write to the PostScript file
% \writetx : writes directly to the PostScript file.  This version is used
%            only for those commands which just change line parameters
%            without drawing.  This routine opens the file and writes the
%            PS file header the first time it is called.
% \w@rps :   lowest level direct write to the PostScript file
\def\writeps #1{\f@lushbs
                \f@lushmove
                \p@athtrue
                \writetx {#1}}
\def\writetx #1{\ift@extonly
                  \global\t@extonlyfalse
                  \t@xdpsfn \p@sfile
                  \t@dropen \p@sfile
                \fi
                \w@rps {#1}}
\def\w@rps #1{\immediate\write\drawfile {#1}}

% ===== Open a PostScript file
% Open a PostScript file, write the definitions used by TeXdraw.
\def\t@xdpsfn #1{%
  \global\advance \t@xdnum by 1
  \ifnum \t@xdnum<10
    \xdef #1{\jobname.ps\the\t@xdnum}
  \else
    \xdef #1{\jobname.p\the\t@xdnum}
  \fi
}
\def\t@dropen #1{%
  \immediate\openout\drawfile=#1%
  \w@rps {\p@b PS-Adobe-3.0 EPSF-3.0}%
  \w@rps {\p@p BoundingBox: (atend)}%
  \w@rps {\p@p Title: TeXdraw drawing: #1}%
  \w@rps {\p@p Pages: 1}%
  \w@rps {\p@p Creator: \TeXdrawId}%
  \w@rps {\p@p CreationDate: \the\year/\the\month/\the\day}%
  \w@rps {50 dict begin}%
  \w@rps {/mv {stroke moveto} def}%
  \w@rps {/lv {lineto} def}%
  \w@rps {/st {currentpoint stroke moveto} def}%
  \w@rps {/sl {st setlinewidth} def}%
  \w@rps {/sd {st 0 setdash} def}%
  \w@rps {/sg {st setgray} def}%
  \w@rps {/bs {gsave} def /es {stroke grestore} def}%
  \w@rps {/fl \l@br gsave setgray fill grestore}%
  \w@rps    { currentpoint newpath moveto\r@br\space def}%
  \w@rps {/fp {gsave setgray fill grestore st} def}%
  \w@rps {/cv {curveto} def}%
  \w@rps {/cr \l@br gsave currentpoint newpath 3 -1 roll 0 360 arc}%
  \w@rps    { stroke grestore\r@br\space def}%
  \w@rps {/fc \l@br gsave setgray currentpoint newpath}%
  \w@rps    { 3 -1 roll 0 360 arc fill grestore\r@br\space def}%
  \w@rps {/ar {gsave currentpoint newpath 5 2 roll arc stroke grestore} def}%
  \w@rps {/el \l@br gsave /svm matrix currentmatrix def}%
  \w@rps    { currentpoint translate scale newpath 0 0 1 0 360 arc}%
  \w@rps    { svm setmatrix stroke grestore\r@br\space def}%
  \w@rps {/fe \l@br gsave setgray currentpoint translate scale newpath}%
  \w@rps    { 0 0 1 0 360 arc fill grestore\r@br\space def}%
  \w@rps {/av \l@br /hhwid exch 2 div def /hlen exch def}%
  \w@rps    { /ah exch def /tipy exch def /tipx exch def}%
  \w@rps    { currentpoint /taily exch def /tailx exch def}%
  \w@rps    { /dx tipx tailx sub def /dy tipy taily sub def}%
  \w@rps    { /alen dx dx mul dy dy mul add sqrt def}%
  \w@rps    { /blen alen hlen sub def}%
  \w@rps    { gsave tailx taily translate dy dx atan rotate}%
  \w@rps    { (V) ah ne {blen 0 gt {blen 0 lineto} if} {alen 0 lineto} ifelse}%
  \w@rps    { stroke blen hhwid neg moveto alen 0 lineto blen hhwid lineto}%
  \w@rps    { (T) ah eq {closepath} if}%
  \w@rps    { (W) ah eq {gsave 1 setgray fill grestore closepath} if}%
  \w@rps    { (F) ah eq {fill} {stroke} ifelse}%
  \w@rps    { grestore tipx tipy moveto\r@br\space def}%
  \w@rps {\p@sfactor\space \p@sfactor\space scale}%
  \w@rps {1 setlinecap 1 setlinejoin}%
  \w@rps {3 setlinewidth [] 0 setdash}%
  \w@rps {0 0 moveto}%
}

% Notes:
% - mv (move to) This command includes a stroke before the moveto.  The
%      stroke terminates a path and the move begins another path.
% - bs (begin segment) encloses a segment in a gsave/grestore to keep
%      changes to line parameters local.
% - es (end segment) does a "stroke grestore" to make sure lines inside
%      the segment use the line parameters local to that segment
% - ar (arc) The path is generated and stroked inside a gsave/grestore,
%      leaving the current path intact.
% - cr (circle) The path is generated and stroked inside a gsave/grestore,
%      leaving the current path intact.
% - fc (filled circle) The path is generated and filled inside a gsave/
%      grestore, keeping the fill level local to the circle.  The current
%      path is left intact.
% - el (ellipse) The path is generated and stroked inside a gsave/grestore,
%      leaving the current path intact.  The elliptical path is defined
%      with different x and y scaling, then stroked with default scaling
%      to give a constant line thickness.
% - fe (filled ellipse) The path is generated and filled inside a gsave/
%      grestore, leaving the current path intact.
% - fl (fill) The current path is closed and filled inside a gsave/restore,
%      keeping the fill level local.  A newpath terminates the path.
% - fp (fill path) The current path is closed and then filled inside a
%      gsave/grestore.  Finally the closed path is stroked, implicitly
%      terminating the path.
% - av (arrow vector) The arrow vector is drawn inside a gsave/grestore.
%      The line width and type are those currently in effect.  After the
%      grestore, the current path is continued with a move to the tip of
%      the vector.

% ===== Close the PostScript file
% Write a trailer with the BoundingBox, close the file.  Note that the
% BoundingBox may be larger than the commands in the PostScript file
% indicate.  This is due to the fact that multiple move commands in
% a row are collapsed into a single move.  The BoundingBox information
% includes the effect of the moves which were expunged.
\def\t@drclose {%
  \bgroup
    \w@rps {stroke end showpage}%
    \w@rps {\p@p Trailer:}%
    \pixtobp \xminpix \l@lxbp  \pixtobp \yminpix \l@lybp
    \pixtobp \xmaxpix \u@rxbp  \pixtobp \ymaxpix \u@rybp
    \w@rps {\p@p BoundingBox: \l@lxbp\space \l@lybp\space
                              \u@rxbp\space \u@rybp}%
    \w@rps {\p@p EOF}%
  \egroup
  \immediate\closeout\drawfile
}

% ===============================================================
\catcode`\@=\catamp



%%%@@@ end input from texdraw


%%%@@@ input from paths
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%              Input-Datei zum Erzeugen von Gitterpunktwegen                  %
%                     von Christian Krattenthaler.                            %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%entering a path:
%  \Pfad(x-coordinate of starting point,y-coordinate of starting point),path
%  as 1-2-3-4-word\endPfad
%(1=step in x-direction, 2=step in y-direction, 3=upward diagonal step,  4=downward diagonal step)
%
%entering a dotted path:
%  \SPfad(x-coordinate of starting point,y-coordinate of starting point),path
%  as 1-2-3-4-word\endSPfad
%(1=step in x-direction, 2=step in y-direction, 3=upward diagonal step,  4=downward diagonal step)
%
%coordinate-axes:
%  \Koordinatenachsen(length of positive x-axes, length of positive  y-axes)(length of negative x-axes, length of negative y-axes)
%The length of negative axes are entered as negative numbers.
%
%lattice:
%  \Gitter(number of points in positive x-direction, number of points in  positive y-direction)(number of points in negative x-direction, number of  points in negative y-direction)
%
%diagonal lines:
%  \Diagonale(x-coordinate of SW-most point,y-coordinate of SW-most  point)length of the projection on the x-axes
%
%antidiagonal lines:
%  \AntiDiagonale(x-coordinate of NW-most point,y-coordinate of NW-most  point)length of the projection on the x-axes
%
%vectors:
%  \Vektor(x-coordinate of incline, y-coordinate of  incline)length(x-coordinate of starting point, y-coordinate of starting point)
%
%labelling of points:
%  \Label[location?]{[label]}(x-coordinate,y-coordinate)
%where:
%  [location?]=\l,\lo,\lu,\r,\ro,\ru,\o,\u
%and l=left, r=right, u=bottom, o=top.
%In addition, if by \Einheit?cm the basic unit is changed, there exist
%\llo,\loo,\llu,\luu,\rro,\roo,\rru,\ruu.
%
%The basic unit can be changed by entering
%  \Einheit=?cm
%The default is \Einheit=0.5cm.
%
%The thickness of the paths can be changed by entering
%     \PfadDicke{?cm}
%The default is \PfadDicke=1pt.
%
%The following point sizes are available:
%\DuennPunkt, \NormalPunkt, \DickPunkt. Syntax:
%  \DickPunkt(x-coordinate,y-coordinate), etc.
%
%Besides, a circle is available by \Kreis. Syntax:
%  \Kreis(x-coordinate,y-coordinate)
%
\catcode`\@=11
\font\tenln    = line10
\font\tenlnw   = linew10

\newskip\Einheit \Einheit=0.5cm
\newcount\xcoord \newcount\ycoord
\newdimen\xdim \newdimen\ydim \newdimen\PfadD@cke \newdimen\Pfadd@cke

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%LaTeX counters, dimensions, variables for lines%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcount\@tempcnta
\newcount\@tempcntb

\newdimen\@tempdima
\newdimen\@tempdimb


\newdimen\@wholewidth
\newdimen\@halfwidth

\newcount\@xarg
\newcount\@yarg
\newcount\@yyarg
\newbox\@linechar
\newbox\@tempboxa
\newdimen\@linelen
\newdimen\@clnwd
\newdimen\@clnht

\newif\if@negarg

\def\@whilenoop#1{}
\def\@whiledim#1\do #2{\ifdim #1\relax#2\@iwhiledim{#1\relax#2}\fi}
\def\@iwhiledim#1{\ifdim #1\let\@nextwhile=\@iwhiledim
        \else\let\@nextwhile=\@whilenoop\fi\@nextwhile{#1}}

\def\@whileswnoop#1\fi{}
\def\@whilesw#1\fi#2{#1#2\@iwhilesw{#1#2}\fi\fi}
\def\@iwhilesw#1\fi{#1\let\@nextwhile=\@iwhilesw
         \else\let\@nextwhile=\@whileswnoop\fi\@nextwhile{#1}\fi}


\def\thinlines{\let\@linefnt\tenln \let\@circlefnt\tencirc
  \@wholewidth\fontdimen8\tenln \@halfwidth .5\@wholewidth}
\def\thicklines{\let\@linefnt\tenlnw \let\@circlefnt\tencircw
  \@wholewidth\fontdimen8\tenlnw \@halfwidth .5\@wholewidth}
\thinlines
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\PfadD@cke1pt \Pfadd@cke0.5pt
\def\PfadDicke#1{\PfadD@cke#1 \divide\PfadD@cke by2 \Pfadd@cke\PfadD@cke  
\multiply\PfadD@cke by2}
\long\def\LOOP#1\REPEAT{\def\BODY{#1}\ITERATE}
\def\ITERATE{\BODY \let\next\ITERATE \else\let\next\relax\fi \next}
\let\REPEAT=\fi
\def\Punkt{\hbox{\raise-2pt\hbox to0pt{\hss$\ssize\bullet$\hss}}}
\def\DuennPunkt(#1,#2){\unskip
  \raise#2 \Einheit\hbox to0pt{\hskip#1 \Einheit
          \raise-2.5pt\hbox to0pt{\hss$\bullet$\hss}\hss}}
\def\NormalPunkt(#1,#2){\unskip
  \raise#2 \Einheit\hbox to0pt{\hskip#1 \Einheit
          \raise-3pt\hbox to0pt{\hss\twelvepoint$\bullet$\hss}\hss}}
\def\DickPunkt(#1,#2){\unskip
  \raise#2 \Einheit\hbox to0pt{\hskip#1 \Einheit
          \raise-4pt\hbox to0pt{\hss\fourteenpoint$\bullet$\hss}\hss}}
\def\Kreis(#1,#2){\unskip
  \raise#2 \Einheit\hbox to0pt{\hskip#1 \Einheit
          \raise-4pt\hbox to0pt{\hss\fourteenpoint$\circ$\hss}\hss}}

%%%%%%%%%%%%%%%%%%%%%
%LaTeX line macros%
%%%%%%%%%%%%%%%%%%%%%
\def\Line@(#1,#2)#3{\@xarg #1\relax \@yarg #2\relax
\@linelen=#3\Einheit
\ifnum\@xarg =0 \@vline
  \else \ifnum\@yarg =0 \@hline \else \@sline\fi
\fi}

\def\@sline{\ifnum\@xarg< 0 \@negargtrue \@xarg -\@xarg \@yyarg -\@yarg
  \else \@negargfalse \@yyarg \@yarg \fi
\ifnum \@yyarg >0 \@tempcnta\@yyarg \else \@tempcnta -\@yyarg \fi
\ifnum\@tempcnta>6 \@badlinearg\@tempcnta0 \fi
\ifnum\@xarg>6 \@badlinearg\@xarg 1 \fi
\setbox\@linechar\hbox{\@linefnt\@getlinechar(\@xarg,\@yyarg)}%
\ifnum \@yarg >0 \let\@upordown\raise \@clnht\z@
   \else\let\@upordown\lower \@clnht \ht\@linechar\fi
\@clnwd=\wd\@linechar
\if@negarg \hskip -\wd\@linechar \def\@tempa{\hskip -2\wd\@linechar}\else
     \let\@tempa\relax \fi
\@whiledim \@clnwd <\@linelen \do
  {\@upordown\@clnht\copy\@linechar
   \@tempa
   \advance\@clnht \ht\@linechar
   \advance\@clnwd \wd\@linechar}%
\advance\@clnht -\ht\@linechar
\advance\@clnwd -\wd\@linechar
\@tempdima\@linelen\advance\@tempdima -\@clnwd
\@tempdimb\@tempdima\advance\@tempdimb -\wd\@linechar
\if@negarg \hskip -\@tempdimb \else \hskip \@tempdimb \fi
\multiply\@tempdima \@m
\@tempcnta \@tempdima \@tempdima \wd\@linechar \divide\@tempcnta \@tempdima
\@tempdima \ht\@linechar \multiply\@tempdima \@tempcnta
\divide\@tempdima \@m
\advance\@clnht \@tempdima
\ifdim \@linelen <\wd\@linechar
   \hskip \wd\@linechar
  \else\@upordown\@clnht\copy\@linechar\fi}

\def\@hline{\ifnum \@xarg <0 \hskip -\@linelen \fi
\vrule height\Pfadd@cke width \@linelen depth\Pfadd@cke
\ifnum \@xarg <0 \hskip -\@linelen \fi}

\def\@getlinechar(#1,#2){\@tempcnta#1\relax\multiply\@tempcnta 8
\advance\@tempcnta -9 \ifnum #2>0 \advance\@tempcnta #2\relax\else
\advance\@tempcnta -#2\relax\advance\@tempcnta 64 \fi
\char\@tempcnta}

\def\Vektor(#1,#2)#3(#4,#5){\unskip\leavevmode
  \xcoord#4\relax \ycoord#5\relax
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \Vector@(#1,#2){#3}\hss}}

\def\Vector@(#1,#2)#3{\@xarg #1\relax \@yarg #2\relax
\@tempcnta \ifnum\@xarg<0 -\@xarg\else\@xarg\fi
\ifnum\@tempcnta<5\relax
\@linelen=#3\Einheit
\ifnum\@xarg =0 \@vvector
  \else \ifnum\@yarg =0 \@hvector \else \@svector\fi
\fi
\else\@badlinearg\fi}

\def\@hvector{\@hline\hbox to 0pt{\@linefnt
\ifnum \@xarg <0 \@getlarrow(1,0)\hss\else
    \hss\@getrarrow(1,0)\fi}}

\def\@vvector{\ifnum \@yarg <0 \@downvector \else \@upvector \fi}

\def\@svector{\@sline
\@tempcnta\@yarg \ifnum\@tempcnta <0 \@tempcnta=-\@tempcnta\fi
\ifnum\@tempcnta <5
  \hskip -\wd\@linechar
  \@upordown\@clnht \hbox{\@linefnt  \if@negarg
  \@getlarrow(\@xarg,\@yyarg) \else \@getrarrow(\@xarg,\@yyarg) \fi}%
\else\@badlinearg\fi}

\def\@upline{\hbox to \z@{\hskip -.5\Pfadd@cke \vrule width \Pfadd@cke
   height \@linelen depth \z@\hss}}

\def\@downline{\hbox to \z@{\hskip -.5\Pfadd@cke \vrule width \Pfadd@cke
   height \z@ depth \@linelen \hss}}

\def\@upvector{\@upline\setbox\@tempboxa\hbox{\@linefnt\char'66}\raise
     \@linelen \hbox to\z@{\lower \ht\@tempboxa\box\@tempboxa\hss}}

\def\@downvector{\@downline\lower \@linelen
      \hbox to \z@{\@linefnt\char'77\hss}}

\def\@getlarrow(#1,#2){\ifnum #2 =\z@ \@tempcnta='33\else
\@tempcnta=#1\relax\multiply\@tempcnta \sixt@@n \advance\@tempcnta
-9 \@tempcntb=#2\relax\multiply\@tempcntb \tw@
\ifnum \@tempcntb >0 \advance\@tempcnta \@tempcntb\relax
\else\advance\@tempcnta -\@tempcntb\advance\@tempcnta 64
\fi\fi\char\@tempcnta}

\def\@getrarrow(#1,#2){\@tempcntb=#2\relax
\ifnum\@tempcntb < 0 \@tempcntb=-\@tempcntb\relax\fi
\ifcase \@tempcntb\relax \@tempcnta='55 \or
\ifnum #1<3 \@tempcnta=#1\relax\multiply\@tempcnta
24 \advance\@tempcnta -6 \else \ifnum #1=3 \@tempcnta=49
\else\@tempcnta=58 \fi\fi\or
\ifnum #1<3 \@tempcnta=#1\relax\multiply\@tempcnta
24 \advance\@tempcnta -3 \else \@tempcnta=51\fi\or
\@tempcnta=#1\relax\multiply\@tempcnta
\sixt@@n \advance\@tempcnta -\tw@ \else
\@tempcnta=#1\relax\multiply\@tempcnta
\sixt@@n \advance\@tempcnta 7 \fi\ifnum #2<0 \advance\@tempcnta 64 \fi
\char\@tempcnta}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\Diagonale(#1,#2)#3{\unskip\leavevmode
  \xcoord#1\relax \ycoord#2\relax
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \Line@(1,1){#3}\hss}}
\def\AntiDiagonale(#1,#2)#3{\unskip\leavevmode
  \xcoord#1\relax \ycoord#2\relax %\advance\xcoord by -0.05\relax
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \Line@(1,-1){#3}\hss}}
\def\Pfad(#1,#2),#3\endPfad{\unskip\leavevmode
  \xcoord#1 \ycoord#2 \thicklines\ZeichnePfad#3\endPfad\thinlines}
\def\ZeichnePfad#1{\ifx#1\endPfad\let\next\relax
  \else\let\next\ZeichnePfad
    \ifnum#1=1
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \vrule height\Pfadd@cke width1 \Einheit depth\Pfadd@cke\hss}%
      \advance\xcoord by 1
    \else\ifnum#1=2
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
        \hbox{\hskip-\PfadD@cke\vrule height1 \Einheit width\PfadD@cke  
depth0pt}\hss}%
      \advance\ycoord by 1
    \else\ifnum#1=3
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \Line@(1,1){1}\hss}
      \advance\xcoord by 1
      \advance\ycoord by 1
    \else\ifnum#1=4
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \Line@(1,-1){1}\hss}
      \advance\xcoord by 1
      \advance\ycoord by -1
    \fi\fi\fi\fi
  \fi\next}
\def\hSSchritt{\leavevmode\raise-.4pt\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.4pt\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.4pt\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.4pt\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.4pt\hbox to0pt{\hss.\hss}\hskip.2\Einheit}
\def\vSSchritt{\vbox{\baselineskip.2\Einheit\lineskiplimit0pt
\hbox{.}\hbox{.}\hbox{.}\hbox{.}\hbox{.}}}
\def\DSSchritt{\leavevmode\raise-.4pt\hbox to0pt{%
  \hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise.2\Einheit\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise.4\Einheit\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise.6\Einheit\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise.8\Einheit\hbox to0pt{\hss.\hss}\hss}}
\def\dSSchritt{\leavevmode\raise-.4pt\hbox to0pt{%
  \hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.2\Einheit\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.4\Einheit\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.6\Einheit\hbox to0pt{\hss.\hss}\hskip.2\Einheit
  \raise-.8\Einheit\hbox to0pt{\hss.\hss}\hss}}
\def\SPfad(#1,#2),#3\endSPfad{\unskip\leavevmode
  \xcoord#1 \ycoord#2 \ZeichneSPfad#3\endSPfad}
\def\ZeichneSPfad#1{\ifx#1\endSPfad\let\next\relax
  \else\let\next\ZeichneSPfad
    \ifnum#1=1
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \hSSchritt\hss}%
      \advance\xcoord by 1
    \else\ifnum#1=2
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
        \hbox{\hskip-2pt \vSSchritt}\hss}%
      \advance\ycoord by 1
    \else\ifnum#1=3
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \DSSchritt\hss}
      \advance\xcoord by 1
      \advance\ycoord by 1
    \else\ifnum#1=4
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit
         \dSSchritt\hss}
      \advance\xcoord by 1
      \advance\ycoord by -1
    \fi\fi\fi\fi
  \fi\next}
\def\Koordinatenachsen(#1,#2){\unskip
 \hbox to0pt{\hskip-.5pt\vrule height#2 \Einheit width.5pt depth1 \Einheit}%
 \hbox to0pt{\hskip-1 \Einheit \xcoord#1 \advance\xcoord by1
    \vrule height0.25pt width\xcoord \Einheit depth0.25pt\hss}}
\def\Koordinatenachsen(#1,#2)(#3,#4){\unskip
 \hbox to0pt{\hskip-.5pt \ycoord-#4 \advance\ycoord by1
    \vrule height#2 \Einheit width.5pt depth\ycoord \Einheit}%
 \hbox to0pt{\hskip-1 \Einheit \hskip#3\Einheit
    \xcoord#1 \advance\xcoord by1 \advance\xcoord by-#3
    \vrule height0.25pt width\xcoord \Einheit depth0.25pt\hss}}
\def\Gitter(#1,#2){\unskip \xcoord0 \ycoord0 \leavevmode
  \LOOP\ifnum\ycoord<#2
    \loop\ifnum\xcoord<#1
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit\Punkt\hss}%
      \advance\xcoord by1
    \repeat
    \xcoord0
    \advance\ycoord by1
  \REPEAT}
\def\Gitter(#1,#2)(#3,#4){\unskip \xcoord#3 \ycoord#4 \leavevmode
  \LOOP\ifnum\ycoord<#2
    \loop\ifnum\xcoord<#1
      \raise\ycoord \Einheit\hbox to0pt{\hskip\xcoord \Einheit\Punkt\hss}%
      \advance\xcoord by1
    \repeat
    \xcoord#3
    \advance\ycoord by1
  \REPEAT}
\def\Label#1#2(#3,#4){\unskip \xdim#3 \Einheit \ydim#4 \Einheit
  \def\lo{\advance\xdim by-.5 \Einheit \advance\ydim by.5 \Einheit}%
  \def\llo{\advance\xdim by-.25cm \advance\ydim by.5 \Einheit}%
  \def\loo{\advance\xdim by-.5 \Einheit \advance\ydim by.25cm}%
  \def\o{\advance\ydim by.25cm}%
  \def\ro{\advance\xdim by.5 \Einheit \advance\ydim by.5 \Einheit}%
  \def\rro{\advance\xdim by.25cm \advance\ydim by.5 \Einheit}%
  \def\roo{\advance\xdim by.5 \Einheit \advance\ydim by.25cm}%
  \def\l{\advance\xdim by-.30cm}%
  \def\r{\advance\xdim by.30cm}%
  \def\lu{\advance\xdim by-.5 \Einheit \advance\ydim by-.6 \Einheit}%
  \def\llu{\advance\xdim by-.25cm \advance\ydim by-.6 \Einheit}%
  \def\luu{\advance\xdim by-.5 \Einheit \advance\ydim by-.30cm}%
  \def\u{\advance\ydim by-.30cm}%
  \def\ru{\advance\xdim by.5 \Einheit \advance\ydim by-.6 \Einheit}%
  \def\rru{\advance\xdim by.25cm \advance\ydim by-.6 \Einheit}%
  \def\ruu{\advance\xdim by.5 \Einheit \advance\ydim by-.30cm}%
  #1\raise\ydim\hbox to0pt{\hskip\xdim
     \vbox to0pt{\vss\hbox to0pt{\hss$#2$\hss}\vss}\hss}%
}
\catcode`\@=13

%Christian's texdraw definitions
\def\ldreieck{\bsegment
  \rlvec(0.866025403784439 .5) \rlvec(0 -1)
  \rlvec(-0.866025403784439 .5)
  \savepos(0.866025403784439 -.5)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\rdreieck{\bsegment
  \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)  \rlvec(0 1)
  \savepos(0 -1)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\rhombus{\bsegment
  \rlvec(0.866025403784439 .5) \rlvec(0.866025403784439 -.5)
  \rlvec(-0.866025403784439 -.5)  \rlvec(0 1)
  \rmove(0 -1)  \rlvec(-0.866025403784439 .5)
  \savepos(0.866025403784439 -.5)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\RhombusA{\bsegment
  \rlvec(0.866025403784439 .5) \rlvec(0.866025403784439 -.5)
  \rlvec(-0.866025403784439 -.5) \rlvec(-0.866025403784439 .5)
  \savepos(0.866025403784439 -.5)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\RhombusB{\bsegment
  \rlvec(0.866025403784439 .5) \rlvec(0 -1)
  \rlvec(-0.866025403784439 -.5) \rlvec(0 1)
  \savepos(0 -1)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\RhombusC{\bsegment
  \rlvec(0.866025403784439 -.5) \rlvec(0 -1)
  \rlvec(-0.866025403784439 .5) \rlvec(0 1)
  \savepos(0.866025403784439 -.5)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\hdSchritt{\bsegment
  \lpatt(.05 .13)
  \rlvec(0.866025403784439 -.5)
  \savepos(0.866025403784439 -.5)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }
\def\vdSchritt{\bsegment
  \lpatt(.05 .13)
  \rlvec(0 -1)
  \savepos(0 -1)(*ex *ey)
        \esegment
  \move(*ex *ey)
        }




%%%@@@ end input from paths

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Auxiliary macros:                                                           %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\arccot}{\operatorname{arccot}}			
\newcommand{\HypFsimple}[2]{\sideset{_#1}{_#2}{F}}
\newcommand{\HypF}[5]{\sideset{_#1}{_#2}{F}\!\left[\matrix  
#3\\#4\endmatrix;{\displaystyle #5}\right]}
\def\po#1#2{(#1)_{#2}}
\newcommand{\card}[1]{\left|#1\right|}
\newcommand{\abs}[1]{\left|#1\right|}
\newcommand{\floor}[1]{\left\lfloor#1\right\rfloor}
\newcommand{\ceil}[1]{\left\lceil#1\right\rceil}
\newcommand{\arcsinh}{\operatorname{arcsinh}}
\newcommand{\sgn}{\operatorname{sgn}}
\newcommand{\mydet}[1]{\det\left(#1\right)}
\newcommand{\dx}{{\,dx}}
\def\ep{\varepsilon}
\def\la{\lambda}

\newtheorem{thm}{Theorem}
\newtheorem{dfn}[thm]{Definition}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{lem}[thm]{Lemma}
\newtheorem*{conj}{Conjecture}

\theoremstyle{remark}
\newtheorem*{rem}{Remark}

\catcode`\@=11
\def\sideset#1#2#3{%
  \@mathmeasure\z@\displaystyle{#3}%
  \global\setbox\@ne\vbox to\ht\z@{}\dp\@ne\dp\z@
  \setbox\tw@\box\@ne
  \@mathmeasure4\displaystyle{\copy\tw@#1}%
  \@mathmeasure6\displaystyle{#3{#2}}%
  \dimen@-\wd6 \advance\dimen@\wd4 \advance\dimen@\wd\z@
  \hbox to\dimen@{}\mathop{\kern-\dimen@\box4\box6}%
}
\catcode`\@=12


\numberwithin{equation}{section}

\def\newstep#1{\par\vskip6pt\noindent{\it #1.\/ }}


% Imaginary unit:
\def\complexi{{\sqrt{-1}}}
\def\complexikl{{(\sqrt{-1})}}

\begin{document}
\begin{thebibliography}{10}

\bibitem{AlCaAA}
W. A. Al-Salam and L. Carlitz, {\em Some determinants of Bernoulli, Euler 
and related numbers}, Portugaliae Math\@. {\bf 18} (1959), 91--99.

\bibitem{AskeyCHP}
R.~Askey, {\em Continuous Hahn polynomials\/}, J. Phys\@. A --
Math\@. Gen\@. {\bf 18} (1985), L1017--L1019.

\bibitem{AtSuAA} N. M. Atakishiyev and S. K. Suslov, {\em The Hahn
and Meixner polynomials of an imaginary argument and some of their
applications}, J. Phys\@. A -- Math\@. Gen\@. {\bf 18} (1985),
1583--1596.

\bibitem{Bromwich}
T.~J.~Bromwich, {\em An Introduction to the Theory of Infinite Series\/},
MacMillan, London, 1949.

\bibitem{CiucAH}
M.    Ciucu, {\em Plane partitions I: A generalization of MacMahon's
formula}, preprint, {\tt math/9808017}.

\bibitem{CiEKAA}
M. Ciucu, T. Eisenk\"olbl, C. Krattenthaler and D. Zare, {\em
Enumeration of lozenge tilings of hexagons with a central 
triangular hole}, preprint.

\bibitem{CiucKratAB}
M.~Ciucu and C.~Krattenthaler, {\em The number of centered lozenge tilings of a
symmetric hexagon\/}, J.\ Combin\@. Theory Ser.~A {\bf 86} (1999), 103--126.

\bibitem{CiKrAB} 
M.    Ciucu and C. Krattenthaler, {\em Plane partitions II: $5\frac {1}{2}$ 
symmetry classes}, in: Combinatorial Methods in Representation 
Theory, M. Kashiwara, K. Koike, S. Okada, I. Terada, H. Yamada,
eds., Advanced Study in Pure Mathematics, RIMS, Kyoto (to appear).

\bibitem{CiKrAC} 
M.    Ciucu and C. Krattenthaler, {\em Enumeration of lozenge 
tilings of hexagons with cut off corners}, in preparation.

\bibitem{CoLPAA} 
H.    Cohn, M. Larsen and J. Propp, 
{\em The shape of a typical boxed plane partition},
New York J. Math\@. {\bf 4} (1998), 137--166.

\bibitem{DT}
G. David and C. Tomei, {\em The problem of the calissons}, 
Amer\@. Math\@. Monthly\@. {\bf 96} (1989), 429--431.

\bibitem{Erdelyi}
A. Erd\'elyi, V. Magnus, F. Oberhettinger and F. Tricomi, {\em Higher transcendental functions}, McGraw--Hill, New
York, 1953.

\bibitem{FuKrAC}
M. Fulmek and C. Krattenthaler, {\em The number of rhombus
tilings of a symmetric hexagon which contain a fixed rhombus on the
symmetry axis, I}, Ann.\ Combin.\ 2 (1998), 19--40.

\bibitem{Knuth}
R. L. Graham, D. E. Knuth and O. Patashnik,
{\em Concrete Mathematics\/},
Addison Wesley, Reading, Massachusetts, 1988.

\bibitem{GuOVAA}
A. J. Guttmann, A. L. Owczarek and X. G. Viennot, {\em 
Vicious walkers and Young tableaux I: Without walls}, J.~Phys\@. A -- 
Math\@. Gen\@. {\bf 31} (1998), 8123--8135.

\bibitem{HeGeAA} 
H.    Helfgott and I. M. Gessel,
{\em Exact enumeration of certain tilings of diamonds and hexagons with
defects}, Elect.\ J.\ Combin.\ {\bf 6} (1) (1999), \#R16, 26~pp.

\bibitem{JohaAA}
K. Johansson, {\em Discrete orthogonal polynomial ensembles and the
Plancherel measure}, preprint, {\tt math/9906120}.

\bibitem{JoThAA}
W. B. Jones and W. J. Thron,
{\em Continued Fractions}, Addison--Wesley, Reading, Massachusetts,
1980.

\bibitem{KarrAA}
M.    Karr, {\em Summation in finite terms}, J. Assoc\@. Comput\@.
Machinery {\bf 28} (1981), 305--350.

\bibitem{KoSwAA}
R.    Koekoek and R. F. Swarttouw,
{\em The Askey--scheme of hypergeometric orthogonal polynomials and 
its $q$-analogue}, TU Delft, The Netherlands, 1998; on the WWW:
\newline 
{\tt ftp://ftp.twi.tudelft.nl/TWI/publications/tech-reports/1998/DUT-TWI-98-17.ps.gz}.

\bibitem{KratBI}
C.~Krattenthaler, {\em An alternative evaluation of the Andrews--Burge  
determinant\/}, in: Mathematical Essays in Honor of Gian-Carlo Rota,
B.~E.~Sagan, R.~P.~Stanley, eds., Progress in Math., vol.~161, 
Birkh\"auser, Boston, 1998, pp.~263--270.

\bibitem{KratBD}
C.~Krattenthaler, {\em Determinant identities and a generalization of the  
number of totally symmetric self-complementary plane partitions\/},
Electron.\ J.\ Combin.\ {\bf 4}(1) (1997), \#R27, 62~pp.

\bibitem{KratBN}
C.    Krattenthaler, {\em Advanced determinant calculus}, S\'eminaire 
Lotharingien Combin\@. {\bf 42} (1999) (``The Andrews Festschrift"),
paper~B42q, 67~pp.

\bibitem{KupeAA}
G.    Kuperberg, {\em Symmetries of plane partitions and the permanent 
determinant method}, J.~Combin\@. Theory Ser\@. A {\bf 68} (1994),
115--151.

\bibitem{KupeAF}
G.    Kuperberg, {\em Four symmetry classes of plane partitions under one
roof}, J. Combin\@. Theory Ser.~A {\bf 75} (1996), 295--315.

\bibitem{KupeAG}
G.    Kuperberg, {\em An exploration of the permanent-determinant
method}, Electron\@. J. Combin\@. {\bf 5} (1998), \#R46, 34~pp.

\bibitem{Leclerc}
B.~Leclerc, {\em On certain formulas of Karlin and Szeg\H o},
S\'eminaire Lotharingien Combin\@. {\bf 41} (1998), paper~B41d, 21~pp.

\bibitem{MacMahon}
P.~A.~MacMahon, {\em Combinatory Analysis\/}, vol.~2, Cambridge University
Press, 1916; reprinted by Chelsea, New York, 1960.

\bibitem{MallAA}
C. Mallinger, {\em Algorithmic manipulations and transformations of 
univariate holonomic functions and sequences}, diploma thesis, RISC, 
J. Kepler University, Linz, 1996.

\bibitem{MehtAB}
M. L. Mehta, {\em Matrix theory}, Editions de Physique, Orsay,
France, 1989.

\bibitem{MuirAB}
T. Muir, {\em The theory of determinants in the historical order of
development}, 4~vols., Macmillan, London, 1906--1923. 

\bibitem{MuirAD}
T. Muir, {\em A treatise on the theory of determinants}, 
Longmans, Green \& Co., London, 1933.

\bibitem{Noerlund}
N.~E.~N\"orlund, {\em Vorlesungen \"uber Differenzenrechnung\/}, Springer,
Berlin, 1924.

\bibitem{OkKrAA}
S.    Okada and C. Krattenthaler, {\em The number of rhombus tilings of 
a ``punctured" hexagon and the minor summation formula}, Adv\@. Appl\@. 
Math\@. {\bf 21} (1998), 381--404.

\bibitem{PaScAA} P.    Paule and M. Schorn, {\em A
Mathematica version of Zeilberger's algorithm for proving binomial
coefficient identities}, J. Symbol\@. Comp\@. {\bf 20} (1995),
673--698.

\bibitem{PeWZAA} M.    Petkov\v sek, H. Wilf and D. Zeilberger,
{\em A=B}, A.~K.~Peters, Wellesley, 1996.

\bibitem{PerronKB}
O.~Perron, {\em Die Lehre von den Kettenbr\"uchen\/},
B.~G.~Teubner, Stuttgart, 1977.

\bibitem{ProcAE}
R. A. Proctor, {\em Shifted plane partitions of trapezoidal shape},
Proc\@. Amer\@. Math\@. Soc\@. {\bf 89} (1983), 553--559.

\bibitem{ProcAD}
R. A. Proctor, {\em Bruhat lattices, plane partition generating functions, 
and minuscule representations}, Europ\@. J. Combin\@. {\bf 5} (1984), 
331--350.

\bibitem{ProcAB}
R. A. Proctor, {\em New symmetric plane partition identities from invariant 
theory work of DeConcini and Procesi}, Europ\@. J. Combin\@. {\bf
11} (1990), 289--300.

\bibitem{PropAA}
J.~Propp, {\em Twenty open problems on enumeration of matchings\/},
manuscript, 1996, {\tt math.CO/9801060}.

\bibitem{Propp2}
J.~Propp, {\em Enumeration of matchings: Problems and progress},
in: "New Perspectives in Geometric
 Combinatorics", L.~Billera, A.~Bj\"orner, C.~Greene, R.~Simion, and
R.~P.~Stanley, eds.,
 Mathematical Sciences Research Institute publication \#37, 
Cambridge University Press, 1999 (to appear).

\bibitem{Prudnikov}
A. P.~Prudnikov, Yu\@. A.~Brychkov, O. I.~Marichev, {\em Integrals and Series\/},
vol.~3: More Special Functions, Gordon and Breach, New York, London, 1989.

\bibitem{RobbAA}
D. P. Robbins. {\em The story of $1,2,7,42,429,7436,\dots$},
The Math\@. Intelligencer {\bf 13} (1991), 12--19.

\bibitem{SaZiAA}
B. Salvy and P. Zimmermann, {\em Gfun: a Maple package for the
manipulation of generating and holonomic functions in one 
       variable}, ACM Trans\@. Math\@.
       Software {\bf 20} (1994).

\bibitem{SlatAC}
L.~J.~Slater,
{\em Generalized Hypergeometric Functions\/},
Cambridge University Press, Cambridge, 1966.

\bibitem{SzegoOP}
G.~Szeg\H o,
{\em Orthogonal Polynomials\/},
American Mathematical Society Colloquium Publications Vol.~XXIII, New York, 1959.

\bibitem{VienAE}
X.~Viennot,
{\em  Une th\'eorie combinatoire des polyn\^omes orthogonaux g\'en\'eraux},
UQAM, Montreal, Quebec, 1983.

\bibitem{WallCF}
H.~S.~Wall,
{\em Analytic Theory of Continued Fractions},
Van Nostrand, New York, 1948.


\bibitem{ZeilAM} D.    Zeilberger,
{\em A fast algorithm for proving terminating hypergeometric
identities}, Discrete Math\@. {\bf 80} (1990), 207--211.

\bibitem{ZeilAV} D.    Zeilberger,
{\em The method of creative telescoping},
J. Symbolic Comput\@. {\bf 11} (1991), 195--204.

\end{thebibliography}


\title[The number of rhombus tilings]{The number of rhombus
tilings of a symmetric hexagon which contain a fixed rhombus on the
symmetry axis, II}

\author[M.~Fulmek and C.~Krattenthaler]
{M.~Fulmek and C.~Krattenthaler$^\dagger$}
\address{
Institut f\"ur Mathematik der Universit\"at Wien\\
Strudlhofgasse 4, A-1090 Wien, Austria}
\email{{\tt Mfulmek@Mat.Univie.Ac.At, Kratt@Pap.Univie.Ac.At}\newline\leavevmode\indent
{\it WWW}: {\tt http://radon.mat.univie.ac.at/\~{}mfulmek}\newline\leavevmode\indent
\hphantom{{\it WWW}: }{\tt http://radon.mat.univie.ac.at/People/kratt}
}

\thanks{$^\dagger$ Research partially supported by the Austrian
Science Foundation FWF, grant P13190-MAT}
\subjclass{Primary 05A15;
 Secondary 05A16 05A17 05A19 05B45 11B68 30B70 33C20 33C45 52C20}
\keywords{rhombus tilings, lozenge tilings, plane partitions,
nonintersecting lattice paths, determinant evaluations, Bernoulli numbers, 
Bernoulli polynomials,
Hankel determinants, continued fractions, orthogonal polynomials,
continuous Hahn polynomials}

\begin{abstract}
We compute the number
of rhombus tilings of a hexagon with side lengths $N,M,N,N,M,N$,
with $N$ and $M$ having the same parity, which
contain a particular rhombus next to the center of the hexagon.
The special case $N=M$ of one of our results solves a problem posed by Propp.
In the proofs, Hankel determinants featuring Bernoulli numbers play
an important role.
\end{abstract}

\maketitle


%%%@@@ input from intro
\section{Introduction}

Let $a$, $b$ and $c$ be positive integers, and consider a 
hexagon with side lengths $a,b,c,a,b,c$ whose angles are $120^\circ$
(see Figure~\ref{fig:0}.a). 
The subject of enumerating rhombus tilings 
of this hexagon (cf\@. Figure~\ref{fig:0}.b;
here, and in the sequel, by a
rhombus we always mean a rhombus with side lengths 1 and angles of
$60^\circ$ and $120^\circ$) gained a lot of interest recently.
This interest comes from two facts. First, it is a rich source of
non-trivial enumeration problems which have (or appear to have)
beautiful solutions (see e.g\@. 
\cite{CiucAH,CiEKAA,CiucKratAB,CiKrAB,CiKrAC,FuKrAC,KupeAA,Propp2,RobbAA}). 
Second, these problems are very often related to 
the theory of symmetric functions and/or the representation theory of
classical and quantum Lie algebras, and to statistical physics
(sometimes in disguise; see e.g\@. 
\cite{CoLPAA,GuOVAA,JohaAA,KupeAF,KupeAG,OkKrAA,ProcAE,ProcAD,ProcAB}).

\begin{figure}
%\vbox{
\centertexdraw{
  \drawdim truecm  \linewd.02
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -2)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -3)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus 
  \move (-1.732050807568877 -4)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus 
  \move (-1.732050807568877 -5)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus 
%Ein Rhombus Tiling des a,b,c,a,b,c Hexagon
\move(8 0)
\bsegment
  \drawdim truecm  \linewd.02
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -2)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -3)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus 
  \move (-1.732050807568877 -4)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus 
  \move (-1.732050807568877 -5)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus 
  \linewd.12
  \move(0 0)
  \RhombusA \RhombusB \RhombusB 
  \RhombusA \RhombusA \RhombusB \RhombusA \RhombusB \RhombusB
  \move (-0.866025403784439 -.5)
  \RhombusA \RhombusB \RhombusB \RhombusB \RhombusB
  \RhombusA \RhombusA \RhombusB \RhombusA 
  \move (-1.732050807568877 -1)
  \RhombusB \RhombusB \RhombusA \RhombusB \RhombusB \RhombusA
  \RhombusB \RhombusA \RhombusA 
  \move (1.732050807568877 0)
  \RhombusC \RhombusC \RhombusC 
  \move (1.732050807568877 -1)
  \RhombusC \RhombusC \RhombusC 
  \move (3.464101615137755 -3)
  \RhombusC 
  \move (-0.866025403784439 -.5)
  \RhombusC
  \move (-0.866025403784439 -1.5)
  \RhombusC
  \move (0.866025403784439 -2.5)
  \RhombusC \RhombusC 
  \move (0.866025403784439 -3.5)
  \RhombusC \RhombusC \RhombusC 
  \move (2.598076211353316 -5.5)
  \RhombusC 
  \move (0.866025403784439 -5.5)
  \RhombusC 
  \move (-1.732050807568877 -3)
  \RhombusC 
  \move (-1.732050807568877 -4)
  \RhombusC 
  \move (-1.732050807568877 -5)
  \RhombusC \RhombusC 
\esegment
\htext (-1.5 -9){\small a. A hexagon with sides $a,b,c,a,b,c$,}
\htext (-1.5 -9.5){\small \hphantom{a. }where $a=3$, $b=4$, $c=5$}
\htext (6.8 -9){\small b. A rhombus tiling of a hexagon}
\htext (6.8 -9.5){\small \hphantom{b. }with sides $a,b,c,a,b,c$}
\rtext td:0 (4.3 -4){$\sideset {}  c 
    {\left.\vbox{\vskip2.2cm}\right\}}$}
\rtext td:60 (2.6 -.5){$\sideset {} {} 
    {\left.\vbox{\vskip1.7cm}\right\}}$}
\rtext td:120 (-.44 -.25){$\sideset {}  {}  
    {\left.\vbox{\vskip1.3cm}\right\}}$}
\rtext td:0 (-2.3 -3.6){$\sideset {c}  {} 
    {\left\{\vbox{\vskip2.2cm}\right.}$}
\rtext td:240 (0 -7){$\sideset {}  {}  
    {\left.\vbox{\vskip1.7cm}\right\}}$}
\rtext td:300 (3.03 -7.25){$\sideset {}  {}  
    {\left.\vbox{\vskip1.4cm}\right\}}$}
\htext (-.9 0.1){$a$}
\htext (2.8 -.1){$b$}
\htext (3.2 -7.8){$a$}
\htext (-0.3 -7.65){$b$}
%}
}
%\vskip10pt
\caption{}
\label{fig:0}
\end{figure}

As is well-known, the total number of rhombus tilings of a hexagon
with side lengths $a,b,c,a,b,c$ equals
\begin{equation}
\label{eq:MacMahon}
\prod_{i=1}^a\prod_{j=1}^b\prod_{k=1}^c\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
(This follows from MacMahon's enumeration 
\cite[Sec.~429, $q\rightarrow 1$; proof in Sec.~494]{MacMahon}
of all plane partitions contained in an $a\times b\times c$ box,
as these are in bijection with rhombus tilings of a hexagon
with side lengths $a,b,c,a,b,c$, as explained e.g\@. in \cite{DT}.)

A natural question to be asked is what the distribution of the
rhombi in a random tiling is. 
On an {\it asymptotic} level, this question was answered by Cohn, 
Larsen and Propp \cite{CoLPAA}. On the {\it exact\/} (enumerative) level, 
the significant contributions are \cite{CiucKratAB,FuKrAC,HeGeAA}.
The most general result was obtained in \cite{FuKrAC} by the authors, where
the number of all rhombus
tilings of a hexagon with side lengths $N,M,N,N,M,N$ was computed,
which contain an arbitrary fixed rhombus
on the symmetry axis which cuts through the sides of length $M$.

The purpose of this paper is to add other results in this direction.
We compute 
the number of all rhombus tilings of a hexagon 
which contain a particular rhombus on the ``other" symmetry axis,
i.e., the symmetry axis which runs {\em in parallel\/} to the sides
of length $M$. In difference to \cite{FuKrAC}, we are not able to
solve this problem for an {\em arbitrary} rhombus on this symmetry
axis, but only for rhombi which are close to the center. In fact, the
case of the central rhombus is already covered by the papers 
\cite{CiucKratAB,FuKrAC,HeGeAA}. We provide results for the ``next"
three cases, i.e., for a rhombus which is by one, two, or three
``half units" off the center, see
Theorems~\ref{thm:MEven}--\ref{thm:NOdd2} below. In fact, as our proofs
show, the computations become increasingly harder, and more
elaborate, as we move the
rhombus gradually farther away from the center, so that it seems highly
unlikely that a uniform formula, similar to the one in \cite{FuKrAC}
for the other symmetry axis,
can be found for an arbitrary rhombus. (For further comments
on this issue see Section~\ref{sec:open}, (2).)

Here are our results.
(Each of the following formulas has to be interpreted as the
appropriate limit if singularities are encountered. For example, if
we directly set $m=1$ in \eqref{eq:NEven2} then the term $(m-1)$
in the second line becomes 0, and, on the other hand, we have a
singularity in the sum in the third line caused by the term $(3-n-m)_h$
in the summand for $h=n-1$. The correct way to interpret the
expression is as the limit as $m$ goes to 1.)

\begin{figure}
\bigskip
\centertexdraw{
% \move(0 0)
% \bsegment
  \drawdim truecm  \linewd.02
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-2.598076211353316 -1.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-2.598076211353316 -1.5)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-2.598076211353316 -2.5)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus
  \move (0.866025403784439 -1.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % symmetry axis g
  \linewd.05
  \move (0.866025403784439 -5.75)
  \rlvec(0 6.5)
% \esegment
\move(8 0)
\bsegment
  \drawdim truecm  \linewd.02
  \rhombus \rhombus \rhombus \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rdreieck \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-1.732050807568877 -2)
  \rdreieck \rhombus \rhombus \rhombus \rhombus
  \move (-1.732050807568877 -3)
  \rdreieck \rhombus \rhombus \rhombus
  \move (0.866025403784439 -1.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % symmetry axis g
  \linewd.05
  \move (0.866025403784439 -5.75)
  \rlvec(0 6.5)
\esegment
\htext (-2.5 -6.4){\small a.}
\htext (-2 -6.5){\small A hexagon with sides $N,M,N,$}
\htext (-2 -7){\small $N,M,N$, where $N=4$, $M=2$.}
\htext (5.3 -6.4){\small b.}
\htext (5.8 -6.5){\small A hexagon with sides $N,M,N,$}
\htext (5.8 -7){\small $N,M,N$, where $N=3$, $M=3$.}
\htext (-1.4 -0.1){$N$}
\htext (2.9 -0.1){$N$}
\htext (4.5 -2.6){$M$}
\htext (7.1 0.1){$N$}
\htext (10.4 0.1){$N$}
\htext (5.7 -2.6){$M$}
}
\caption{}
\label{fig:1}
\end{figure}

\begin{thm}
\label{thm:MEven}
Let $n$ and $m$ be positive integers. The number
of rhombus tilings of a hexagon with side lengths $2n,2m,2n,2n,2m,2n$, which
contain the rhombus above and next to the center of the hexagon
{\em (}see
Figure~\ref{fig:1}.a; the rhombus which is contained in every
tiling is shaded{\em)}, equals
\begin{multline}
\label{eq:MEven}
\frac{
n m \binom{2n}{n}\binom{2n-1}{n}\binom{2m}{m}
}{
\binom{4n+2m-1}{2n+m}
}
\Biggl(
	-\frac{1}{(n+m)^2}
	+\frac{4n+2}{(n+1)(2n-1)(n+m-1)(n+m+1)}\\
	\cdot\sum_{h=0}^{n-1}
		\frac{
			\po{2}{h}\,\po{1-n}{h}\,\po{\frac {3} {2}+n}{h}\,\po{1-n-m}{h}\,\po{1+n+m}{h}
		}{
			\po{1}{h}\,\po{2+n}{h}\,\po{\frac {3} {2}-n}{h}\,\po{2+n+m}{h}\,\po{2-n-m}{h}
		}
\Biggr)\\
\times
\prod_{i=1}^{2n}\prod_{j=1}^{2m}\prod_{k=1}^{2n}\frac{i+j+k-1}{i+j+k-2}.
\end{multline}
where the shifted factorial
$(a)_k$ is defined by $(a)_k:=a(a+1)\cdots(a+k-1)$,
$k\ge1$, $(a)_0:=1$.
\end{thm}

\begin{thm}
\label{thm:MOdd}
Let $n$ be a nonnegative integer and $m$ be a positive integer. The number
of rhombus tilings of a hexagon with side lengths
$2n+1,2m-1,2n+1,2n+1,2m-1,2n+1$, which
contain the rhombus above and next to the center of the hexagon
{\em (}see
Figure~\ref{fig:1}.b; the rhombus which is contained in every
tiling is shaded{\em)}, equals
\begin{multline}
\label{eq:MOdd}
\frac{
(n+1) m \binom{2n}{n} \binom{2n+1}{n} \binom{2m-1}{m}
}{
\binom{4n+2m}{2n+m}
}
\Biggl(
	\frac{1}{(n+m)^2}
	+\frac{4n}{(n+1)(2n-1)(n+m-1)(n+m+1)}\\
\cdot	\sum_{h=0}^{n-1}
		\frac{
			\po{2}{h}\,\po{1-n}{h}\,\po{\frac {3} {2}+n}{h}\,\po{1-n-m}{h}\,\po{1+n+m}{h}
		}{
			\po{1}{h}\,\po{2+n}{h}\,\po{\frac {3} {2}-n}{h}\,\po{2+n+m}{h}\,\po{2-n-m}{h}
		}
\Biggr)\\
\times
\prod_{i=1}^{2n+1}\prod_{j=1}^{2m-1}\prod_{k=1}^{2n+1}\frac{i+j+k-1}{i+j+k-2}.
\end{multline}
\end{thm}

\begin{thm}
\label{thm:NEven}
Let $n$ and $m$ be positive integers. The number
of rhombus tilings of a hexagon with side lengths $2n,2m-1,2n,2n,2m-1,2n$,
which contain the rhombus above and next to the central rhombus
(see Figure~\ref{fig:1a}.a; the rhombus which is contained in
every tiling is shaded), equals
\begin{multline}
\label{eq:NEven}
\frac{
	(2 m-1) \binom{2 m-2}{m-1} \binom{2 n-4}{n-2} 
	\binom{2 n+2}{n+1}
}{
	\left(n + m - 1\right)  
	\left(n + m \right)  \binom{4 n + 2 m - 3}{2 n + m - 2}
}
\\
\times
\Biggl(
	\frac{
		n\,\left(n + 1\right)\,\left(2 n - 3\right) \,
		\left(2 n - 1\right) \,
		\left({m^2} - m + n + 2 m\,n + {n^2} + 1\right)
	}{
		(n - 1)\,\left(n + m - 1\right) \,\left(n + m \right)
		\,\left(2 n + 1\right)
	}  \kern3.5cm\\
       + \frac {6} {(n+m-2)(n+m+1)}
		\sum_{h = 0}^{n - 2}
			\frac{
				\po{3}{h}\,
				\po{{\frac{5}{2}}}{h}\,
				\po{2 - n}{h}\,
				\po{{\frac{3}{2}} + n}{h}\,
				\po{2 - n - m}{h}\,
				\po{1 + n + m}{h}
			}{
				\po{1}{h}\,
				\po{{\frac{3}{2}}}{h}\,
				\po{2 + n}{h}\,
				\po{{\frac{5}{2}} - n}{h}\,
				\po{2 + n + m}{h}\,
				\po{3 - n - m}{h}
			}
\Biggr)\\
\times
\prod_{i=1}^{2n}\prod_{j=1}^{2m-1}\prod_{k=1}^{2n}\frac{i+j+k-1}{i+j+k-2}.
\end{multline}

\end{thm}

\begin{thm}
\label{thm:NOdd}
Let $n$ and $m$ be positive integers. The number
of rhombus tilings of a hexagon with side lengths
$2n-1,2m,2n-1,2n-1,2m,2n-1$, which
contain the rhombus above and next to the central rhombus 
(see Figure~\ref{fig:1a}.b; the rhombus which is contained in
every tiling is shaded), equals
\begin{multline}
\label{eq:NOdd}
\frac{
	(2 m - 1) \binom{2 m - 2}{m - 1} \binom{2 n - 4}{n - 2} 
	\binom{2 n + 2}{n + 1}
}{
	\left(n + m - 1\right)  
	\left( n + m \right)  \binom{4 n + 2 m - 3}{2 n + m - 2}
}
\\
\times
\Biggl(
	\frac{
		n\,\left(n + 1\right)\,\left(2 n - 3\right) \,
		\left(2 n - 1\right) \,
		\left({m^2} - m - 3n + 2 m\,n + {n^2} + 2\right)
	}{
		(n - 1)\,\left(n + m - 1\right) \,\left(n + m \right)
		\,\left(2 n + 1\right)
	}  \kern3.5cm\\
      +  \frac {6} {(n+m-2)(n+m+1)}
		\sum_{h = 0}^{n - 2}
			\frac{
				\po{3}{h}\,
				\po{{\frac{5}{2}}}{h}\,
				\po{2 - n}{h}\,
				\po{{\frac{3}{2}} + n}{h}\,
				\po{2 - n - m}{h}\,
				\po{1 + n + m}{h}
			}{
				\po{1}{h}\,
				\po{{\frac{3}{2}}}{h}\,
				\po{2 + n}{h}\,
				\po{{\frac{5}{2}} - n}{h}\,
				\po{2 + n + m}{h}\,
				\po{3 - n - m}{h}
			}
\Biggr)\\
\times
\prod_{i=1}^{2n-1}\prod_{j=1}^{2m}\prod_{k=1}^{2n-1}\frac{i+j+k-1}{i+j+k-2}.
\end{multline}
\end{thm}

\begin{figure}
\bigskip
\centertexdraw{
% Linkes Bild
  % Einstellungen
  \drawdim truecm  \linewd.02
  % Streifen, die schraeg nach rechts unten verlaufen; oben beginnend:
  % Erster Streifen
  \rhombus \rhombus \ldreieck
  % Zweiter Streifen
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \ldreieck
  % \move (-1.732050807568877 -1)
  % \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  % Dritter Streifen
  % \move (-1.732050807568877 -1)
  \move (-0.866025403784439 -.5)
  \rdreieck \rhombus \rhombus \rhombus \ldreieck
  % Vierter Streifen
  \move (-0.866025403784439 -1.5)
  \rdreieck \rhombus \rhombus \rhombus
  % Fvnfter Streifen
  \move (-0.866025403784439  -2.5)
  \rdreieck \rhombus \rhombus
  % Markierter Rhombus
  \move (0.866025403784439 -.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % Symmetry axis g
  \linewd.05
  \move (0.866025403784439 -4.75)
  \rlvec(0 5.5)
% Rechtes Bild
\move(8 0)
\bsegment
  % Einstellungen
  \drawdim truecm  \linewd.02
  % Streifen, die schraeg nach rechts unten verlaufen; oben beginnend:
  % Erster Streifen
  \rhombus \rhombus \rhombus \ldreieck
  % Zweiter Streifen
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  % Dritter Streifen
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus
  % Vierter Streifen
  \move (-1.732050807568877 -1)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus
  % Fvnfter Streifen
  \move (-1.732050807568877 -2)
  \rdreieck
  \rhombus \rhombus \rhombus
  % Markierter Rhombus
  \move (0.866025403784439 -.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % Symmetry axis g
  \linewd.05
  \move (0.866025403784439 -4.75)
  \rlvec(0 5.5)
\esegment
\htext (-2.5 -5.4){\small a.}
\htext (-2 -5.5){\small A hexagon with sides $N,M,N,$}
\htext (-2 -6){\small $N,M,N$, where $N=2$, $M=3$.}
\htext (5.3 -5.4){\small b.}
\htext (5.8 -5.5){\small A hexagon with sides $N,M,N,$}
\htext (5.8 -6){\small $N,M,N$, where $N=3$, $M=2$.}
% \htext (-1.1 -0.2){$N$}
% \htext (2.4 -0.2){$N$}
% \htext (3.8 -2.1){$M$}
% \htext (7.5 0.1){$N$}
% \htext (9.9 0.1){$N$}
% \htext (6.4 -2.1){$M$}
%
\htext (-0.6 0.1){$N$}
\htext (1.9 0.1){$N$}
\htext (3.0 -2.1){$M$}
\htext (7.0 -0.2){$N$}
\htext (10.4 -0.2){$N$}
\htext (5.5 -2.1){$M$}
}
\caption{}
\label{fig:1a}
\end{figure}

\begin{thm}
\label{thm:NEven2}
Let $n$ and $m$ be positive integers. The number
of rhombus tilings of a hexagon with side lengths $2n,2m,2n,2n,2m,2n$,
which contain the rhombus above and next to the rhombus which is adjacent
to the center of the hexagon
(see Figure~\ref{fig:1b}.a; the rhombus which is contained in
every tiling is shaded), equals
\begin{multline}
\label{eq:NEven2}
\frac{
	\binom{n+m-1}{m} 
	\binom{2n+2}{n-1}
	\binom{2n+m-1}{n}
	\binom{2n+m}{2n+1}
}{
	2(2n-3)(2n-1)(2n+2)(n+m-1)(n+m+1)
	\binom{4n-1}{2n}
	\binom{4n+2m-1}{2m}
}
\\
\times\Biggl( 
{\frac {2   \left( n+2 \right)  \left( n+3 \right)  
     \left( 2 n -1\right) \left( 2 n-3 \right) X(m,n) } 
   {\left( n+m -1\right)  {{\left( n+m \right) }^2} 
     \left( n+m+1 \right) }}
	-
	\frac{
		24(m-1)(2n+m+1)(2n+1)(2n+3)
	}{
		(n+m-2)(n+m+2)
	}\\
	\cdot\sum_{h = 0}^{n - 1}
			\frac{
				\po{4}{h}\,
				\po{1 - n}{h}\,
				\po{{\frac{5}{2}} + n}{h}\,
				\po{2 - n - m}{h}\,
				\po{2 + n + m}{h}
			}{
				\po{1}{h}\,
				\po{4 + n}{h}\,
				\po{{\frac{5}{2}} - n}{h}\,
				\po{3 + n + m}{h}\,
				\po{3 - n - m}{h}
			}
\Biggr)
\prod_{i=1}^{2n}\prod_{j=1}^{2m}\prod_{k=1}^{2n}\frac{i+j+k-1}{i+j+k-2},
\end{multline}
where 

\vskip-10pt
{\small
\begin{multline*} 
X(m,n)= -1 + 4 {m^2} - 3 {m^4} - n + 8 m n + 11 {m^2} n - 12 {m^3} n - 
  6 {m^4} n + 8 {n^2} + 22 m {n^2} - 4 {m^2} {n^2} - 
  24 {m^3} {n^2}\\
 - 2 {m^4} {n^2} + 15 {n^3} + 16 m {n^3} - 
  28 {m^2} {n^3} - 8 {m^3} {n^3} - 5 {n^4} - 8 m {n^4} - 
  12 {m^2} {n^4} - 14 {n^5} - 8 m {n^5} - 2 {n^6}.
\end{multline*}
}
\end{thm}

\begin{thm}
\label{thm:NOdd2}
Let $n$ and $m$ be positive integers. The number
of rhombus tilings of a hexagon with side lengths
$2n+1,2m-1,2n+1,2n+1,2m-1,2n+1$, which
contain the rhombus above and next to the rhombus which is adjacent
to the center of the hexagon
(see Figure~\ref{fig:1b}.b; the rhombus which is contained in
every tiling is shaded), equals
\begin{multline}
\label{eq:NOdd2}
\frac{
	\binom{2m-1}{m}
	\binom{n+m-1}{m-1}
	\binom{2n+2}{n-1}
	\binom{2n+m}{n+1}
}{
	(m+1)(m+2)(2n-3)(n+m-1)(n+m+1)
	\binom{2n+m}{m+2}
	\binom{4n+2m}{2n+m}
}
\\
\times\Biggl(
{\frac { \left( n+2 \right)  \left( n+3 \right)  
     \left( 2 n-1 \right) \left( 2 n-3 \right) Y(m,n) }
   {\left( n + m-1 \right)  {{\left( n + m \right) }^2} 
     \left( n + m+1 \right) }}
	-
	\frac{
		24(m-1)n(2n+m+1)(2n+3)
	}{
		(n+m-2)(n+m+2)
	}\\
	\cdot\sum_{h = 0}^{n - 1}
			\frac{
				\po{4}{h}\,
				\po{1 - n}{h}\,
				\po{{\frac{5}{2}} + n}{h}\,
				\po{2 - n - m}{h}\,
				\po{2 + n + m}{h}
			}{
				\po{1}{h}\,
				\po{4 + n}{h}\,
				\po{{\frac{5}{2}} - n}{h}\,
				\po{3 + n + m}{h}\,
				\po{3 - n - m}{h}
			}
\Biggr)
\prod_{i=1}^{2n+1}\prod_{j=1}^{2m-1}\prod_{k=1}^{2n+1}\frac{i+j+k-1}{i+j+k-2},
\end{multline}
where 

\vskip-10pt
{\small
\begin{multline*} 
Y(m,n)= 1 - 2 {m^2} + {m^4} + 5 n - 4 m n - 2 {m^2} n + 
       4 {m^3} n - 3 {m^4} n + 22 {n^2} - 4 m {n^2} - 
       4 {m^2} {n^2} \\
- 12 {m^3} {n^2} - 2 {m^4} {n^2} + 50 {n^3} - 
       16 m {n^3} - 26 {m^2} {n^3} - 8 {m^3} {n^3} + 39 {n^4} - 
       28 m {n^4} - 12 {m^2} {n^4} + 5 {n^5} - 8 m {n^5} - 2 {n^6}.
\end{multline*}
}
\end{thm}

\begin{figure}
\caption{}
\label{fig:1b}
\bigskip
\centertexdraw{
% \move(0 0)
% \bsegment
  \drawdim truecm  \linewd.02
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-2.598076211353316 -1.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-2.598076211353316 -1.5)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-2.598076211353316 -2.5)
  \rdreieck
  \rhombus \rhombus \rhombus \rhombus
  \move (0.866025403784439 -0.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % symmetry axis g
  \linewd.05
  \move (0.866025403784439 -5.75)
  \rlvec(0 6.5)
% \esegment
\move(8 0)
\bsegment
  \drawdim truecm  \linewd.02
  \rhombus \rhombus \rhombus \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  \move (-1.732050807568877 -1)
  \rdreieck \rhombus \rhombus \rhombus \rhombus \rhombus
  \move (-1.732050807568877 -2)
  \rdreieck \rhombus \rhombus \rhombus \rhombus
  \move (-1.732050807568877 -3)
  \rdreieck \rhombus \rhombus \rhombus
  \move (0.866025403784439 -0.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % symmetry axis g
  \linewd.05
  \move (0.866025403784439 -5.75)
  \rlvec(0 6.5)
\esegment
\htext (-2.5 -6.4){\small a.}
\htext (-2 -6.5){\small A hexagon with sides $N,M,N,$}
\htext (-2 -7){\small $N,M,N$, where $N=4$, $M=2$.}
\htext (5.3 -6.4){\small b.}
\htext (5.8 -6.5){\small A hexagon with sides $N,M,N,$}
\htext (5.8 -7){\small $N,M,N$, where $N=3$, $M=3$.}
\htext (-1.4 -0.1){$N$}
\htext (2.9 -0.1){$N$}
\htext (4.5 -2.6){$M$}
\htext (7.1 0.1){$N$}
\htext (10.4 0.1){$N$}
\htext (5.7 -2.6){$M$}
}
\end{figure}



In general, the sum in \eqref{eq:MEven} and \eqref{eq:MOdd} (note
that it is indeed exactly the same sum), the sum in 
\eqref{eq:NEven} and \eqref{eq:NOdd} 
(it is indeed exactly the same sum), and the sum in 
\eqref{eq:NEven2} and \eqref{eq:NOdd2} (again, 
it is indeed exactly the same sum), does not simplify. 
However, in the case that $n$ and $m$ are roughly of the same size,
the sum does simplify. For the sake of brevity, we give here just a
sample of corollaries to Theorems~\ref{thm:MEven} and \ref{thm:MOdd},
the first two statements of which solve the second part of Problem~4 in
Propp's list \cite{PropAA}. We wish to emphasize that there are
similar corollaries to Theorems~\ref{thm:NEven}--\ref{thm:NOdd2}.

\begin{cor}\label{thm:Propp}
Let $n$ be a nonnegative integer. The number of 
rhombus tilings of a hexagon with all sides of length $2n$,
which contain the rhombus above and next to the center of the hexagon, equals
\begin{equation}
\label{eq:ProppEven}
\left(
	\frac{1}{3}-
	\frac{1}{12}
	\frac{
		\binom{2n}{n}^3
	}{
		\binom{6n}{3n}
	}
\right)
\prod_{i=1}^{2n}\prod_{j=1}^{2n}\prod_{k=1}^{2n}\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
The number of 
rhombus tilings of a hexagon with all sides of length $2n+1$,
which contain the rhombus above and next to the center of the hexagon, equals
\begin{equation}
\label{eq:ProppOdd}
\left(
	\frac{1}{3}+
	\frac{1}{3}
	\frac{
		\binom{2n}{n}^3
	}{
		\binom{6n+2}{3n+1}
	}
\right)
\prod_{i=1}^{2n+1}\prod_{j=1}^{2n+1}\prod_{k=1}^{2n+1}\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
The number of 
rhombus tilings of a hexagon with side lengths $2n,2n+2,2n,2n,2n+2,2n$,
which contain the rhombus above and next to the center of the hexagon, equals
\begin{equation}
\label{eq:ProppEven1}
\left(
	\frac{1}{3}-
	\frac{(10n+2)}{(6n+3)}
	\frac{
		\binom{2n}{n}^3
	}{
		\binom{6n+2}{3n+1}
	}
\right)
\prod_{i=1}^{2n}\prod_{j=1}^{2n+2}\prod_{k=1}^{2n}\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
The number of 
rhombus tilings of a hexagon with side lengths $2n+1,2n-1,2n+1,2n+1,2n-1,2n+1$,
which contain the rhombus above and next to the center of the hexagon, equals
\begin{equation}
\label{eq:ProppOdd1}
\left(
	\frac{1}{3}+
	\frac{(10n+3)}{24n}
	\frac{
		\binom{2n}{n}^3
	}{
		\binom{6n}{3n}
	}
\right)
\prod_{i=1}^{2n+1}\prod_{j=1}^{2n-1}\prod_{k=1}^{2n+1}\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
The number of 
rhombus tilings of a hexagon with side lengths $2n+2,2n,2n+2,2n+2,2n,2n+2$,
which contain the rhombus above and next to the center of the hexagon, equals
\begin{equation}
\label{eq:ProppEven2}
\left(
	\frac{1}{3}+
	4\frac{
		\binom{2n}{n}^3
	}{
		\binom{6n+4}{3n+2}
	}
\right)
\prod_{i=1}^{2n+2}\prod_{j=1}^{2n}\prod_{k=1}^{2n+2}\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
The number of 
rhombus tilings of a hexagon with side lengths $2n+3,2n-1,2n+3,2n+3,2n-1,2n+3$,
which contain the rhombus above and next to the center of the hexagon, equals
\begin{equation}
\label{eq:ProppOdd2}
\left(
	\frac{1}{3}+
	\frac{2(6n^2+9n+2)}{(n+1)^2}
	\frac{
		\binom{2n}{n}^3
	}{
		\binom{6n+4}{3n+2}
	}
\right)
\prod_{i=1}^{2n+3}\prod_{j=1}^{2n-1}\prod_{k=1}^{2n+3}\frac{i+j+k-1}{i+j+k-2}.
\end{equation}
\end{cor}

If $m$ is not $n+1$, $n$ or $n-1$, then one may still try to obtain at
least estimates for the number of rhombus tilings that contain this
particular rhombus.
Indeed,
from Theorems~\ref{thm:MEven}--\ref{thm:NOdd2}, we are able to derive
an ``arcsine law" for this kind of enumeration, which is analogous
to the ones in \cite{CiucKratAB} and \cite{FuKrAC}.

\begin{cor}\label{thm:asy}
Let $a$ be any nonnegative real number, and consider a hexagon with
side lengths $2n,2m,2n,2n,2m,2n$.
For $m\sim an$, the proportion
of the rhombus tilings that contain the
rhombus above and next to the center of the hexagon
in the total number of rhombus tilings is
$\sim\ \frac {2} {\pi}\arcsin(1/(a+1))$
as $n$ tends to infinity.
The same is true for a
hexagon with side lengths $2n+1,2m-1,2n+1,2n+1,2m-1,2n+1$. Moreover,
for hexagons 
with side lengths $2n,2m-1,2n,2n,2m-1,2n$ or hexagons
with side lengths $2n-1,2m,2n-1,2n-1,2m,2n-1$,
the same result holds for the proportion 
of the rhombus tilings that contain 
the rhombus above and next to the central rhombus
in the total number of rhombus tilings, as well as 
for the proportion of the rhombus tilings that contain 
the rhombus above and next to the rhombus which is adjacent to the
center of the hexagon
in the total number of rhombus tilings of hexagons 
with side lengths $2n,2m,2n,2n,2m,2n$ or hexagons
with side lengths $2n+1,2m-1,2n+1,2n+1,2m-1,2n+1$.
\end{cor}
Also this result is (as well as Corollary~4 in \cite{CiucKratAB}
and Theorem~1.3 in \cite{FuKrAC}; see the
respective comments in \cite{FuKrAC}) 
in accordance with Conjecture~1 in
\cite{CoLPAA}, to which it adds evidence in further special instances.

In the next sections we describe proofs of 
Theorems~\ref{thm:MEven}--\ref{thm:NOdd2},
and of 
Corollaries~\ref{thm:Propp} and \ref{thm:asy}.
In Section~2 we provide proofs of Corollaries~\ref{thm:Propp} and
\ref{thm:asy},
and we outline the proofs of
Theorems~\ref{thm:MEven}--\ref{thm:NOdd2},
the latter consisting 
of two basic steps. In the first step we
build on the approach of Helfgott and Gessel in 
\cite{HeGeAA}, a short summary of which is the contents of Section~3. 
It allows to write the
number that we are interested in 
in form of a determinant. The evaluation
of this determinant is not easy and is carried out in detail in
Section~4. For the evaluation we follow a ``method" that was
first introduced in \cite{KratBD} (see the tutorial description in
\cite[Sec.~2.4]{KratBN} or \cite[Sec.~2]{KratBI}). 
For accomplishing the required computations, we need
to evaluate certain Hankel determinants featuring Bernoulli numbers,
which are, in fact, of independent interest.
As it turns out, some of the evaluations of these Hankel determinants are
already known, provided certain results about
orthogonal polynomials, in particular, about
continuous Hahn polynomials, and continued fractions are properly combined. 
For the convenience of the reader,
we collect these facts, and their implications, in Section~5.
In particular, the evaluation of the relevant Hankel determinants
featuring Bernoulli numbers is given in
Theorem~\ref{thm:HankelBernoulli}.
However, in the proofs of Theorems~\ref{thm:NEven2} and \ref{thm:NOdd2}
(more precisely, in the proof of the subordinate Lemma~\ref{lem:Det3})
we encounter a certain Hankel determinant of Bernoulli numbers
(see \eqref{eq:extra-Bernoulli}), the evaluation of which requires
considerable effort. (This is one of the added difficulties mentioned
earlier in comparison to the proofs of Theorems~\ref{thm:MEven} and
\ref{thm:MOdd}.) We evaluate this Hankel determinant by combining the
knowledge about continuous Hahn polynomials with a recent theorem on
orthogonal polynomials due to Leclerc \cite{Leclerc} (restated here
as Theorem~\ref{thm:Leclerc}), and applying some integral calculus
(see the proof of Lemma~\ref{lem:weird}). Section~6 is devoted to
provide the details of these calculations. In
Section~\ref{sec:further} we make
explicit a few unusual evaluations of Hankel determinants of
Bernoulli polynomials, which are implicit in the proofs
of our enumeration results.
Finally, in Section~\ref{sec:open}, we point to further directions in this
research, and propose a few open problems.




%%%@@@ end input from intro


%%%@@@ input from outline
\section{Outline of proofs}
Here we outline the proofs of Theorems~\ref{thm:MEven}--\ref{thm:NOdd2},
and we deduce Corollaries~\ref{thm:Propp} and \ref{thm:asy}. 
We fill in the details in the subsequent sections.

\smallskip
{\sc Proof of Theorems~\ref{thm:MEven}--\ref{thm:NOdd2}}.
Following the approach of Helfgott and Gessel \cite{HeGeAA} (see
Section~3), we may
write the number of rhombus tilings of a hexagon with side lengths
$N,M,N,N,M,N$, which
contain an {\em arbitrary\/} rhombus on the $(N+M)$-long vertical symmetry
axis (see Figure~\ref{figure-thm6}),
% above and next to the center of the hexagon,
in form of a determinant. This determinant is given by
Proposition~\ref{thm:HeGe-det}.
% with $l=(N+M)/2$. 
That is to say, in
order to prove Theorems~\ref{thm:MEven} and \ref{thm:MOdd}, we need
to evaluate the determinant in \eqref{eq:HeGe-det} with
$l=\frac{N+M}{2}$, and in order to prove Theorems~\ref{thm:NEven} and \ref{thm:NOdd}, we need to evaluate the same determinant with
$l=\frac{N+M+1}{2}$, and in order to prove Theorems~\ref{thm:NEven2} and \ref{thm:NOdd2}, we need to evaluate the same determinant with
$l=\frac{N+M+2}{2}$. Modulo replacement of parameters, we may thus
concentrate on the determinants $D(n,n,N)$, $D(n,n-1,N)$ and
$D(n,n-2,N)$, where
\begin{equation}
\label{eq:gen-det}
D(a,b,N):=\det_{1\le i,j\le  N}\bigg(\sum_{s=-a}^{b-1} s^{i+j}\bigg).
\end{equation}

The evaluation of 
determinants $D(n,n,N)$, $D(n,n-1,N)$ and $D(n,n-2,N)$ is then carried out in
Lemmas~\ref{lem:Det1}, \ref{lem:Det2} and \ref{lem:Det3}, respectively.

Theorem~\ref{thm:MEven} follows upon combining
Proposition~\ref{thm:HeGe-det} with $N=2n$, $M=2m$, $l=n+m$ and 
Lemma~\ref{lem:Det1} with
$n$ replaced by $n+m$ and $N=2n-1$, and some rearrangement of terms.
Similarly, Theorem~\ref{thm:MOdd} follows upon combining
Proposition~\ref{thm:HeGe-det} with $N=2n+1$, $M=2m-1$, $l=n+m$ and 
Lemma~\ref{lem:Det1} with $n$ replaced by $n+m$ and $N=2n$.

Likewise, Theorem~\ref{thm:NEven} follows upon combining
Proposition~\ref{thm:HeGe-det} with $N=2n$, $M=2m-1$, $l=n+m$ and 
\eqref{eq:Det2odd} with
$n$ replaced by $n+m$ and $m$ replaced by $n$, and
Theorem~\ref{thm:NOdd} follows upon combining
Proposition~\ref{thm:HeGe-det} with $N=2n-1$, $M=2m$, $l=n+m$ and 
\eqref{eq:Det2even} with $n$ replaced by $n+m$ and $m$ replaced by $n-1$.

Likewise, Theorem~\ref{thm:NEven2} follows upon combining
Proposition~\ref{thm:HeGe-det} with $N=2n$, $M=2m$, $l=n+m+1$ and 
\eqref{eq:Det3odd} with
$n$ replaced by $n+m+1$ and $m$ replaced by $n-1$, and
Theorem~\ref{thm:NOdd2} follows upon combining
Proposition~\ref{thm:HeGe-det} with $N=2n+1$, $M=2m-1$, $l=n+m+1$ and 
\eqref{eq:Det3even} with $n$ replaced by $n+m+1$ and $m$ replaced by $n$.
\hfil\qed

\bigskip
{\sc Proof of Corollary~\ref{thm:Propp}}.
We have to compute the value of the expressions \eqref{eq:MEven}
and \eqref{eq:MOdd} for $m=n+1$ (in order to establish
\eqref{eq:ProppEven1} and \eqref{eq:ProppOdd}), 
$m=n$ (in order to establish
\eqref{eq:ProppEven} and \eqref{eq:ProppOdd1}), 
and $m=n-1$ (in order to establish
\eqref{eq:ProppEven2} and \eqref{eq:ProppOdd2}).
% markus: Checke Gleichungsnummern!
Clearly, except for
trivial manipulations, we will be done once we are able to evaluate
the sum in \eqref{eq:MEven} and \eqref{eq:MOdd} (it is indeed the
same sum!) for $m=n+1$, $m=n$, respectively $m=n-1$.

We treat the case $m=n$ first.
We claim that 
\begin{equation} \label{eq:Summe}
	\sum_{h=0}^{n-1}
		\frac{
		\po{2}{h}\,\po{3/2+n}{h}\,\po{1-n}{h}\,\po{1+2n}{h}\,\po{1-2n}{h}
		}{
		\po{1}{h}\,\po{3/2-n}{h}\,\po{2+n}{h}\,\po{2-2n}{h}\,\po{2+2n}{h}
		}
=\frac {(n+1)(2n-1)^2} {2n^2}\left(\frac {1} {6}+\frac {1} {3}
\frac {\binom {6n}{3n}} {{\binom {2n}{n}}^3}\right).
\end{equation}
Let us denote the sum by $S(n)$ and its summand by $F(n,i)$.
We use the Gosper--Zeilberger algorithm \cite{PeWZAA,ZeilAM,ZeilAV}
to obtain the relation
\begin{multline} \label{eq:WZ}
6 n^2  (n+2) ( 6 n+1) ( 6 n+5) \,F(n,i) - 
     6 (n+1) (2 n-1)^2  (3 n+1) ( 3 n+2) \,F(n+1,i) \\
=G(n,i+1)-G(n,i),
\end{multline}
with
\begin{multline*} 
G(n,i)=\frac {n (n+2) (2n-2h-1) (2n-h-1) 
    }
  {(h+1) (h - n) (2n-h) (2n - h + 1) (2n + h + 2)}\\
\times(
 144 n^5
- 432 h^2 n^4 
- 432 h n^4 
+ 312 n^4 
- 936 h^2 n^3 
- 936 h n^3 
+ 236 n^3 
+ 108 h^4 n^2 \\
\quad \quad + 216 h^3 n^2 
- 588 h^2 n^2 
- 696 h n^2 
+ 70 n^2 
+ 117 h^4 n 
+ 234 h^3 n 
- 83 h^2 n 
- 200 h n \\
+ 6 n 
+ 24 h^4 
+ 48 h^3 
+ 6 h^2 
-18 h 
)F(n,i).
\end{multline*}
Summation of the relation \eqref{eq:WZ} from $i=0$ to $i=n-1$, and little
rearrangement, leads to the recurrence
\begin{multline*}
6 n^2  (n+2) ( 6 n+1) ( 6 n+5) \,S(n) - 
     6 (n+1) (2 n-1)^2  (3 n+1) ( 3 n+2) \,S(n+1) \\
=\frac{( n+2) (2 n-1)^2 (36 n^3 + 60 n^2 + 29 n + 3)}{2 (n+1)}.
\end{multline*}
for the sum in \eqref{eq:Summe}. (Paule and Schorn's \cite{PaScAA} {\sl Mathematica}
implementation of the Gosper--Zeilberger algorithm,
which is the one we used, gives this recurrence directly.) Since
$S(1)=1$, and since the right-hand side of \eqref{eq:Summe} satisfies the same
recurrence, equation \eqref{eq:Summe} is proved.

The procedure in the other two cases is analogous. The respective
evaluations that we need to prove are
\begin{equation} \label{eq:Summe1}
	\sum_{h=0}^{n-1}
		\frac{
		\po{2}{h}\,\po{3/2+n}{h}\,\po{1-n}{h}\,\po{2+2n}{h}\,\po{-2n}{h}
		}{
	\po{1}{h}\,\po{3/2-n}{h}\,\po{2+n}{h}\,\po{1-2n}{h}\,\po{3+2n}{h}
		}
=\frac {(n+1)^2(2n-1)} {(2n+1)^2}\left(-\frac {2} {3}+\frac {1} {3}
\frac {\binom {6n+2}{3n+1}} {{\binom {2n}{n}}^3}\right)
\end{equation}
in the case that $m=n+1$, and
\begin{equation} \label{eq:Summe2}
	\sum_{h=0}^{n-1}
		\frac{
	\po{2}{h}\,\po{3/2+n}{h}\,\po{1-n}{h}\,\po{2n}{h}\,\po{2-2n}{h}
		}{
	\po{1}{h}\,\po{3/2-n}{h}\,\po{2+n}{h}\,\po{3-2n}{h}\,\po{1+2n}{h}
		}
=\frac {2n(n+1)} {2n+1}\left({1}+\frac {n} {12(2n-1)}
\frac {\binom {6n-2}{3n-1}} {{\binom {2n-2}{n-1}}^3}\right)
\end{equation}
in the case that $m=n-1$. We leave it to the reader to fill in the
details.
\hfil\qed

\bigskip
{\sc Proof of Corollary~\ref{thm:asy}}.
We concentrate first on the case of a hexagon with side lengths 
$2n,2m,2n,2n,2m,2n$ and the rhombus above and next to the center 
of the hexagon. 

{}From MacMahon's formula \eqref{eq:MacMahon} for the total number of rhombus
tilings together with Theorem~\ref{thm:MEven} we infer
that the proportion of the rhombus tilings that contain the
rhombus above and next to the center of the hexagon with sides
lengths $2n,2m,2n,2n,2m,2n$
in the total number of rhombus tilings is given by
\begin{multline*}
\frac{
n m \binom{2n}{n}\binom{2n-1}{n}\binom{2m}{m}
}{
\binom{4n+2m-1}{2n+m}
}
\Biggl(
	-\frac{1}{(n+m)^2}
	+\frac{4n+2}{(n+1)(2n-1)(n+m-1)(n+m+1)}\\
	\times\sum_{h=0}^{n-1}
		\frac{
			\po{2}{h}\,\po{3/2+n}{h}\,\po{1-n}{h}\,\po{1+n+m}{h}\,\po{1-n-m}{h}
		}{
			\po{1}{h}\,\po{3/2-n}{h}\,\po{2+n}{h}\,\po{2-n-m}{h}\,\po{2+n+m}{h}
		}
\Biggr).
\end{multline*}
Using the standard hypergeometric notation
\begin{equation} \label{eq:hypergeom}
\HypF{r}{s}{a_1,\dots,a_r}{b_1,\dots,b_s}{z} =
	\sum_{k=0}^\infty
		\frac{
			\po{a_1}{k}\cdots\po{a_r}{k}
		}{
			k!\,\po{b_1}{k}\cdots\po{b_s}{k}
		}z^k,
\end{equation}
we may write the above expression as
\begin{multline} \label{eq:very-well-poised}
\frac{
n m \binom{2n}{n}\binom{2n-1}{n}\binom{2m}{m}
}{
\binom{4n+2m-1}{2n+m}
}
\Biggl(
	-\frac{1}{(n+m)^2}
	+\frac{4n+2}{(n+1)(2n-1)(n+m-1)(n+m+1)}\\
	\times
{}_7F_6\!\left[\begin{matrix} 2,2,1 + n + m, 1 - n - m, 1, {\frac 3 2} + n, 1 - n\\
1,2-n-m,2+n+m,2,{\frac 3 2}-n,2+n\end{matrix}; 1\right]
\Biggr).
\end{multline}
Now we observe that the sum in this expression is in fact a
very-well-poised hypergeometric series, and therefore we can transform it
into a more convenient form by applying 
Whipple's transformation (see \cite[(2.4.1.1)]{SlatAC})
\begin{multline}
\label{eq:Whipple}
\HypF{7}{6}{a,1+\frac{a}{2},b,c,d,e,-N}%
{\frac{a}{2},1+a-b,1+a-c,1+a-d,1+a-e,1+a+N}{1}\\
=\frac{\po{a+1}{N}\,\po{a-d-e+1}{N}}{\po{a-d+1}{N}\,\po{a-e+1}{N}}
\HypF{4}{3}{a-b-c+1,d,e,-N}{a-b+1,a-c+1,-a+d+e-N}{1}
\end{multline}
(where $N$ is a nonnegative integer)
with $a=2$, $b=1+n+m$, $c=1-n-m$, $d=1$, $e=3/2+n$, and $N=n-1$ to it.
Thus we obtain the expression
\begin{multline}\label{eq:4F3}
\frac{
n m \binom{2n}{n}\binom{2n-1}{n}\binom{2m}{m}
}{
\binom{4n+2m-1}{2n+m}
}
\Biggl(
	-\frac{1}{(n+m)^2}
	+\frac{2n+1}{(n+m-1)(n+m+1)}\\
	\times
   {} _{4} F _{3} \!\left [ \begin{matrix} { 1, 1, {\frac 3 2} + n, 1 - n}\\ 
   { 2 - n - m, 2 + n + m, {\frac 3 2}}\end{matrix} ; {\displaystyle 1}\right ] 
\Biggr).
\end{multline}
Now we substitute $m\sim a n$ and perform the limit
$n\rightarrow\infty$. The asymptotics of the binomials appearing in
front of the expression \eqref{eq:4F3} is easily determined by means of Stirling's
formula. For the $\HypFsimple{4}{3}$-series itself,
we may exchange limit and summation
by uniform convergence,
\begin{equation}\label{eq:2F1}
\lim_{n\rightarrow\infty}\HypF{4}{3}{1,1,\frac{3}{2}+n,1-n}%
{2-n-an,2+n+an,\frac{3}{2}}{1}=
\HypF{2}{1}{1,1}{\frac{3}{2}}{ {\frac {1} {(a+1)^2}} }.
\end{equation}
Combining all this, and making use of the identity 
(see \cite[p.~463, (133)]{Prudnikov})
\begin{equation}\label{eq:arcsin}
\HypF{2}{1}{1,1}{\frac{3}{2}}{z}=\frac{\arcsin\sqrt{z}}{\sqrt{z(1-z)}},
\end{equation}
we obtain exactly $\frac {2} {\pi}\arcsin (1/(a+1))$ as the
asymptotic form of \eqref{eq:4F3}, and, hence, as the asymptotic form
of the proportion of rhombus tilings in the statement of the
corollary, as desired.

\smallskip
The case of a hexagon with side lengths 
$2n+1,2m-1,2n+1,2n+1,2m-1,2n+1$ can be handled in (almost) the same
way because the sums in \eqref{eq:MEven} and \eqref{eq:MOdd} are
exactly the same.

\smallskip
For the next two cases, i.e., in order to estimate
\eqref{eq:NEven} and \eqref{eq:NOdd}, we proceed in a similar way.
Again we write the sum which appears in \eqref{eq:NEven} and
\eqref{eq:NOdd} as a $_7F_6$-series,
$${}_7F_6\!\left[\begin{matrix} 3,\frac {5} {2},1+n+m,2-n-m,2,\frac {3} {2}+n,
2-n\\\frac {3} {2},3-n-m,2+n+m,2,\frac {5} {2}-n,2+n\end{matrix}; 1\right],$$
apply Whipple's transformation \eqref{eq:Whipple}, and then let $n$ tend
to infinity. Here, the $_2F_1$-series which is obtained is a slightly
different one as before (compare \eqref{eq:2F1}),
$${}_2F_1\!\left[\begin{matrix} 1,2\\\frac {5} {2}\end{matrix}; \frac {1}
{(a+1)^2}\right].$$
In order to be able to use \eqref{eq:arcsin}, we use the relation
$${}_2F_1\!\left[\begin{matrix} 1,2\\\frac {5} {2}\end{matrix}; \frac {1}
{(a+1)^2}\right]=
-\frac {3(a+1)^2} {2} + 
  {\frac {{{3\left( a+1 \right) }^2}\,
 } 2}      {} _{2} F _{1} \!\left [ \begin{matrix} { 1, 1}\\
{{\frac 3 2}}\end{matrix} ; {\displaystyle \frac {1} {(a+1)^2}}\right
       ].
$$
The computation is then completed by straightforward use of Stirling's
formula, and subsequent simplification.

The remaining two cases, i.e., the estimations of \eqref{eq:NEven2}
and \eqref{eq:NOdd2}, can be dealt with in just the same manner.
The $_2F_1$-series which is obtained here is again slightly
different. It reads
$${}_2F_1\!\left[\begin{matrix} 1,2\\\frac {3} {2}\end{matrix}; \frac {1}
{(a+1)^2}\right]=
-{\frac {{{\left( a+1 \right) }^2}} 2} + 
  {\frac {{{\left( a+1 \right) }^2}
 } 2}
      {} _{2} F _{1} \!\left [ \begin{matrix} { 1, 1}\\ { {\frac 1
2}}\end{matrix} ;
       {\displaystyle \frac {1} {(a+1)^2}}\right ].
$$
To the $_2F_1$-series on the right-hand side 
we apply the formula (see \cite[p.~464, (138)]{Prudnikov})
$${} _{2} F _{1} \!\left [ \matrix { 1, 1}\\ { {\frac 1 2}}\endmatrix ;
   {\displaystyle z}\right ]  = 
  {\frac 1 {1 - z}} + {\frac {{\sqrt{z}}\,\arcsin {\sqrt{z}}} 
     {{{\left( 1 - z \right) }^{{\frac 3 2}}}}}.
$$
Again, the computation is then completed by straightforward use of Stirling's
formula, and subsequent simplification.
\hfil\qed



%%%@@@ end input from outline


%%%@@@ input from rhombdet
\section{From rhombus tilings to determinants}
\label{sec:rhombdet}

This section is entirely based on ideas by Helfgott and Gessel 
\cite{HeGeAA}.
These allow us to find a determinantal expression for the number 
of rhombus tilings
of a hexagon with side lengths $N,M,N,N,M,N$ that contain an {\em arbitrary\/}
fixed rhombus on the $(N+M)$-long symmetry axis (the vertical
symmetry axis in Figure~\ref{figure-thm6}). We shall state
two auxiliary results (Propositions~\ref{prop:HeGe1} and \ref{prop:HeGe}) 
without proof (the reader can find the details in
\cite{HeGeAA}), and then derive Helfgott and Gessel's determinant
(Proposition~\ref{thm:HeGe-det}). It is the
specialization $l=(N+M)/2$ of Proposition~\ref{thm:HeGe-det} (compare
the paragraph above \eqref{eq:gen-det}) which in the long run leads 
to a proof of our Theorems~\ref{thm:MEven} and \ref{thm:MOdd}, it is the
specialization $l=(N+M+1)/2$ of Proposition~\ref{thm:HeGe-det}
which in the long run leads 
to a proof of our Theorems~\ref{thm:NEven} and \ref{thm:NOdd}, and it is the
specialization $l=(N+M+2)/2$ of Proposition~\ref{thm:HeGe-det}
which in the long run leads 
to a proof of our Theorems~\ref{thm:NEven2} and \ref{thm:NOdd2}.
We do want to alert the reader that we use a different convention
in our figures of how to draw the hexagons than Helfgott and Gessel. 
To be precise, our figures turn into those in \cite{HeGeAA} by
a rotation by $90^\circ$.

\begin{figure}
\bigskip
\centertexdraw{
\drawdim truecm
% First drawing:
  \linewd.02
  % 1 row:
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  % 2. row
  \move (-0.866025403784439 -.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  % 3. row
  \move (-1.732050807568877 -1)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \ldreieck
  % 4. row
  \move (-2.598076211353316 -1.5)
  \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  % 5. row
  \move (-2.598076211353316 -1.5)
  \rdreieck \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  % Shade FIXED rhombus in this row
  \move (0.866025403784439 -3.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.4
  \esegment
  % 6. row
  \move (-2.598076211353316 -2.5)
  \rdreieck \rhombus \rhombus \rhombus \rhombus \rhombus
  % 7. row
  \move (-2.598076211353316 -3.5)
  \rdreieck \rhombus \rhombus \rhombus \rhombus
  \htext (-1.4 -0.2){$N$} \htext (2.6 -0.2){$N$}
  \htext (-3 -3){$M$}     \htext (4.6 -3){$M$}
  \htext (-1.4 -6){$N$}   \htext (2.6 -6){$N$}
  % \htext (1 -0.1){$6$}
  % \htext (0.80 -1.1){$5$}
  % \htext (0.80 -2.1){$4$}
  % \htext (0.80 -3.1){$3$}
  \htext (0.80 -4.1){$2$}
  % \htext (0.80 -5.1){$1$}
  % \htext (0.80 -6.1){$0$}
}
\caption{}
\label{figure-thm6}
\end{figure}

The first observation is that for any rhombus tiling of the hexagon
with side lengths $N,M,N,N,M,N$, 
there are exactly $N$ rhombi of the tiling that are
cut in two by the $(N+M)$-long symmetry axis. Removing these rhombi, and cutting 
the hexagon in two along the symmetry axis, leaves two symmetric halves of trapezoidal
shape with $N$ ``dents''. The following statement
counts rhombus tilings of such ``dented trapezoids''. (This is Lemma~2 in
\cite{HeGeAA}.)

\begin{figure}
\bigskip
\centertexdraw{
  \drawdim truecm  \linewd.02
  \ldreieck
  \move (-0.866025403784439 -.5)
  \rhombus %\ldreieck
  \move (-1.732050807568877 -1)
  \rhombus \rhombus %\ldreieck
  \move (-2.598076211353316 -1.5)
  \rhombus \rhombus \rhombus
  \ldreieck
  \move (-2.598076211353316 -1.5)
  \rdreieck
  \rhombus \rhombus \rhombus
  \ldreieck
  \move (-2.598076211353316 -2.5)
  \rdreieck
  \rhombus \rhombus \rhombus %\ldreieck
  \move (-2.598076211353316 -3.5)
  \rdreieck
  \rhombus \rhombus \rhombus
\htext (-1.4 -0.2){$N$}
\htext (-3.5 -3){$M$}
\htext (-1.4 -6){$N$}
\htext (1 -0.1){$6$}
\htext (1 -1.1){$5$}
\htext (1 -2.1){$4$}
\htext (1 -3.1){$3$}
\htext (1 -4.1){$2$}
\htext (1 -5.1){$1$}
\htext (1 -6.1){$0$}
}
\caption{}
\label{figure-prop4}
\end{figure}

\begin{prop}\label{prop:HeGe1}
The number of rhombus tilings of
a semi-hexagon with side lengths $N,\break M,N$ (i.e., the ``half'' of a
hexagon with side lengths 
$N,M,N,N,M,N$) and $N$ ``dents'' at positions $0\leq r_0<\dots<r_{N-1}<N+M$ 
(see Figure~\ref{figure-prop4}, where $N=4$, $M=3$, 
and the ``dents'' are at positions $0$, $1$, $4$ and $5$) is
\begin{equation*}
\left(\prod_{i=1}^{N-1}\frac{1}{i!}\right)\prod_{0\leq i<j\le N-1}(r_i-r_j)
=\left(\prod_{i=1}^{N-1}\frac{1}{i!}\right)
	\det_{0\le i,j\le N-1}\left(p_i(r_j)\right) ,
\end{equation*}
where $p_i(x)$ is an {\em arbitrary monic\/} polynomial of degree $i$
in $x$.
(The ``standard'' case would be  $p_i(x)=x^i$, which corresponds to the
Vandermonde determinant.)
\end{prop}

{From} this proposition, Helfgott and Gessel 
deduce another enumeration result. (This is Proposition~4 in \cite{HeGeAA}.)

\begin{figure}
\bigskip
\centertexdraw{
  \drawdim truecm  \linewd.02
  % 1 row:
  \rhombus \rhombus \rhombus \rhombus \ldreieck
  % 2. row
  \move (-0.866025403784439 -.5)
  \rhombus
  \move (0.866025403784439 -1.5)
  \rhombus \rhombus \rhombus \ldreieck
  % Shade allowed rhombus in this row
  \move (0.866025403784439 -0.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % 3. row
  \move (-1.732050807568877 -1)
  \rhombus \rhombus
  \move (0.866025403784439 -2.5)
  \rhombus \rhombus \rhombus \ldreieck
  % Shade allowed rhombus in this row
  \move (0.866025403784439 -1.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % 4. row
  \move (-2.598076211353316 -1.5)
  \rhombus \rhombus \rhombus
  \move (0.866025403784439 -3.5)
  \rhombus \rhombus \rhombus
  % Shade allowed rhombus in this row
  \move (0.866025403784439 -2.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % 5. row
  \move (-2.598076211353316 -1.5)
  \rdreieck \rhombus \rhombus \rhombus \rhombus \rhombus \rhombus
  % 6. row
  \move (-2.598076211353316 -2.5)
  \rdreieck \rhombus \rhombus \rhombus %\ldreieck
  \move (0.866025403784439 -5.5)
  \rhombus
  % Shade allowed rhombus in this row
  \move (0.866025403784439 -4.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
  % 7. row
  \move (-2.598076211353316 -3.5)
  \rdreieck \rhombus \rhombus \rhombus
  % Shade allowed rhombus in this row
  \move (0.866025403784439 -5.5)
  \bsegment
    \rlvec(0.866025403784439 -.5) \rlvec(-0.866025403784439 -.5)
	   \rlvec(-0.866025403784439 .5)
    \lfill f:0.8
  \esegment
   % Show the "barriers" which must not be crossed by a horizontal rhombus
  \linewd.1
  \move (0.866025403784439 -4.5)
  \rlvec(0 1)
  \move (0.866025403784439 -0.5)
  \rlvec(0 1)
  \htext (-1.4 -0.2){$N$} \htext (2.6 -0.2){$N$}
  \htext (-3.5 -3){$M$}     \htext (4.6 -3){$M$}
  \htext (-1.4 -6){$N$}   \htext (2.6 -6){$N$}
  \htext (0.80 -1.1){$5$}
  \htext (0.80 -2.1){$4$}
  \htext (0.80 -3.1){$3$}
  \htext (0.80 -5.1){$1$}
  \htext (0.80 -6.1){$0$}
}
\caption{}
\label{figure-prop5}
\end{figure}

\begin{prop}
\label{prop:HeGe}
Let $L$ be a subset of $0,1,\dots,N+M-1$ of cardinality at least $N$.
Then the number of rhombus tilings of
a hexagon with side lengths $N,M,N,N,M,N$, in which the set of 
rhombi on the $(N+M)$-long symmetry axis is a subset of $L$
(given in terms of the numbers of the rhombi, where the rhombi on the
symmetry axis are numbered from bottom to
top as $0,1,\dots,N+M-1$; see Figure~\ref{figure-prop5}, where $N=4$,
$M=3$, $L=\{0,1,3,4,5\}$; the set $L$ consists of the shaded rhombi), is
\begin{equation}\label{eq:HeGe}
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
\det_{0\le i,j\le  N-1}\bigg(\sum_{s\in L}p_i(s)p_j(s)\bigg),
\end{equation}
where, again, $p_i(x)$ is an {\em arbitrary monic\/} polynomial of degree
$i$ in $x$.
\end{prop}

{From} this proposition 
we can derive the following determinantal expression for
the number of rhombus tilings
of a hexagon with side lengths $N,M,N,N,M,N$ which contain an arbitrary
fixed rhombus on the $(N+M)$-long symmetry axis (see \cite{HeGeAA}).

\begin{prop}
\label{thm:HeGe-det}
The number of rhombus tilings of
a hexagon with side lengths $N,M,N,\break N,M,N$ which contain the $l$-th 
rhombus, $0\leq l\leq N+M-1$, on the $(N+M)$-long symmetry axis
(see Figure~\ref{figure-thm6}; the rhombus which is contained in every
tiling is shaded, i.e.,
$l=2$), is
\begin{equation}
\label{eq:HeGe-det}
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
\det_{1\le i,j\le  N-1}\bigg(\sum_{s=-l}^{N+M-l-1}s^{i+j}\bigg).
\end{equation}
\end{prop}

\begin{proof}
Let us first count the complementary set, i.e., the rhombus tilings
which do {\em not\/} contain rhombus $l$.
Obviously, we get this number from Proposition~\ref{prop:HeGe} with
$L=\{0,1,\dots,N+M-1\}\setminus\{l\}$. Then, if in addition we choose
$p_i(x)=(x-l)^i$, by formula \eqref{eq:HeGe} this number is
\begin{equation}
\label{eq:aux}
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
	\det_{0\le i,j\le  N-1}\Bigg(\underset{s\ne l}{\sum_{s=0}^{N+M-1}}(s-l)^{i+j}\Bigg)
=
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
	\det_{0\le i,j\le  N-1}\Bigg(\underset{s\ne
0}{\sum_{s=-l}^{N+M-l-1}}s^{i+j}\Bigg).
\end{equation}
We have to subtract this number from the total number of possible rhombus
tilings. This can be again expressed by making use of 
Proposition~\ref{prop:HeGe}, this time with $L=\{0,1,\dots,N+M-1\}$.
So, choosing $p_i(x)=(x-l)^i$ in \eqref{eq:HeGe} again, we obtain
\begin{equation}
\label{eq:auxtotal}
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
	\det_{0\le i,j\le N-1}\bigg({\sum_{s=0}^{N+M-1}}(s-l)^{i+j}\bigg)
=
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
	\det_{0\le i,j\le N-1}\bigg({\sum_{s=-l}^{N+M-l-1}}s^{i+j}\bigg)
\end{equation}
for the total number of rhombus tilings.

It should be observed that the determinants in \eqref{eq:aux} and
\eqref{eq:auxtotal} are almost the same. The only difference
is the $(0,0)$-entry, which is by 1 less in \eqref{eq:aux} than in
\eqref{eq:auxtotal}. Therefore, if we expand the determinant in
\eqref{eq:aux} with respect to the first row, we may rewrite \eqref{eq:aux} 
as
\begin{equation}
\label{eq:aux2}
\left(\prod_{i=1}^{N-1}\frac{1}{i!^2}\right)
\left(\det_{0\le i,j\le  N-1}\bigg(\sum_{s=-l}^{N+M-l-1}s^{i+j}\bigg) -
\det_{1\le i,j\le N-1}\bigg(\sum_{s=-l}^{N+M-l-1}s^{i+j}\bigg)\right).
\end{equation}
The number of rhombus tilings that we are interested in is the
difference of \eqref{eq:auxtotal} and \eqref{eq:aux2}, which is
exactly \eqref{eq:HeGe-det}.
\end{proof}




%%%@@@ end input from rhombdet


%%%@@@ input from proofs
\section{Determinant evaluations}
\label{sec:proofs}

\begin{lem} \label{lem:Det1}
Let $n$ and $N$ be positive integers. Then the determinant $D(n,n,N)$,
as defined in {\em\eqref{eq:gen-det}}, is equal to
\begin{multline} \label{eq:Det1}
n^N\prod_{i=1}^{\floor{N/2}}\big((n^2-i^2)^{N-2i+1}
(n^2-(i-1/2)^2)^{N-2i+1}\big)\\
\times
\frac{
	2^{N^2}N!\,(N+1)!\,\po{n-\ceil{N/2}}{2\ceil{N/2}+1}\prod_{i=1}^{N} i!^4	
}{
	n\floor{N/2}!^2\ceil{N/2}!^2 \prod_{i=1}^{2N+1}i!
}\\
\times\Biggl(	
	(-1)^N +
	\frac{
		4(N+2)n^2\floor{N/2}\ceil{N/2}
	}{
		(N-1)N(n^2-1)(\floor{N/2}+1)(\ceil{N/2}+1)
	}\\
\times	\sum_{h=0}^{\ceil{N/2}-1}
		\frac{
		\po{2}{h}\,\po{1-\ceil{N/2}}{h}\,\po{3/2+\ceil{N/2}}{h}\,
		\po{1-n}{h}\,\po{1+n}{h}
	}{
		\po{1}{h}\,\po{2+\ceil{N/2}}{h}\,\po{3/2-\ceil{N/2}}{h}\,
	\po{2+n}{h}\,\po{2-n}{h}
	}
\Biggr).
\end{multline}
\end{lem}

\begin{proof}
We proceed in several steps. An outline is as
follows.
In the first step we make the obvious observation that
$D(n,n,N)$ is actually a polynomial in $n$, of degree at most
$N(N+2)$. 
Next, we show that $D(n,n,N)$, as polynomial in $n$, has a lot of linear factors.
More precisely, in the second step, we show that $n^N$ is a factor of $D(n,n,N)$.
Then, in the third step, we show that $\prod _{i=1}
^{\floor{N/2}}(n^2-i^2)^{N-2i+1}$ is a factor of $D(n,n,N)$. Moreover,
in the fourth step, we show that $\prod _{i=1}
^{\floor{N/2}}(n^2-(i-1/2)^2)^{N-2i+1}$ is a factor of $D(n,n,N)$. 
 From a
combination of these four steps we are forced to conclude that
\begin{equation} \label{eq:polydef}
D(n,n,N)=n^N\bigg(\prod_{i=1}^{\floor{N/2}}(n^2-i^2)^{N-2i+1}
(n^2-(i-1/2)^2)^{N-2i+1}\bigg)	\cdot P(n,N),
\end{equation}
where $P(n,N)$ is a polynomial in $n$ of degree at most $2\ceil{N/2}$.
Finally, in the fifth step, we evaluate $P(n,N)$ at
$n=-\ceil{N/2},-\ceil{N/2}+1,\dots,\ceil{N/2}-1,\ceil{N/2}$.
Namely, we show that 
\begin{equation}
\label{eq:poly-at-0-ganz}
P(0,N)=\frac {(-1)^{N(N-1)/2}
\,3^{-N}}
{\prod_{i=1}^{\floor{N/2}}i^{2(N-2i+1)}\,(i-1/2)^{2(N-2i+1)}}
\prod_{i = 1}^{N-1}\left(\frac{i\,{{\left( i+1 \right) }^4}\,
             \left( i+2 \right) }{\left( 2 i+1 \right) \,
             {{\left(  2 i+2 \right) }^2}\,\left(  2 i+3 \right) }
         \right)^{N-i},
\end{equation}
and, for $1\le e\le\ceil{N/2}$, that
\begin{multline}
\label{eq:poly-at-eplus-ganz}
P(\pm e,N) =\dfrac {(-1)^{(N-2e+1)(N-2e)/2}\,2^{N-2e+1}
\prod_{i=1}^{2e-1}i!^2}
{\displaystyle e^N\underset{i\ne e}{\prod _{i=1} ^{\floor{N/2}}}(e-i)^{N-2i+1}
\prod_{i=1}^{\floor{N/2}}(e+i)^{N-2i+1}
(e^2-(i-1/2)^2)^{N-2i+1}}
\\
\times
\left(\frac{\left( 2 e \right) !^4}
  {\left( 4 e+1 \right) !}\right)^{N-2e+1}
   \prod_{i = 1}^{N-2e}\left(\frac{i\,\left( 2 e + i \right) ^4\,
            \left( 4 e + i \right) }{\left(  4 e + 2 i -1\right) \,
            {{\left( 4 e + 2 i \right) }^2}\,
            \left(  4 e + 2 i +1\right) }\right)^{N-2e+1-i}.
\end{multline}
Clearly, this determines a polynomial of maximal
degree $2\ceil{N/2}$ uniquely. In fact, an explicit expression for $P(n,N)$ can
immediately be written down using Lagrange interpolation.
As it turns out, the resulting expression for $P(n,N)$ is exactly
the expression covering the last three lines of \eqref{eq:Det1}.
In view of \eqref{eq:polydef},
this would finish the proof of the Lemma.

\newstep{Step 1. $D(n,n,N)$ is a polynomial in $n$ of degree at most $N(N+2)$}
It is standard that sums of powers, such as the entries of $D(n,n,N)$,
can be expressed using Bernoulli numbers. More precisely, we have
(cf.~\cite[p. 269ff]{Knuth}),
\begin{equation}
\label{eq:BernoulliP}
\sum _{s=-n} ^{n-1}s^{m}=\sum_{\ell =0}^m\frac {1}
{\ell +1}\binom{m}{\ell }B_{m-\ell }\left( n^{\ell +1}-(-n)^{\ell +1}\right),
\end{equation}
where $B_\ell $ denotes the $\ell $-th Bernoulli number.
Hence, the $(i,j)$-entry of $D(n,n,N)$ is a polynomial in $n$ of degree
$i+j+1$. Thus, by expanding the determinant $D(n,n,N)$ according to the
definition of a determinant and determining the degree of each term, 
it follows that $D(n,n,N)$ is a polynomial
in $n$ of degree at most $\sum _{1\le i,j\le N} ^{}(i+j+1)=2\binom
{N+1}2 +N=N(N+2)$.

{From} \eqref{eq:BernoulliP} we may read off another property of
$D(n,n,N)$, which we record here for later use,
\begin{equation}
D(-n,-n,N) 	= (-1)^ND(n,n,N).\label{eq:symD}
\end{equation}

\newstep{Step 2. $n^N$ is a factor of $D(n,n,N)$}
{From} the definition \eqref{eq:gen-det} of the entries of $D(n,n,N)$ and
\eqref{eq:BernoulliP} it is immediate that $n$ divides each entry of
$D(n,n,N)$. Hence, $n^N$ divides $D(n,n,N)$.

\newstep{Step 3. $\prod _{i=1}
^{\floor{N/2}}(n^2-i^2)^{N-2i+1}$ is a factor of $D(n,n,N)$}
In view of \eqref{eq:symD}, it suffices to prove that $(n-e)^{N-2e+1}$
divides $D(n,n,N)$ for $e=1,2,\dots,\floor{N/2}$. 
In order to do so, we claim that for each such $e$ there are $N-2e+1$
linear combinations of the columns, which are themselves linearly
independent, that vanish for $n=e$. 
Define the coefficients $a_j$ by 
\begin{equation}
\label{eq:lincomb-coefficients}
\underset{z\ne0}{\prod_{z=-e}^{e-1}}(x-z)=
\po{x-e+1}{e-1}\,\po{x+1}{e}=
\sum_{j=0}^{2e-1} a_jx^j.
\end{equation}
Then we claim that for $k=2e,2e+1,\dots,N$ we have
\begin{equation}
\label{eq:lincomb-definition}
\sum_{j=k-2e+1}^{k}a_{j-k+2e-1}\cdot\left(\text{column $j$ 
in $D(e,e,N)$}\right)=0.
\end{equation}
To verify \eqref{eq:lincomb-definition}, for $i=1,2,\dots,N$ we have
to prove that 
\begin{equation}
\label{eq:lincomb-compute}
\sum_{j=k-2e+1}^{k}a_{j-k+2e-1}\sum_{s=-e}^{e-1}s^{i+j}=0.
\end{equation}
To do so, we interchange sums in \eqref{eq:lincomb-compute},
and then use \eqref{eq:lincomb-coefficients}, to obtain
\begin{equation*}
\sum_{s=-e}^{e-1}
	\left(s^{i+k-2e+1}\sum_{j=0}^{2e-1}a_{j}s^{j}\right)
=\sum_{s=-e}^{e-1}
\bigg(s^{i+k-2e+1}\underset{z\ne0}{\prod_{z=-e}^{e-1}}(s-z)\bigg)
\end{equation*}
for the left-hand side in \eqref{eq:lincomb-compute},
which clearly vanishes for any integer $e>0$.

\newstep{Step 4. $\prod _{i=1}
^{\floor{N/2}}(n^2-(i-1/2)^2)^{N-2i+1}$ is a factor of $D(n,n,N)$}
In view of \eqref{eq:symD}, it suffices to prove that $(n-(e-1/2))^{N-2e+1}$
divides $D(n,n,N)$ for $e=1,2,\dots,\floor{N/2}$. In order to do so,
we claim, in a manner analogous
to Step~3, that for each such $e$ there are $N-2e+1$
linear combinations of the columns, which are themselves linearly
independent, that vanish for $n=e-1/2$. 
Define the coefficients $\tilde a_j$ by 
\begin{equation}
\label{eq:lincomb-coefficients-halb}
\prod_{z=-e+1}^{e-1}(x-z+1/2)=
\po{x-e+3/2}{2e-1}=
\sum_{j=0}^{2e-1} \tilde a_jx^j.
\end{equation}
Then we claim that for $k=2e,2e+1,\dots,N$ we have
\begin{equation}
\label{eq:lincomb-definition-halb}
\sum_{j=k-2e+1}^{k}\tilde a_{j-k+2e-1}\cdot\left(\text{column $j$ 
in $D(e-1/2,e-1/2,N)$}\right)=0.
\end{equation}
To verify \eqref{eq:lincomb-definition-halb}, for $i=1,2,\dots,N$ we have
to prove that 
\begin{equation*}
\sum_{j=k-2e+1}^{k}\tilde a_{j-k+2e-1}
\sum_{\ell =0}^{i+j}\frac {1}
{\ell +1}\binom{i+j}{\ell }B_{i+j-\ell }\left( (e-1/2)^{\ell
+1}-(-e+1/2)^{\ell +1}\right)=0,
\end{equation*}
where we used the right-hand side of formula \eqref{eq:BernoulliP} to
express the entries of $D(n,n,N)$. By a variation of
\eqref{eq:BernoulliP}, the last equation can be rewritten as
\begin{equation}
\label{eq:lincomb-compute-halb}
\sum_{j=k-2e+1}^{k}\tilde a_{j-k+2e-1}
\sum _{s=-e+1} ^{e-1}(s-1/2)^{i+j}=0.
\end{equation}
Again, to show that \eqref{eq:lincomb-compute-halb} is true,
we interchange sums,
and then use \eqref{eq:lincomb-coefficients-halb}, to obtain
\begin{equation*}
\sum_{s=-e+1}^{e-1}
	\left((s-1/2)^{i+k-2e+1}\sum_{j=0}^{2e-1}\tilde a_{j}(s-1/2)^{j}\right)
=\sum_{s=-e+1}^{e-1}
	\left((s-1/2)^{i+k-2e+1}{\prod_{z=-e+1}^{e-1}}(s-z)\right)
\end{equation*}
for the left-hand side in \eqref{eq:lincomb-compute-halb},
which clearly vanishes for any integer $e>0$.

\newstep{Step 5. Evaluation of the polynomial $P(n,N)$ at
$n=-\ceil{N/2},-\ceil{N/2}+1,\dots,\ceil{N/2}$}
We start by observing that the symmetry relation \eqref{eq:symD} for
$D(n,n,N)$ is ``inherited" by $P(n,N)$. To be precise, we have
\begin{equation}
P(-n,N) 	= P(n,N).\label{eq:symP}
\end{equation}
In view of \eqref{eq:symP} it suffices to determine the evaluations of
$P(n,N)$ at $n=0,1,\dots,\ceil{N/2}$. 

What we would like to do is, for any $e$ with $0\le 
e\le \ceil{N/2}$,
to set $n=e$ in
\eqref{eq:polydef}, compute $D(e,e,N)$,
and then express $P(e,N)$ as
the ratio of $D(e,e,N)$ and the right-hand side
product evaluated 
at $n=e$. Unfortunately, this is typically a ratio $0/0$ and, hence,
undetermined. So, we have to first divide both sides of \eqref{eq:polydef}
by the appropriate power of $(n-e)$, and only then set $n=e$.

This program is easily carried out for $e=0$.
As we observed in Step~2, each entry of $D(n,n,N)$ is divisible by $n$.
Hence, division of both sides of \eqref{eq:polydef} by $n^N$ and then
specializing to $0$, transforms
\eqref{eq:polydef} into the equation
\begin{equation*}
P(0,N)=\frac{
	\det_{1\le i,j\le N}\left(2B_{i+j}\right)
}{\prod_{i=1}^{\floor{N/2}}i^{2(N-2i+1)}\,(i-1/2)^{2(N-2i+1)}}.
\end{equation*}
The Hankel determinant of Bernoulli numbers which appears in the
right-hand side expression can be evaluated by 
Theorem~\ref{thm:HankelBernoulli} with $n$ replaced by $N$ and $a=b=c=d=1$.
Thus, we obtain \eqref{eq:poly-at-0-ganz}.
\smallskip


The case $1\leq e \leq\ceil{N/2}$ requires more work. First, we have
to ``preprocess'' the determinant $D(n,n,N)$. Define the coefficients
$a_j$ as before in \eqref{eq:lincomb-coefficients},
\begin{equation*}
\underset{z\ne0}{\prod_{z=-e}^{e-1}}(x-z)=
\po{x-e+1}{e-1}\,\po{x+1}{e}=
\sum_{j=0}^{2e-1} a_jx^j.
\end{equation*}
Then, for $k=N,N-1,\dots,2e$,
in this order, add 
\begin{equation*}
\sum_{j=k-2e+1}^{k-1}a_{j-k+2e-1}\cdot\left(\text{column $j$ 
in $D(n,n,N)$}\right)
\end{equation*}
to column $k$. Thus, by \eqref{eq:lincomb-definition}, each entry in
column $k$, $k=2e,2e+1,\dots,N$, will be divisible by $(n-e)$ after
performing these operations. Next, apply the analogous row operations.
I.e., for $k=N,N-1,\dots,2e$,
in this order, add 
\begin{equation*}
\sum_{j=k-2e+1}^{k-1}a_{j-k+2e-1}\cdot\left(\text{row $j$ 
in $D(n,n,N)$}\right)
\end{equation*}
to row $k$.

Now we divide $(n-e)^{N-2e+1}$ on both sides of \eqref{eq:polydef},
and only then set $n=e$. As a result, from equation \eqref{eq:polydef} 
we obtain
\begin{align} \notag
P(e,N)&=\dfrac {\det_{1\le i,j,\le N}\left(\begin{matrix}
X&*\\0&Y\end{matrix}\right)} 
{\displaystyle e^N\underset{i\ne e}{\prod _{i=1} ^{\floor{N/2}}}(e-i)^{N-2i+1}
\prod_{i=1}^{\floor{N/2}}(e+i)^{N-2i+1}
(e^2-(i-1/2)^2)^{N-2i+1}} \\
&=\dfrac {\det X\,\det Y}
{\displaystyle e^N\underset{i\ne e}{\prod _{i=1} ^{\floor{N/2}}}(e-i)^{N-2i+1}
\prod_{i=1}^{\floor{N/2}}(e+i)^{N-2i+1}
(e^2-(i-1/2)^2)^{N-2i+1}},
\label{eq:matXY}
\end{align}
where $X$ is the $(2e-1)\times (2e-1)$ matrix
$X=\big(\sum_{s=-e}^{e-1} s^{i+j}\big)_{i,j=1,\dots, 2e-1}$, and
where $Y$ is the $(N-2e+1)\times (N-2e+1)$ matrix
$(Y_{ij})_{i,j=1,\dots, N-2e+1}$ whose entries are given by
$$Y_{ij}=\bigg(\frac {1}{n-e}
\sum_{s=-n}^{n-1}s^{i+j}\left(\po{s-e+1}{e-1}\,\po{s+1}{e}\right)^2\bigg)
\bigg\vert_{n=e}\ .$$
Alternatively, 
\begin{equation} \label{eq:Yij} 
Y_{ij}=\frac {d} {d n}\bigg(
\sum_{s=-n}^{n-1}s^{i+j}\left(\po{s-e+1}{e-1}\,\po{s+1}{e}\right)^2\bigg)
\bigg\vert_{n=e}\ .
\end{equation}
The expression on the right-hand side is a certain linear combination
of expressions of the form
$$H_m:=\frac {d} {d n}\bigg(
\sum_{s=-n}^{n-1}s^m\bigg)
\bigg\vert_{n=e}\ .
$$
In order to compute $H_m$, we use \eqref{eq:BernoulliP} (and
variations thereof) to rewrite it as
\begin{align}
\notag
H_m&=\sum _{\ell =0} ^{m}\binom m\ell  B_{m-\ell }(e^\ell
+(-e)^\ell )\\
&=2B_m+ \sum _{\ell =1} ^{m}m\binom {m-1}{\ell -1}\frac {1} {\ell }
B_{m-\ell }(e^\ell +(-e)^\ell )\notag\\
&=2B_m+ \sum _{s=0} ^{e-1}ms^{m-1}-\sum _{s=-e} ^{-1}ms^{m-1}\notag\\
&=2B_m+ \sum _{s=0} ^{e-1}\left(\frac {d} {ds}s^m\right)
-\sum _{s=-e} ^{-1}\left(\frac {d} {ds}s^m\right).\label{eq:H_m}
\end{align}
Thus, using the symbolic notation $B^k\equiv B_k$ for the
Bernoulli numbers, a combination of \eqref{eq:Yij} and \eqref{eq:H_m}
leads to
\begin{align*} 
Y_{ij}&=2B^{i+j}\left(\po{B-e+1}{e-1}\,\po{B+1}{e}\right)^2+
\sum _{s=0} ^{e-1}\frac {d} {ds}\left(s^{i+j}
\left(\po{s-e+1}{e-1}\,\po{s+1}{e}\right)^2\right)\\
&\quad \quad 
-\sum _{s=-e} ^{-1}\frac {d} {ds}\left(s^{i+j}
\left(\po{s-e+1}{e-1}\,\po{s+1}{e}\right)^2\right)\\
&=2B^{i+j}\left(\po{B-e+1}{e-1}\,\po{B+1}{e}\right)^2,
\end{align*}
as each summand in either sum on the right-hand side vanishes.

Substituting all this into \eqref{eq:matXY}, we arrive at
\begin{equation}
\label{eq:poly-at-eplus}
P(e,N) = \dfrac{D(e,e,2e-1)
	\det_{1\le i,j\le N-2e+1}\left(
		2B^{i+j}\left(\po{B-e+1}{e-1}\,\po{B+1}{e}\right)^2
	\right)}
{\displaystyle e^N\underset{i\ne e}{\prod _{i=1} ^{\floor{N/2}}}(e-i)^{N-2i+1}
\prod_{i=1}^{\floor{N/2}}(e+i)^{N-2i+1}
(e^2-(i-1/2)^2)^{N-2i+1}},
\end{equation}
which is valid for $1\leq e\leq \ceil{N/2}$.
The determinant $D(e,e,2e-1)$ is evaluated separately in
Lemma~\ref{lem:detklein}. 
The Hankel determinant of (linear combinations of) Bernoulli numbers
which appears in the 
right-hand side expression can be evaluated by 
Theorem~\ref{thm:HankelBernoulli} with $n$ replaced by $N-2e+1$, 
$a=b=e+1$ and $c=d=e$.
Thus, we obtain \eqref{eq:poly-at-eplus-ganz}.

\smallskip
This completes the proof of the Lemma.
\end{proof}

\begin{lem} \label{lem:Det2}
Let $n$ and $m$ be positive integers. Then the determinant
$D(n,n-1,2m-1)$,
as defined in \eqref{eq:gen-det}, is equal to
\begin{multline}\label{eq:Det2odd}
% Der Faktor:
\frac{{2^{- 4 m + 5}}}{\po{
       {n- m + \frac{1}{2}}}{2 m - 1}\,
      \left( \prod_{i = 1}^{4 m - 1}i! \right) }\,\left(2 m + 1\right) \,
      {{\left(2 n - 1\right) }^2}\,{\binom{2 m}{m}}\,
      {\binom{2 m - 4}{m - 2}}
		\\\times
		\po{n-m}{m - 2}\,
      \po{n + 2}{m - 2}
      \left( \prod_{i = 1}^{2 m - 1}{{i!}^4} \right) \,
      \left( \prod_{i = 1}^{2 m - 2}
         \po{2 n -i-2}{2i + 3}\right)
\\\times
\Biggl(
{\frac {m \left( 2 m-3 \right)  \left( 2 m-1 \right)  
      \left( n-2 \right)  \left( n+1 \right)  
      \left( n^2-n + 2 m+1 \right) } 
    {\left( m-1 \right)  \left( 2 m+1 \right) }}\kern5cm\\
+
  {\frac {6(n-1)n} {(m+1)}
		\sum_{h=0}^{m-2}
			\frac{
				\po{3}{h}\,\po{\frac{5}{2}}{h}\,
                \po{2 - m}{h}\,\po{\frac{3}{2} + m}{h}\,
					\po{2 - n}{h}\,\po{1 + n}{h}
			}{
				\po{1}{h}\,\po{\frac{3}{2}}{h}\,
                \po{2 + m}{h}\,\po{\frac{5}{2} - m}{h}\,
					\po{2 + n}{h}\,\po{3 - n}{h}
			}
	}
\Biggr).
\end{multline}
Likewise, the determinant $D(n,n-1,2m)$ is equal to
\begin{multline}\label{eq:Det2even}
% Der Faktor
\frac{4}{\prod_{i = 1}^{4 m + 1}
      i!}\,\left(2 m + 3\right) \,{{\left(2 n - 1\right) }^2}\,
     {\binom{2 m - 2}{m - 1}}\,{\binom{2 m + 2}{m + 1}}%\,
	  \\\times
     \po{n-m}{m - 2}\,
     \po{n + 2}{m - 2}\,
     \po{n- m - 1}{2 m + 2}
     \left( \prod_{i = 1}^{2 m}{{i!}^4} \right) \,
     \left( \prod_{i = 1}^{2 m - 2}
        \po{2 n -i-2}{2i + 3} \right)
% Die Summe:
\\\times
\Biggl(
{\frac {\left( 2 m-1 \right)  \left( m+1 \right)  \left( 2 m+1 \right)  
     \left(  n-2 \right)  \left( n+1 \right)  
     \left( n^2-n-2 m  \right) } {m \left( 2 m+3 \right)
}}\kern5cm\\
+
  {\frac {6(n-1)n} {(m+2)}
		\sum_{h=0}^{m-1}
			\frac{
				\po{3}{h}\,\po{\frac{5}{2}}{h}\,
                \po{1 - m}{h}\,\po{\frac{5}{2} + m}{h}\,
					\po{2 - n}{h}\,\po{1 + n}{h}
			}{
				\po{1}{h}\,\po{\frac{3}{2}}{h}\,
                \po{3 + m}{h}\,\po{\frac{3}{2} - m}{h}\,
					\po{2 + n}{h}\,\po{3 - n}{h}
			}
	}
\Biggr)
\end{multline}
\end{lem}

\begin{proof}
Basically, the proof proceeds in the same way as above. By 
considerations which parallel Steps~1--4 of the previous proof, 
we deduce that
\begin{multline} \label{eq:polydef2}
D(n,n-1,N)=
	\left(n-\frac{1}{2}\right)^N
	\left((n-1)n\right)^{N-1}\\
\times
	\Bigg(\prod_{i=2}^{\ceil{(N+1)/2}}\left(
		\left(n-i+\frac{1}{2}\right)
        \left(n+i-\frac{3}{2}\right)(n-i)(n+i-1)
	\right)^{N-2i+2}\Bigg)	\cdot Q(n,N),
\end{multline}
where $Q(n,N)$ is a polynomial in $n$ of degree at most $2\ceil{(N+1)/2}$.

Also, Step~5 of the previous proof has a parallel here. Eventually, 
this yields evaluations of $Q(n,N)$ at
$n=-\ceil{(N+1)/2}+1,-\ceil{(N+1)/2}+2,
\dots,\ceil{(N+1)/2}-1,\ceil{(N+1)/2}$. In particular, the symmetry
relation which plays the role of \eqref{eq:symP} in this new context is
\begin{equation}
Q(-n+1,N) 	= Q(n,N),\label{eq:symQ}
\end{equation}
while the evaluation of determinant $D(e,e-1,2e-2)$ which is 
needed here (and replaces the evaluation of $D(e,e,2e-1)$ 
in this new context; compare \eqref{eq:poly-at-eplus}) 
is evaluated separately in Lemma~\ref{lem:detklein2}.

Unfortunately, this is not good enough. The polynomial $Q(n,N)$
is a polynomial of maximal degree $2\ceil{(N+1)/2}$, but by now we have
found only $2\ceil{(N+1)/2}$ explicit special evaluations of
$Q(n,N)$. Hence, we need one more information about $Q(n,N)$.
We get this missing piece of information by computing the 
leading coefficient of $Q(n,N)$. This is easily done. By the
definition of $D(n,n-1,N)$, given by \eqref{eq:gen-det}, by the
analogue of \eqref{eq:BernoulliP},
\begin{equation}
\label{eq:BernoulliP2}
\sum _{s=-n} ^{n-2}s^{m}=\sum_{\ell =0}^m\frac {1}
{\ell +1}\binom{m}{\ell }B_{m-\ell }\left( (n-1)^{\ell +1}-(-n)^{\ell +1}\right),
\end{equation}
and by \eqref{eq:polydef2}, 
the leading coefficient is given by
\begin{equation*}
\det_{1\leq i,j\leq N}\left(
	\begin{cases}
		\frac{2}{i+j+1} & \text{if $i+j$ is even,} \\
		0 & \text{otherwise.}
	\end{cases}
\right)  .
\end{equation*}
It is easy to see, by sorting rows and columns with odd indices
to the beginning, that this determinant equals the product of the
minor consisting of the odd rows and columns times the minor 
consisting of the even rows and columns, explicitly
\begin{equation*}
\det_{1\leq i,j\leq \ceil{N/2}}\left(
		\frac{2}{2i+2j-1} 
\right)  
\det_{1\leq i,j\leq \floor{N/2}}\left(
		\frac{2}{2i+2j+1} 
\right)  .
\end{equation*}
Both of these determinants are easily evaluated by means of the
Cauchy determinant evaluation (see \cite[vol.~III, p.~311]{MuirAB})
\begin{equation*}
\det_{1\leq i,j\leq K}\left(
		\frac{1}{x_i+y_j} 
\right)  =
	\frac{
		\prod_{1\leq i<j\leq K}(x_i-x_j)(y_i-y_j)
	}{
		\prod_{1\leq i, j\leq K}(x_i+y_j)
	}.
\end{equation*}

For the simplifications of the resulting expression for $Q(n,N)$ 
it turns out to be convenient to separate the cases $N=2m$ and $N=2m-1$.

\smallskip
This concludes the proof of the Lemma.
\end{proof}

\begin{lem}
\label{lem:Det3}
Let $n$ and $m$ be positive integers. 
Then the determinant $D(n,n-2,2m+1)$,
as defined in {\em\eqref{eq:gen-det}}, is equal to
\begin{multline}
\label{eq:Det3odd}
\frac{
	(2m-2)!\,(2m+2)!\,
	\po{n-m-1}{2m+1}\,\po{n-m-2}{2m+3}
}{
	(m-1)!\,m!\,(m+1)!\,(m+4)!\,(4m+3)!\,n(n-2)
}
\\
\times
\prod_{i=0}^{2m}
\frac{
	(i+1)!^3\,\po{2n -i- 2}{2i+1}
}{
	(2m+i+2)!
}%\\ \times
\Biggl( {\frac {2   \left( m+3 \right)  \left( m+4 \right)  
     \left( 2 m-1 \right) \left( 2 m +1\right)  x(m,n) }
   {\left( n-2 \right)  {{\left( n-1 \right) }^2} n}}\\
	-
	\frac{
		24(2m + 3)(2m + 5)(n - m - 3)(n + m + 1)
	}{
		(n - 3)(n + 1)
	}\\ 
\cdot
		\sum_{h = 0}^{m}
			\frac{
				\po{4}{h}\,
				\po{- m}{h}\,
				\po{{\frac{7}{2}} + m}{h}\,
				\po{3 - n}{h}\,
				\po{1 + n}{h}
			}{
				\po{1}{h}\,
				\po{5 + m}{h}\,
				\po{{\frac{3}{2}} - m}{h}\,
				\po{2 + n}{h}\,
				\po{4 - n}{h}
			}
\Biggr),
\end{multline}
where 

\vskip-10pt
{\small
\begin{multline*} 
x(m,n) = -72 m - 204 {m^2} - 212 {m^3} - 
       96 {m^4} - 16 {m^5} - 30 n - 86 m n - 68 {m^2} n - 
       16 {m^3} n \\
- 29 {n^2} + 3 m {n^2} + 26 {m^2} {n^2} + 
       8 {m^3} {n^2} + 44 {n^3} + 40 m {n^3} + 8 {m^2} {n^3} - 
       11 {n^4} - 10 m {n^4} - 2 {m^2} {n^4} .
\end{multline*}
}
\vskip-10pt

\noindent
Likewise, the determinant $D(n,n-2,2m)$ is equal to
\begin{multline}
\label{eq:Det3even}
\frac{
	(2m-2)!\,(2m+2)!\,
	\po{n-m-1}{2m+1}^2
}{
	(2m-3)\,(m-1)!\,m!\,(m+1)!\,(m+3)!\,(4m+1)!\,n(n-2)
}
\\
\times
\prod_{i=0}^{2m - 1}
\frac{
	(i+1)!^3\,\po{2n -i- 2}{2i+1}
}{
	(2m+i+1)!
}
\Biggl( {\frac {  \left( m+2 \right)  \left( m+3 \right)  
     \left( 2 m-3 \right) \left( 2 m-1 \right)  y(m,n) }
   {\left( n-2 \right)  {{\left( n-1 \right) }^2} n}}\\
	-
	\frac{
		24m(2m + 3)(n - m - 2)(n + m)
	}{
		(n - 3)(n + 1)
	}%\\ \times
		\sum_{h = 0}^{m - 1}
			\frac{
				\po{4}{h}\,
				\po{1 - m}{h}\,
				\po{{\frac{5}{2}} + m}{h}\,
				\po{3 - n}{h}\,
				\po{1 + n}{h}
			}{
				\po{1}{h}\,
				\po{4 + m}{h}\,
				\po{{\frac{5}{2}} - m}{h}\,
				\po{2 + n}{h}\,
				\po{4 - n}{h}
			}
\Biggr),
\end{multline}
where 

\vskip-10pt
{\small
\begin{multline*} 
y(m,n)=     12 {m^2} + 44 {m^3} + 48 {m^4} + 16 {m^5} + 16 m n + 
       28 {m^2} n + 16 {m^3} n + 4 {n^2} - 20 m {n^2}\\
 - 
       22 {m^2} {n^2} - 8 {m^3} {n^2} - 4 {n^3} + 12 m {n^3} + 
       8 {m^2} {n^3} + {n^4} - 3 m {n^4} - 2 {m^2} {n^4} .
\end{multline*}
}
\end{lem}

\begin{proof}
Basically, the proof proceeds in the same was as in the preceding lemmas.
By considerations which parallel Steps 1--4 of the proof of Lemma
\ref{lem:Det1}, we deduce that
\begin{equation} \label{eq:polydef3}
D(n,n-2,N)=
	(n-1)^N\Bigg(\prod_{i=1}^{N-1}\left((n-1)^2-\frac{i^2}{4}\right)^{N-i}
	\prod_{i=2}^{\floor{N/2}}\left((n-1)^2-i^2\right)\Bigg)\cdot R(n,N),
\end{equation}
where $R(n,N)$ is a polynomial in $n$ of degree at most $2\ceil{(N)/2}+2$
if $N\geq 2$.

Step~5 of the proof of Lemma \ref{lem:Det1} has a parallel here, too, which
yields evaluations of $R(n,N)$ at
$n=-\ceil{N/2}+1,\dots,0,2,\dots,\ceil{N/2}+1$.
In particular, the symmetry relation which plays the role of \eqref{eq:symP}
in this context is
\begin{equation}
\label{eq:symR}
R(-n+2,N)=R(n,N).
\end{equation}
While the evaluation of the determinant $D(e,e-2,2e-3)$ which is needed here
is done separately in Lemma~\ref{lem:detklein3} for $e>2$, 
the case $e=2$ leads to the $2\times2$ determinant
$\det_{1\le i,j\le 2}\left((-1)^{i+j}+(-2)^{i+j}\right)=4$.

The evaluation of $R(n,N)$ at $n=1$ requires extra treatment.
First, factoring $(n-1)$ out of each 
column of $D(n,n-2,N)$ and then setting $n=1$ yields 
the determinant $\det_{1\le i,j\le N}\big(2(B-1)^{i+j}\big)$
(using again the symbolic notation $B^k\equiv B_k$). 
By row and column operations,
this determinant can be transformed into the form
\begin{equation*}
2^N\det_{1\le i,j\le N}\big(B^{i+j-2}(B-1)^2\big).
\end{equation*}
This determinant agrees with the determinant on the left-hand side of
\eqref{eq:HankelBernoulli} with $n=N$, $a=b=0$ and $c=d=2$ (adopting
the usual convention $(B+1)_{-1}:=1/B$). 
Unfortunately, formula \eqref{eq:HankelBernoulli} does not hold for
this choice of parameters. In fact, the evaluation of this
determinant is rather tedious
and therefore given separately in Lemma~\ref{lem:extra-Bernoulli}.

Altogether, we have $2\ceil{N/2}+1$ evaluations for our polynomial
$R(n,N)$ at special values of $n$ so far. 
So we are short of exactly two informations on $R(n,N)$.
We can get these missing
informations by computing the leading coefficient of the polynomial,
which is done in the same way as in the proof of the preceding lemma,
and by exploiting the symmetry \eqref{eq:symR} once again.

For the simplifications of the resulting expression for $R(n,N)$ 
it turns out to be again convenient to separate the cases $N=2m$ and
$N=2m+1$.

\smallskip
This concludes the proof of the Lemma.
\end{proof}



%%%@@@ end input from proofs


%%%@@@ input from lemma

\begin{lem}
\label{lem:detklein}
For $e\geq 1$, we have
\begin{equation}
\label{eq:detklein}
D(e,e,2e-1)=
	\det_{1\le i,j\le  2e-1}\left(\sum_{s=-e}^{e-1} s^{i+j}\right) =
\prod_{i=1}^{2e-1}i!^2.
\end{equation}
{\em(}The determinant $D(a,b,N)$ was defined in
\eqref{eq:gen-det}.{\em)}
\end{lem}

\begin{proof}
For $\ell \ge1$ define the polynomials $p_\ell (x)$ by
$p_\ell (x):=(x-\ceil{\ell /2}+1)_\ell $.
Furthermore, define coefficients $a_{\ell ,k}$, $k=1,2,\dots,\ell $, by
$p_\ell (x)=\sum _{k=1} ^{\ell }a_{\ell ,k}x^k$.
Now, for $\ell =2e-1,2e-2,\dots,2$, in this order, 
we replace column $c_\ell $ of the matrix underlying $D(e,e,2e-1)$ by
the linear combination of columns
$\sum _{k=1} ^{\ell }a_{\ell ,k}c_k,$
and afterwards do the same sort of replacement in the rows. These
operations yield a matrix with $(i,j)$-entry equal to
\begin{equation} \label{eq:Sum1}
\sum _{s=-e} ^{e-1}p_i(s)p_j(s)=
\sum _{s=-e} ^{e-1}(s-\ceil{i/2}+1)_i\,(s-\ceil{j/2}+1)_j.
\end{equation}

The next sequence of operations will turn this matrix into upper
triangular form, so that the determinant is easily obtained by
forming the product of the diagonal entries.

To begin with, it should be noted that for $j=2e-1$ the sum \eqref{eq:Sum1}
consists of just the term corresponding to $s=-e$. We subtract
$p_j(-e)/p_{2e-1}(-e)$ times column $2e-1$ from column $j$,
$j=1,2,\dots,2e-2$. The previous observation tells us that these
operations have the effect that the entries that are in one of the
columns $j=1,2,\dots,2e-2$ become
\begin{equation} \label{eq:Sum2}
\sum _{s=-e+1} ^{e-1}p_i(s)p_j(s)=
\sum _{s=-e+1} ^{e-1}(s-\ceil{i/2}+1)_i\,(s-\ceil{j/2}+1)_j,
\end{equation}
i.e., the summand corresponding to $s=-e$ has been eliminated. In
particular, all the entries in the last row, except the rightmost
entry, of course, are 0.

Next consider column $2e-2$. After the above column operations, the
sum \eqref{eq:Sum2} with $j=2e-2$ 
which defines the entries collapses to just the term
corresponding to $s=e-1$. We subtract
$p_j(e-1)/p_{2e-2}(e-1)$ times column $2e-2$ from column $j$,
$j=1,2,\dots,2e-3$. The previous observation tells us that these
operations have the effect that now the entries that are in one of the
columns $j=1,2,\dots,2e-3$ become
\begin{equation} \label{eq:Sum3}
\sum _{s=-e+1} ^{e-2}p_i(s)p_j(s)=
\sum _{s=-e+1} ^{e-2}(s-\ceil{i/2}+1)_i\,(s-\ceil{j/2}+1)_j,
\end{equation}
i.e., the summand corresponding to $s=e-1$ has been eliminated as well. In
particular, all the entries in the next-to-last row, except the two rightmost
entries, of course, are 0.

If we continue in the same manner, then eventually we arrive at an
upper triangular matrix whose $i$-th diagonal entry is $i!^2$. 
Thus, the result \eqref{eq:detklein} follows.
\end{proof}

\begin{lem}
\label{lem:detklein2}
For $e\geq 2$ we have
\begin{equation}
\label{eq:detklein2}
D(e,e-1,2e-2)=
	\det_{1\le i,j\le  2e-2}\left(\sum_{s=-e}^{e-2} s^{i+j}\right) =
	\prod_{i=1}^{2e-2}i!^2.
\end{equation}
{\em(}The determinant $D(a,b,N)$ was defined in
\eqref{eq:gen-det}.{\em)}
\end{lem}
\begin{proof}
This identity can be established in essentially the same manner as the
previous lemma. The basic difference is that one has to replace the
polynomials $p_\ell (x)$ in the previous proof by the polynomials
\begin{equation*}
q_\ell (x):=
\begin{cases}
	x			&\text{if $\ell =1$,}\\
	\po{x-\floor{\ell /2}+1}{\ell }	&\text{if $\ell >1$.}
\end{cases}
\end{equation*}
Everything else is completely analogous. We leave the details to the
reader.
\end{proof}

\begin{lem}
\label{lem:detklein3}
For $e\geq 3$ we have
\begin{equation}
\label{eq:detklein3}
D(e,e-2,2e-3)=
	\det_{1\le i,j\le  2e-3}\left(\sum_{s=-e}^{e-3} s^{i+j}\right) =
	\prod_{i=1}^{2e-3}i!^2.
\end{equation}
{\em(}The determinant $D(a,b,N)$ was defined in
\eqref{eq:gen-det}.{\em)}
\end{lem}
\begin{proof}
This identity can be established in essentially the same manner as the
previous lemmas. The basic difference is that one has to replace the
polynomials in the previous proofs by the polynomials
\begin{equation*}
r_\ell(x):=
\begin{cases}
	x			&\text{if $\ell =1$,}\\
	x(x+1)	&\text{if $\ell =2$,}\\
	\po{x-\ceil{\ell /2}+2}{\ell }	&\text{if $\ell >2$.}
\end{cases}
\end{equation*}
Everything else is completely analogous. We leave the details to the
reader.
\end{proof}



%%%@@@ end input from lemma


%%%@@@ input from orthpol
\section{Orthogonal polynomials, continued fractions, and Hankel
determinants of Bernoulli numbers}
\label{sec:orthpol}
In this section we review some facts about the interrelations between
orthogonal polynomials, continued fractions, and Hankel determinants.
Good sources for information about these topics are
\cite{JoThAA,KoSwAA,PerronKB,SzegoOP,VienAE,WallCF}.

To begin with, we recall Favard's Theorem.

\begin{thm} \label{thm:Favard}
{\em (Cf\@.  \cite[Th\'eor\`eme~9 on
p.~I-4]{VienAE} or \cite[Theorem 50.1]{WallCF}).}
Let $(p_n(x))_{n\ge0}$ be a sequence of monic polynomials, the
polynomial $p_n(x)$ having degree $n$. Then the sequence $(p_n(x))$ is
(formally) orthogonal with respect to a linear functional $L$, i.e.,
$L(p_n(x)p_m(x))=\delta_{mn}c_n$ for some sequence $(c_n)_{n\ge0}$ of
nonzero numbers, with $\delta_{m,n}$ denoting the Kronecker delta (i.e.,
$\delta_{m,n}=1$ if $m=n$ and $\delta_{m,n}=0$ otherwise) if and only
if there exist sequences $(a_n)_{n\ge1}$ and $(b_n)_{n\ge1}$, with
$b_n\ne0$ for all $n\ge1$, such that the three-term recurrence
\begin{equation}
p_{n+1}(x)=(a_{n}+x)p_{n}(x)-b_{n}p_{n-1}(x),\quad \quad 
\text{ for } n\geq 1,
\label{eq:three-term}
\end{equation}
holds, with initial conditions $p_0(x)=1$ and $p_1(x)=x+a_0$.
\end{thm}

It is a simple fact that, given a linear functional, the
corresponding orthogonal polynomials can be expressed in form of
certain determinants.

\begin{thm} \label{thm:ortho}
{\em (Cf\@. \cite[Theorem~7.15, (7.2.41)]{WallCF} or \cite[(36)]{VienAE}).}
Let $L$ be a linear functional defined on polynomials with moments
$\mu_n=L(x^n)$. Then the corresponding set 
$(p_n(x))_{n\ge0}$ of orthogonal polynomials is given by
\begin{equation}
\label{eq:ortho}
p_{n}(x)=d_n^{-1}\det\begin{pmatrix}
		\mu_0	&	\mu_1	&	\mu_2	&	\dots &	\mu_n \\
		\mu_1	&	\mu_2	&	\dots	&	\mu_n	&	\mu_{n+1} \\
		\mu_2	&	\dots	&	\mu_n	&	\mu_{n+1}&\mu_{n+2} \\
		\hdotsfor{5} \\
		\mu_{n-1} &	\mu_n	&	\mu_{n+1}&\dots	&\mu_{2n-1} \\
		1	&	x&\dots&	x^{n-1}&x^n 
	\end{pmatrix},
\end{equation}
where $d_n=\det_{0\le i,j\le n-1}(\mu_{i+j})$.
\end{thm}

The next theorem addresses the relation between
orthogonal polynomials and continued fractions, the link being the
generating function of the moments.
\begin{thm} \label{thm:momentgf}
{\em (Cf\@. \cite[Theorem 51.1]{WallCF} or \cite[Proposition~1, (7), on p.~V-5]{VienAE})}.
Let $(p_n(x))_{n\ge0}$ be a sequence of monic polynomials, the
polynomial $p_n(x)$ having degree $n$, which is orthogonal with
respect to some functional $L$. Let
\begin{equation}
p_{n+1}(x)=(a_{n}+x)p_{n}(x)-b_{n}p_{n-1}(x)
\label{eq:three-term2}
\end{equation}
be the corresponding three-term recurrence which is guaranteed by
Favard's theorem. Then the generating function for the moments
$\mu_n=L(x^n)$ satisfies
\begin{equation}
\label{eq:momentgf}
\sum_{n=0}^\infty{\mu_n}x^n=\cfrac{
	\mu_0}
		{1+a_0x-\cfrac{
			b_1x^2}
				{1+a_1x-\cfrac{
					b_2x^2}
						{1+a_2x-\cdots}}}\quad .
\end{equation}
\end{thm}

We remark that a continued fraction of the type as in
\eqref{eq:momentgf} is called a {\em J-fraction}.

The next theorem addresses the relation between J-fractions and
Hankel determinants.
\begin{thm}
\label{cor:cfracHankel}
{\em (Cf\@. \cite[Theorem 51.1]{WallCF} or
\cite[Corollaire~6, (19), on p.~IV-17]{VienAE})}.
Let $(\mu_n)_{n\ge0}$ be a sequence of numbers whose generating
function, when written in terms of a J-fraction, is given by
\eqref{eq:momentgf}. 
Then the Hankel determinant $\det_{0\le i,j\le n-1}(\mu_{i+j})$
equals $\mu_0^nb_1^{n-1}b_2^{n-2}\cdots b_{n-2}^2b_{n-1}$.
\end{thm}

The family of orthogonal polynomials that is of interest to us is the
{\em continuous Hahn polynomials}, which were first studied by
Atakishiyev and Suslov \cite{AtSuAA} and Askey \cite{AskeyCHP}. 
The theorem below lists the
relevant facts about these polynomials.
\begin{thm} \label{thm:cHahn}
{\em (Cf\@. \cite[Section 1.4]{KoSwAA})}.
The continuous Hahn polynomials $(p_n(a,b,c,d;x))_{n\ge0}$ are
the monic polynomials defined as the terminating hypergeometric series
\begin{multline} \label{eq:cHahn}
p_n(a,b,c,d;x)\\
=\frac{\complexikl^n \,\po{a+c}{n}\,\po{a+d}{n}}{\po{a+b+c+d+n-1}{n}}
\HypF{3}{2}{-n,n+a+b+c+d-1,a+x\complexi }{a+c,a+d}{1}.
\end{multline}
These polynomials satisfy the three-term recurrence
\begin{equation} \label{eq:cHahnrec}
p_{n+1}(a,b,c,d;x)  
=	(x-A_n(a,b,c,d))p_{n}(a,b,c,d;x)-B_n(a,b,c,d)p_{n-1}(a,b,c,d;x),
\end{equation}
where
\begin{multline}
A_n(a,b,c,d) = \complexi\,\Biggl( a + {\frac{n\,\left(  b + c + n-1 \right) \,
        \left(  b + d + n-1 \right) }{\left(  a + b + c + d + 
          2 n-2 \right) \,\left(  a + b + c + d + 2 n-1 \right) }} + \\
    {\frac{\left( 1 - a - b - c - d - n \right) \,
        \left( a + c + n \right) \,\left( a + d + n \right) }{
        \left( -1 + a + b + c + d + 2 n \right) \,
        \left( a + b + c + d + 2 n \right) }} \Biggr),
\end{multline}
and
\begin{multline}
B_n(a,b,c,d) = -{\frac{n\,\left(a + c + n - 1\right) \,
     \left(b + c + n - 1\right) \,\left(a + d + n - 1\right)}
	  {\left(a + b + c + d + 
       2 n - 3\right) \,{{\left(a + b + c + d + 2 n - 2\right) }^2}}}\\
\times{\frac{\left(b + d + n - 1\right) \,
     \left(a + b + c + d + n - 2\right) }{
     \left(a + b + c + d + 2 n - 1\right) }}.
\end{multline}
They are orthogonal with respect to the functional $L$ which is given
by
\begin{equation} \label{eq:cHahnortho}
L(p(x))=\frac{1}{2\pi}\int_{-\infty}^{\infty}
\Gamma(a+x\complexi )\,\Gamma(b+x\complexi )
\,\Gamma(c-x\complexi )\,\Gamma(d-x\complexi )\,p(x)\,dx\,.
\end{equation}
In this integral, if necessary, the path of integration has to be 
deformed, so that it separates the increasing sequences
$\big((a+k)\complexi\big)_{k\ge0}$ and 
$\big((b+k)\complexi\big)_{k\ge0}$ of poles from
the decreasing sequences 
$\big(-(c+k)\complexi\big)_{k\ge0}$ and
$\big(-(d+k)\complexi\big)_{k\ge0}$ of poles (see \cite{AskeyCHP}).

Explicitly, the orthogonality relation is
\begin{multline}
L(p_m(a,b,c,d;x)p_n(a,b,c,d;x)) \\
=\frac{
\Gamma(n+a+c)\,\Gamma(n+a+d)\,\Gamma(n+b+c)\,\Gamma(n+b+d)\,n!
}{
\po{a+b+c+d+n-1}{n}\,\po{a+b+c+d+n-1}{n+1}\Gamma(n+a+b+c+d-1)
}\delta_{m,n}.
\end{multline}
In particular, 
\begin{equation}
L(1)={\frac{\left(a + c -1\right) !\,\left(b + c - 1\right) !\,
     \left(a + d - 1\right) !\,\left(b + d - 1\right) !}{\left(
       a + b + c + d - 1\right) !}}.
\end{equation}
\end{thm}
\begin{rem}
(1)
The reader should be aware that Theorem~\ref{thm:cHahn} is formulated for 
the {\em monic} 
form of the continuous Hahn polynomials, so that our polynomials are
not the polynomials which are denoted by $p_n(x;a,b,c,d)$ in 
\cite[Section 1.4]{KoSwAA} but those which are denoted there, slightly
confusingly, by $p_n(x)$.

(2) Although the definition \eqref{eq:cHahn} does not show it, the
continuous Hahn polynomials are symmetric in $a$ and $b$, and in $c$
and $d$, because the orthogonality
measure \eqref{eq:cHahnortho} has
these symmetries. In addition, there is another symmetry when the
roles of $a,b$ are interchanged with the roles of $c,d$. Namely, we
have 
\begin{equation} \label{eq:cHahn-symmetry}
p_n(a,b,c,d;x)= (-1)^np_n(c,d,a,b;-x).
\end{equation}
This follows from the fact that, when performing the substitution
$x\to-x$ in the
integral \eqref{eq:cHahnortho}, we obtain almost the same integrand,
the only differences being that the roles of $a,b$ and $c,d$ are
interchanged, and that $p(x)$ is replaced by $p(-x)$. 
The path of integration also does not change, at least as long as
all of $a,b,c,d$ are positive. This is, however, sufficient to
conclude that \eqref{eq:cHahn-symmetry} holds for all $a,b,c,d$
because, for fixed $n$, both sides of \eqref{eq:cHahn-symmetry} are
rational of bounded degree in $a,b,c,d$.
(The same conclusion could also be less elegantly derived by applying some
$_3F_2$-transformation formulas.)
\end{rem}

Now, by combining Theorems~\ref{thm:momentgf}, \ref{cor:cfracHankel},
and \ref{thm:cHahn} we are able to derive without difficulty the
determinant evaluation that we need in Step~5 of the proof of
Lemma~\ref{lem:Det1} and the analogous places in Lemmas~\ref{lem:Det2}
and \ref{lem:Det3} (which, in turn, are essential for the proofs of
Theorems~\ref{thm:MEven}--\ref{thm:NOdd2}). 

\begin{thm} \label{thm:HankelBernoulli}
For positive integers $a,b$ and nonnegative integers $c,d$ there holds
\begin{multline} \label{eq:HankelBernoulli}
\det_{1\le i,j\le n}
\left(
	B^{i+j}\,\po{B+1}{a-1}\,\po{B+1}{b-1}\,\po{-B+1}{c-1}\,\po{-B+1}{d-1}
\right)\\
=(-1)^{n(n-1)/2}\left({\frac{\left(a + c -1\right) !\,\left(b + c - 1\right) !\,
     \left(a + d - 1\right) !\,\left(b + d - 1\right) !}{\left(
       a + b + c + d - 1\right) !}}\right)^n\\
\times\prod_{i=1}^{n-1}\Biggl({\frac{i\,\left(a + c + i - 1\right) \,
     \left(b + c + i - 1\right) \,\left(a + d + i - 1\right)}
	  {\left(a + b + c + d + 
       2 i - 3\right) \,{{\left(a + b + c + d + 2 i - 2\right) }^2}}}\\
\times{\frac{\left(b + d + i - 1\right) \,
     \left(a + b + c + d + i - 2\right) }{
     \left(a + b + c + d + 2 i - 1\right) }}
\Biggr)^{n-i},
\end{multline}
where we have again used the symbolic notation $B^k\equiv B_k$. (In
case that $c$ or $d$ should be zero, we have to interpret $(-B+1)_{-1}$
as $1/(-B)$.)
\end{thm}
\begin{proof}
We claim that the moments for the continuous Hahn
polynomials, i.e., for the linear functional as defined in
\eqref{eq:cHahnortho}, are
\begin{equation}
\label{eq:CHmoments}
\mu_n=(B/\complexi)^n B^2
\,\po{B+1}{a-1}\,\po{B+1}{b-1}\,\po{-B+1}{c-1}\,\po{-B+1}{d-1}.
\end{equation}
Let us suppose
for the moment that this is already established. Then, by
Theorem~\ref{thm:momentgf} and \eqref{eq:cHahnrec} of Theorem~\ref{thm:cHahn},
we have
\begin{multline*}
\sum_{n\geq 0}{(B/\complexi)^n B^2
\,\po{B+1}{a-1}\,\po{B+1}{b-1}\,\po{-B+1}{c-1}\,\po{-B+1}{d-1}}x^n\\ 
=\cfrac{B^2
\,\po{B+1}{a-1}\,\po{B+1}{b-1}\,\po{-B+1}{c-1}\,\po{-B+1}{d-1}	}
		{1+A_0(a,b,c,d)x-\cfrac{
			B_1(a,b,c,d)x^2}
				{1+A_1(a,b,c,d)x-\cfrac{
					B_2(a,b,c,d)x^2}
						{1+A_2(a,b,c,d)x-\cdots}}}\quad .
\end{multline*}
But, by Theorem~\ref{cor:cfracHankel},
this immediately implies the truth of \eqref{eq:HankelBernoulli}.

It remains to verify \eqref{eq:CHmoments}. Using
\eqref{eq:cHahnortho}, this is a rather straightforward computation:
\begin{align}
\mu_n &=
	\frac{1}{2\pi}\int_{-\infty}^{\infty}
	\Gamma(a+x\complexi )\,\Gamma(b+x\complexi )
	\,\Gamma(c-x\complexi )\,\Gamma(d-x\complexi )\,x^n\,dx\notag\\
&\kern-5pt= \frac{1}{2\pi}\int_{-\infty\complexi }^{\infty\complexi }
	\Gamma(a+z)\,\Gamma(b+z)\,\Gamma(c-z)\,\Gamma(d-z)
	\frac{z^n}{\complexikl ^{n+1}}\,dz\notag\\
&\kern-5pt= \frac{1}{2\pi \complexikl^{n+1} }
	\int_{-\infty\complexi }^{\infty\complexi }
	\po{z}{a}\,\po{z}{b}\,\po{-z}{c}\,\po{-z}{d}\left(\Gamma(z)\,\Gamma(-z)\right)^2
	z^n\,dz\notag\\
&\kern-5pt= \frac{1}{2\pi \complexikl^{n+1} }
	\int_{-\infty\complexi }^{\infty\complexi }
	\po{z}{a}\,\po{z}{b}\,\po{-z}{c}\,\po{-z}{d}
		\left(-\frac{\pi}{z\sin \pi z}\right)^2 z^n\,dz\notag\\
&\kern-5pt= \frac{1}{2\pi \complexikl^{n+1} }
	\int_{-\infty\complexi }^{\infty\complexi }
	\po{z+1}{a-1}\,\po{z+1}{b-1}\,\po{-z+1}{c-1}\,\po{-z+1}{d-1}
		\left(\frac{\pi}{\sin\pi z}\right)^2
	z^{n+2}\,dz.\label{eq:momentn}
\end{align}
In the third line we used the relation $\Gamma(a+z)=\po{z}{a}\,\Gamma(z)$
for $a\in {\mathbb N}$ 
(see, e.g., \cite[1.2(2)]{Erdelyi}), and in the fourth line we used 
the formula
$\Gamma(z)\,\Gamma(-z)=-\frac{\pi}{z\sin\pi z}$ (see \cite[1.2(5)]{Erdelyi}).
The reader should notice that, because of the convention regarding
the path of integration in \eqref{eq:cHahnortho}, in case that $c$ or
$d$ are zero the path of
integration in the last line of
\eqref{eq:momentn} is deformed so that it crosses the real axis
between the singularities $z=-1$ and $z=0$.

To finish the calculation, we appeal to the following integral
representation of Bernoulli numbers (see
\cite[p.~75]{Noerlund}, and let $\alpha\to0^+$ in the relevant
identity in the middle of the page)
\begin{equation}
B_\nu=\frac{1}{2\pi \complexi }
\int_{-\infty\complexi }^{\infty\complexi }z^\nu
\left(\frac{\pi}{\sin\pi z}\right)^2 dz.
\end{equation}
(If $\nu=0$ or $\nu=1$ then the path of integration is indented so
that it avoids the singularity $z=0$, passing it on the negative
side, but to the right of the singularity $z=-1$.)
If we use this formula in \eqref{eq:momentn}, 
we obtain \eqref{eq:CHmoments} immediately. 
\end{proof}

At least two special cases of Theorem~\ref{thm:HankelBernoulli} have
explicitly appeared in the literature before. The case $a=b=1$,
$c=d=0$ appears for example in \cite[(3.1)]{AlCaAA}. The case
$a=b=c=d=1$ appears in \cite[App.~A.5, p.~322]{MehtAB}.

\medskip
In the proof of Theorems~\ref{thm:NEven2} and \ref{thm:NOdd2} (to be
precise, in the proof of Lemma~\ref{lem:extra-Bernoulli}), 
we make use of a rather recent result on (formal)
orthogonal polynomials, due to Leclerc \cite[Theorem~1]{Leclerc}.

\begin{thm} \label{thm:Leclerc}
For an arbitrary sequence of numbers $(\mu_n)_{n\ge0}$ let
$(P_n(x))_{n\ge0}$ be the sequence of polynomials defined by
\begin{equation}
\label{eq:defineP}
P_{n}(x):=\det\begin{pmatrix}
		\mu_0	&	\mu_1	&	\mu_2	&	\dots &	\mu_n \\
		\mu_1	&	\mu_2	&	\dots	&	\mu_n	&	\mu_{n+1} \\
		\mu_2	&	\dots	&	\mu_n	&	\mu_{n+1}&\mu_{n+2} \\
		\hdotsfor{5} \\
		\mu_{n-1} &	\mu_n	&	\mu_{n+1}&\dots	&\mu_{2n-1} \\
		1	&	x&\dots&	x^{n-1}&x^n 
	\end{pmatrix}.
\end{equation}
(These are, up to normalization, the orthogonal polynomials
with respect to the linear functional with moments $(\mu_n)$; 
compare Theorem~\ref{thm:ortho}.)
Furthermore, let
$(Q_n(x))_{n\ge0}$ be the sequence of polynomials defined by
\begin{equation*}
Q_{n}(x):=\sum_{k=0}^n\mu_k\binom{n}{k}(-x)^{n-k}.
\end{equation*}
Then, for arbitrary
integers $l\geq1$, $m\geq1$, there holds
\begin{multline}
\label{eq:Leclerc}
\det\begin{pmatrix}
P_{l}(x) & P_{l+1}(x) &\dots & P_{l+m-1}(x) \\
P_{l}^\prime(x) & P_{l+1}^\prime(x) 
	&\dots& P_{l+m-1}^\prime(x) \\
\vdots & \vdots & & \vdots \\
P_{l}^{(m-1)}(x) & P_{l+1}^{(m-1)}(x) &
	\dots& P_{l+m-1}^{(m-1)}(x)
\end{pmatrix} \\
= C_{l,m}
\det\begin{pmatrix}
Q_{m}(x) & Q_{m+1}(x) &\dots& Q_{m+l-1}(x) \\
Q_{m+1}(x) & Q_{m+2}(x) &\dots& Q_{m+l}(x) \\
\vdots & \vdots & & \vdots \\
Q_{m+l-1}(x) & Q_{m+l}(x) &\dots& Q_{m+2l-2}(x)
\end{pmatrix},
\end{multline}
where
\begin{equation}
\label{eq:Cln}
C_{l,m}=(-1)^{lm}\prod_{k=1}^{m-1}k!
\det\begin{pmatrix}
\mu_0 & \mu_1 &\dots& \mu_{k+l-1} \\
\mu_1 & \mu_2 &\dots& \mu_{k+l} \\
\vdots & \vdots & & \vdots \\
\mu_{k+l-1} & \mu_{k+l} &\dots& \mu_{2k+2l-2} \\ 
\end{pmatrix}.
\end{equation}
\end{thm}



%%%@@@ end input from orthpol


%%%@@@ input from lemma2
\section{Auxiliary lemmas}
\begin{lem}
\label{lem:extra-Bernoulli}
For any integer $n\geq 1$, we have
\begin{multline}
\label{eq:extra-Bernoulli}
\det_{1\leq i,j\leq n}\left(B^{i+j-2}(B-1)^2\right)\\
=
{(-1)^{{n(n-1)}/{2}}} {6^{-n}}(1 + 2n + 5n^2 + 4n^3 + n^4)
\prod_{i=1}^n\left(\frac{i(i+1)^4(i+2)}{(2i+1)(2i+2)^2(2i+3)}\right)^{n-i}
\end{multline}
(again, using the symbolic notation $B^k\equiv B_k$).
\end{lem}
\begin{proof}
The determinant in \eqref{eq:extra-Bernoulli} is equal to the determinant in
\eqref{eq:HankelBernoulli} with $a=b=0$ and $c=d=2$ (again, with the
convention that $(B+1)_{-1}$ is interpreted as $1/B$). Unfortunately,
for this choice of parameters, formula \eqref{eq:HankelBernoulli} is
not valid. However, the determinant in \eqref{eq:extra-Bernoulli} is
very close to the determinant in \eqref{eq:HankelBernoulli} with 
$a=b=2$ and $c=d=0$. In fact, because of the well-known property of
Bernoulli numbers that $B_{2k+1}=0$ for all positive integers $k$,
multiplication of all even numbered rows and columns of the
determinant in \eqref{eq:HankelBernoulli} by $-1$ (which does not change the
value of the determinant) with this choice of
parameters transforms the latter 
into the determinant $\Delta_n:=\det_{1\le i,j\le n}(\la_{i+j-2})$, where
\begin{align} \notag
\la_0	&=B^0(B+1)^2= B^{0}(B-1)^2-2,\\
\notag
\la_1	&=-B^1(B+1)^2= B^{1}(B-1)^2+1,\\
\label{eq:la}
\la_n	&=(-1)^nB^n(B+1)^2= B^{n}(B-1)^2\text{ for }n\geq2.
\end{align}
So, because of the deviating definitions of $\la_0$ and $\la_1$, 
the only difference between $\Delta_n$ and the determinant in
\eqref{eq:extra-Bernoulli} is in the top-left entry and its right and
bottom neighbour. Linearity of the determinant $\Delta_n$ 
in the first row and column then implies that
\begin{equation} \label{eq:extra-expansion}
\det_{1\leq i,j\leq n}\left(B^{i+j-2}(B-1)^2\right)=
\Delta_{n} -
\Delta_{n}^{\{1,2\};\{1,2\}} +
2\Delta_{n}^{\{1\};\{1\}} +
2\Delta_{n}^{\{1\};\{2\}}.
\end{equation}
Here, $A^{\{i_1,i_2,\dots\};\{j_1,j_2\dots\}}$ denotes the minor of $A$ with
rows $i_1,i_2,\dots$ and columns $j_1,j_2\dots$ deleted. (Empty minors
are defined to be zero).

Now observe that from Theorem~\ref{thm:HankelBernoulli} with $a=b=2$
and $c=d=0$ we obtain immediately that
\begin{equation} \label{eq:Delta}
\Delta_{n}=(-1)^{{n(n-1)}/2}
\prod_{i=1}^{n}\frac{(i-1)!\,i!^4\,(i+1)!}{(2i)!\,(2i+1)!}.
\end{equation}
Likewise, from Theorem~\ref{thm:HankelBernoulli} with $a=b=1$ and
$c=d=2$ we have
\begin{equation} \label{eq:Delta-1,1}
\Delta_{n}^{\{1\};\{1\}}=(-1)^{{(n-1)(n-2)}/2}
\prod_{i=1}^{n-1}\frac{(i-1)!\,(i+1)!^4\,(i+3)!}{(2i+2)!\,(2i+3)!}.
\end{equation}

Our next observation is that $\Delta_{n}^{\{1\};\{2\}}$ is,
essentially, the coefficient of $x$ in the continuous Hahn polynomial
$p_{n-1}(0,1,2,2;x)$. To make a more precise statement, consider
\eqref{eq:ortho} with $\mu_k=\la_{k+1}$, $k=0,1,\dots$. Then,
obviously, $\Delta_{n}^{\{1\};\{2\}}$ equals
$$(-1)^n\det_{1\le i,j\le n-1}(\la_{i+j-1})
\cdot\big(\text {coefficient of $x$ in } p_{n-1}(x)\big).$$
On the other hand, Theorem~\ref{thm:ortho} says that, with this
choice of the $\mu_k$, the polynomials $p_n(x)$ are orthogonal with
associated moments $\la_{k+1}$, $k=0,1,\dots$. By comparing
\eqref{eq:la} and \eqref{eq:CHmoments} with $a=b=2$, $c=0$, and
$d=1$, we see that we must have
$$p_n(x)=(-\complexi)^n\,p_n(2,2,0,1;-x/\complexi).$$
Therefore, if we remember \eqref{eq:cHahn-symmetry},
we obtain that $\Delta_{n}^{\{1\};\{2\}}$ equals
\begin{equation} \label{eq:0122}
(-1)^n(\complexi)^{n-1}\det_{1\le i,j\le n-1}(\la_{i+j-1})
\cdot\big(\text {coefficient of $x$ in }p_{n-1}(0,1,2,2;x/\complexi)\big).
\end{equation}
The Hankel determinant in this expression can be evaluated 
by using Theorem~\ref{thm:HankelBernoulli} with $a=b=2$, $c=0$, $d=1$, 
and $n$ replaced by $n-1$. By substituting the result in
\eqref{eq:0122} and by using the definition \eqref{eq:cHahn} of
continuous Hahn polynomials with $a=0$, $b=1$, $c=d=2$, we obtain
\begin{multline} \label{eq:Delta-1,2}
\Delta_{n}^{\{1\};\{2\}}=
(-1)^{n(n+1)/2}\frac{n!^2}{(n+3)_{n-1}}
\\ \times
\bigg(\prod_{i=1}^{n-1}\frac{(i-1)!\,i!^2\,(i+1)!^2\,(i+2)!}
{(2i+1)!\,(2i+2)!}\bigg)
\sum_{k=1}^{n-1}
	\frac{\po{1-n}{k}\,\po{n+3}{k}}{k\,(k+1)!^2}.
\end{multline}

\def\a{a}

The remaining minor $\Delta_{n}^{\{1,2\};\{1,2\}}$ requires additional
work. We employ Theorem~\ref{thm:Leclerc} with
$l=n-2$, $m=2$, $x=0$, and $\mu_k=\la_{k+2}$, $k=0,1,\dots$.
With this choice of parameters, the determinant on the
right-hand side of \eqref{eq:Leclerc} is precisely our remaining minor
$\Delta_{n}^{\{1,2\};\{1,2\}}$. The single determinant occurring in the
definition \eqref{eq:Cln} of $C_{n-2,2}$ can be evaluated by
using Theorem~\ref{thm:HankelBernoulli} with $a=b=1$, $c=d=2$, and $n$
replaced by $n-2$, so that we have
\begin{equation}
\label{eq:Cl2}
C_{n-2,2}=(-1)^{(n-1)(n-2)/2}
\prod_{i=1}^{n-1}\frac{(i-1)!\,(i+1)!^4\,(i+3)!}{(2i+2)!\,(2i+3)!}.
\end{equation}
The determinant on the left-hand side \eqref{eq:Leclerc} 
is a $2\times2$-determinant
whose entries are the constant term and the coefficient of $x$, respectively,
of $P_{n-2}(x)$ and $P_{n-1}(x)$. The polynomials $P_n(x)$,
$n=0,1,\dots$, on the other hand, are orthogonal polynomials with
associated moments $\la_{k+2}$, $k=0,1,\dots$ (compare \eqref{eq:defineP} 
and Theorem~\ref{thm:ortho}). By comparing the definition
\eqref{eq:la} of the $\la_i$'s with \eqref{eq:CHmoments}, it is seen
that the polynomials $P_n(x)$ must agree, up to
normalization, with the continuous Hahn polynomials with parameters
$a=b=1$ and $c=d=2$. To be precise, we have
$$P_n(x)=(\complexi)^n\det_{1\le i,j\le
n}(\la_{i+j})\,p_n(1,1,2,2;x/\complexi) .$$
Clearly, the Hankel
determinant on the right-hand side can be once again evaluated
by means of Theorem~\ref{thm:HankelBernoulli} with $a=b=1$ and
$c=d=2$.
In summary, from \eqref{eq:Leclerc} with the above choice of
parameters we infer
\begin{multline} \label{eq:Delta-12,12}
\Delta_{n}^{\{1,2\};\{1,2\}}=
	(-1)^{n(n-1)/2}\frac{\po{3}{n-2}^2\,\po{3}{n-1}^2}
        {\po{n+3}{n-2}\,\po{n+4}{n-1}}
	\left(\prod_{i=1}^{n-2}\frac{(i-1)!\,(i+1)!^4\,(i+3)!}
        {(2i+2)!\,(2i+3)!}\right)
	\\ \times
	\left(c_{0,n-2}c_{1,n-1}-c_{0,n-1}c_{1,n-2}\right),
\end{multline}
where $c_{0,n}$ and $c_{1,n}$ denote the coefficient of $x^0$ and $x^1$,
respectively, in
\begin{equation*}
\sum_{k=0}^{n}\frac{\po{-n}{k}\,\po{n+5}{k}\,\po{1+x}{k}}{k!\,\po{3}{k}^2}.
\end{equation*}
Using hypergeometric notation \eqref{eq:hypergeom}, 
the first of these two, $c_{0,n}$, can be expressed as
$$
  {\frac {4} 
    {\left( 1 + n \right)  \left( 2 + n \right)  \left( 3 + n \right)  
      \left( 4 + n \right) }}
\left(
{} _{2} F _{1} \!\left [ \begin{matrix} { -2 - n, 3 + n}\\ {1}\end{matrix} ;
       {\displaystyle 1}\right ]
-   1 + (n+2)(n+3)\right).
$$
The $_2F_1$-series can be evaluated by means of the Chu--Vandermonde
summation formula (see \cite[(1.7.7); Appendix (III.4)]{SlatAC})
\begin{equation} \label{eq:Chu-Van}
{} _{2} F _{1} \!\left [ \matrix { a, -N}\\ { c}\endmatrix ; {\displaystyle
   1}\right ]  = {\frac {({ \textstyle c-a}) _{N} } 
    {({ \textstyle c}) _{N} }},
\end{equation} 
where $N$ is a nonnegative integer. This yields
\begin{equation} \label{eq:c}
c_{0,n}=\begin{cases}
	\frac{4}{(n+1)(n+4)} &\text{ if $n$ is even,}\\
	\frac{4}{(n+2)(n+3)} &\text{ if $n$ odd.}
\end{cases}
\end{equation}
By combining \eqref{eq:Delta-1,2}, \eqref{eq:Delta-12,12} and
\eqref{eq:c}, we get, after a considerable amount of simplification,
\begin{multline*}
2\Delta_{n}^{\{1\};\{2\}}-\Delta_{n}^{\{1,2\};\{1,2\}}=
        (-1)^{\binom {n-1}2}
	\prod_{i=1}^{n}\frac{(i-1)!\,i!^4\,(i+1)!}{(2i)!\,(2i+1)!}\\
	\times\Biggl(\sum_{k=0}^{n-2}
	\frac{(-1)^k\,(n+k+3)!\,(1+(-1)^n(n+1))}{(k+1)\,(k+2)!^2\,(n-k-2)!}\\
+       \frac {(n+2)!} {(n-1)!}
	\sum_{k=0}^{n-2}
	\frac{(-1)^{n+k-1}(n+k+3)!}{(k+2)!\,(k+3)!\,(n-k-2)!}
        \sum_{j=0}^{k}\frac {1} {j+1}
\Biggr).
\end{multline*}
By Lemma~\ref{lem:weird}, with $n$ replaced by $n-1$, 
this expression reduces to
\begin{equation} \label{eq:Delta-diff}
2\Delta_{n}^{\{1\};\{2\}}-\Delta_{n}^{\{1,2\};\{1,2\}}=
(-1)^{\binom {n+1}2}
((-1)^n(n+1)+2)\frac {(n+2)!} {(n-1)!}
\prod_{i=1}^{n}\frac{(i-1)!\,i!^4\,(i+1)!}{(2i)!\,(2i+1)!}.
\end{equation}
Substituting \eqref{eq:Delta}, \eqref{eq:Delta-1,1}, and
\eqref{eq:Delta-diff} in \eqref{eq:extra-expansion}, and simplifying the
resulting expression, we eventually arrive at \eqref{eq:extra-Bernoulli}.
\end{proof}

% Die m^Áhsame Identit^Ät

\begin{lem}
\label{lem:weird}
For $n\geq 0$, we have
\begin{multline}
\label{eq:weird}
\frac{n!}{(n+3)!}\sum_{k=0}^{n-1}
	\frac{(-1)^k\,(n+k+4)!\,(1-(-1)^n(n+2))}{(k+1)\,(k+2)!^2\,(n-k-1)!}\\
+        \sum_{k=0}^{n-1}
	\frac{(-1)^{n+k}(n+k+4)!}{(k+2)!\,(k+3)!\,(n-k-1)!}
        \sum_{j=0}^{k}\frac {1} {j+1}=(-1)^n(n+2)-2.
\end{multline}
\end{lem}

\begin{proof}
We shall treat the two sums (in the first and the second line in
\eqref{eq:weird}, respectively) separately.

First we consider the sum in the first line of \eqref{eq:weird}.
We replace the term $1/(k+1)$ by $\int_0^1x^k\dx$, interchange
summation and integration, and write the sum
in hypergeometric notation \eqref{eq:hypergeom}. This gives
\begin{equation*}
\frac{1-(-1)^n(n+2)}{\po{n+1}{3}}\int_0^1\left(\frac{n+2}{x^2}
\HypF{3}{2}{n+3,-n-1}{1}{x}-\frac{n+2}{x^2}+\frac{\po{n+1}{3}}{x}
\right)\,{dx}.
\end{equation*}
Using the transformation formula (see \cite[(1.7.1.3), sum
reversed at the right-hand side]{SlatAC})
\begin{equation} \label{eq:T2106}
{} _{2} F _{1} \!\left [ \matrix { a, -N}\\ { c}\endmatrix ; {\displaystyle
   z}\right ]  = {z^N}
  {\frac {  ({ \textstyle c-a}) _{N} } {({ \textstyle c}) _{N} }}
  {} _{2} F _{1} \!\left [ \matrix { -N, 1 - c - N}\\ {
       1 + a - c - N}\endmatrix ; {\displaystyle {\frac {z-1} z}}\right ] 
\end{equation} 
(where $N$ is a nonnegative integer) with $a=n+3$, $N=n+1$, 
$c=1$, and $z=x$, this is transformed into
\begin{multline} \label{eq:ex1}
\frac{1-(-1)^n(n+2)}{\po{n+1}{3}}
\int_0^1\Biggl((n+2)^2(-1)^{n+1}
\sum_{k=0}^{n+1}\frac{(-1)^k\,\po{-n-1}{k}^2}{k!\,(k+1)!}(1-x)^k\,x^{n-k-1}\\
-\frac{n+2}{x^2}+\frac{\po{n+1}{3}}{x}
\Biggr)\,{dx}.
\end{multline}
Recall that for $\Re(\alpha)>-1$ and $\Re(\beta)>-1$ we have the following
integral representation for the {\em Euler beta function},
\begin{equation}
\label{eq:beta-integral}
\int_0^1(1-x)^\alpha x^\beta {\, dx}=
\frac{\Gamma(\alpha+1)\Gamma(\beta+1)}{\Gamma(\alpha+\beta+2)}.
\end{equation}
Use of this identity in \eqref{eq:ex1} 
wherever possible (i.e., it is applied to the summands with 
$k=0,\dots,n-1$) yields
after some simplification the expression
\begin{multline} \label{eq:ex2}
\frac{1-(-1)^n(n+2)}{\po{n+1}{3}}\Biggl(
(-1)^{n+1}\left(1-\sum_{k=0}^n\frac{\po{-n-2}{k}^2}{\po{-n}{k}\,k!}\right)
\\+\int_0^1\left(	% Dieses Integral bereits ausgebessert: So stimmt's!!!
	\frac{\po{n+1}{2}\left((n+3)-(n+2)(1-x)^n\right)}{x}+
	\frac{(n+2)\left((1-x)^{n+1}-1\right)}{x^2}
\right){\, dx}\Biggr).
\end{multline}
We would like to write the sum in the first line as a hypergeometric
series. Unfortunately, this cannot be done by just straighforwardly
extending the summation over all nonnegative $k$ because of the term
$(-n)_k$ in the denominator, which is 0 for $k=n+1$. The way to
overcome this problem is to rewrite the sum as a limit,
\begin{multline*} 
\sum_{k=0}^n\frac{\po{-n-2}{k}^2}{\po{-n}{k}\,k!}\\
=\lim_{\ep\to0}\Bigg({}_2F_1\!\left[\matrix -n-2,n-2\\-n-\ep\endmatrix;
1\right]
-\frac {(-n-2)_{n+1}^2} {(n+1)!\,(-n-\ep)_{n+1}}
-\frac {(-n-2)_{n+2}^2} {(n+2)!\,(-n-\ep)_{n+2}}\Bigg).
\end{multline*}
The $_2F_1$-series can be evaluated by means of the Chu--Vandermonde
summation formula \eqref{eq:Chu-Van}. Substituing the result in
\eqref{eq:ex2}, we obtain
\begin{multline*}
\hskip-8pt
\frac{1-(-1)^n(n+2)}{\po{n+1}{3}}\Biggl(
(-1)^{n+1}\left(
	1-\lim_{\varepsilon\rightarrow0}\left(
	\frac{\po{2-\varepsilon}{n+2}-(n+2)\,(n+2)!\,(1-\varepsilon)-(n+2)!}{
		\po{-n-\varepsilon}{n+2}
	}
\right)\right)\\
+\int_0^1\left(	% Dieses Integral bereits ausgebessert: So stimmt's!!!
	\frac{\po{n+1}{2}\left((n+3)-(n+2)\,(1-x)^n\right)}{x}+
	\frac{(n+2)\left((1-x)^{n+1}-1\right)}{x^2}
\right){\, dx}
\Biggr).
\end{multline*}
Of course, the limit can be computed by de l'Hospital's rule, so that
we get
\begin{multline*}
\frac{1-(-1)^n(n+2)}{\po{n+1}{3}}\Biggl(
(-1)^{n+1}
	+\frac{1}{n!}\left(
	(n+3)!\sum_{k=1}^{n+2}\frac{1}{k+1}-(n+2)!\,(n+2)
\right)\\
+\int_0^1\left(	% Dieses Integral bereits ausgebessert: So stimmt's!!!
	\frac{\po{n+1}{2}\left((n+3)-(n+2)(1-x)^n\right)}{x}+
	\frac{(n+2)\left((1-x)^{n+1}-1\right)}{x^2}
\right){\, dx}
\Biggr).
\end{multline*}
Now, let us turn to the integral. Expanding the integrand by the binomial
theorem and simplifying the result, we obtain
\begin{multline} 
\frac{1-(-1)^{n}(n+2)}{\po{n+1}{3}}\left(
	(-1)^{n+1}-{(n+1)\,(n+2)^2}+
        {(n+1)_3}\sum_{k=1}^{n+2}\frac{1}{k+1}
	\right)\\
\label{eq:ex3}
+\frac{1-(-1)^{n}(n+2)}{\po{n+1}{3}}(n+2)\int_0^1\left(
	\sum_{k=0}^{n-1}\frac{\po{-n-1}{k+2}-(k+2)\po{-n-2}{k+3}}{(k+2)!}x^k
\right){\, dx}.
\end{multline}
Now the integration can be performed without any difficulty. 
We rewrite the resulting sum over $k$ in the second line as a limit,
\begin{multline*} 
\sum_{k=0}^{n-1}\frac{\po{-n-1}{k+2}-(k+2)\po{-n-2}{k+3}}{(k+1)\,(k+2)!}
\\
=\lim _{\ep\to0}\Bigg(\frac {n+1} {\ep}-\frac {n+1} {\ep}
{}_2F_1\!\left[\matrix -n,\ep\\2\endmatrix; 1\right]
+\frac {(n+1)(n+2)} {\ep}-\frac {(n+1)(n+2)} {\ep}
{}_2F_1\!\left[\matrix -n,\ep\\1\endmatrix; 1\right]
\Bigg).
\end{multline*}
Again, Chu--Vandermonde summation \eqref{eq:Chu-Van} can be applied
to evaluate the two $_2F_1$-series. If the result is substituted in
\eqref{eq:ex3}, we get
\begin{multline*}
\frac{1-(-1)^{n}(n+2)}{\po{n+1}{3}}\left(
	(-1)^{n+1}
        - {(n+1)\,(n+2)^2} 
+{(n+1)_3\sum_{k=1}^{n+2}\frac{1}{k+1}}
	\right)\\
+\frac{1-(-1)^{n}(n+2)}{n+3}
	\lim_{\varepsilon\rightarrow0}\left(
	\frac{(n+1)!-\po{2-\varepsilon}{n}}{(n+1)!\,\varepsilon}+
	(n+2)\frac{n!-\po{1-\varepsilon}{n}}{n!\,\varepsilon}
\right).
\end{multline*}
Using de l'Hospital's rule once more, we can compute the limit and obtain after
some simplification the expression
\begin{equation}
\label{eq:simplesumresult}
(1-(-1)^{n}(n+2))
\Biggl(\sum_{k=0}^{n}\frac{2}{k+1}
-	\frac{(-1)^{n}+2n^3+11n^2+19n+11
	}{(n+1)(n+2)(n+3)
	}
\Biggr).
\end{equation}

Now we turn our attention to the double sum in the second line of
\eqref{eq:weird}. Analogously to before,
we replace the term $1/(j+1)$ by $\int_0^1x^j\dx$. This enables
us to evaluate the inner harmonic sum $\sum _{j=0} ^{k}1/(j+1)$ to $\int
_{0} ^{1}(1-x^{k+1})/(1-x)\,dx$. We substitute this in the double sum
in the second line of \eqref{eq:weird}. Using hypergeometric
notation, the result is
\begin{equation*}
(-1)^n(n+2)\int_0^1\left(\frac{1}{1-x}\left(
	\HypF{2}{1}{n+3,-n-1}{2}{1}-
	\frac{1}{x}\HypF{2}{1}{n+3,-n-1}{2}{x}
	\right)+\frac{1}{x}\right){\, dx}.
\end{equation*}
The first hypergeometric series can simply be computed by 
Chu--Vandermonde summation \eqref{eq:Chu-Van}. To the second
hypergeometric series we apply 
the transformation formula (see \cite[(1.8.10),
terminating form]{SlatAC}) 
$$
{} _{2} F _{1} \!\left [ \matrix { a, -N}\\ { c}\endmatrix ; {\displaystyle
   z}\right ]  =
{\frac {    ({ \textstyle c-a}) _{N} } {({ \textstyle c}) _{N} }}
{} _{2} F _{1} \!\left [ \matrix { a,-N}\\ { 1 + a - c -
       N}\endmatrix ; {\displaystyle 1 - z}\right ]  ,
$$
where $N$ is a nonnegative integer.
These operations yield
\begin{equation} \label{eq:ex4}
(-1)^n\int_0^1\left(\frac{(-1)^{n+1}}{(1-x)}
-	\frac{(-1)^{n+1}}{x(1-x)}\HypF{2}{1}{n+3,-n-1}{1}{1-x}
	+\frac{n+2}{x}\right){\, dx}.
\end{equation}
Now we would like to apply the Euler beta integral formula
\eqref{eq:beta-integral} once more. However, this is not possible
just straightforwardly, because the beta integral on the left-hand
side of \eqref{eq:beta-integral} is not defined for $\beta=-1$.
In order to overcome this problem, we first rewrite the term $1/x(1-x)$
(which appears in the second term of the integrand in \eqref{eq:ex4})
as $1/x+1/(1-x)$, and then replace all occurrences of $1/x$ by
$\lim_{\ep\to0^+}x^{1-\ep}$, so that \eqref{eq:ex4} becomes
\begin{multline*}
\int_0^1\lim_{\varepsilon\rightarrow0^+}\Biggl(
\sum_{k=0}^{n+1}\frac{\po{n+3}{k}\,\po{-n-1}{k}}{k!^2}
x^{\varepsilon-1}\,(1-x)^k\\
+\sum_{k=1}^{n+1}\frac{\po{n+3}{k}\,\po{-n-1}{k}}{k!^2}
x^{\varepsilon}\,(1-x)^{k-1}
+(-1)^n(n+2)x^{\varepsilon-1}
\Biggr){\, dx}.
\end{multline*}
Next we interchange limit and integration, and apply the Euler beta
integral formula
\eqref{eq:beta-integral} wherever possible. In the result, the first
sum can easily be evaluated by Chu--Vandermonde summation
\eqref{eq:Chu-Van}. Subsequently, we compute the limit 
by using de l'Ho\-spi\-tal's rule again. This yields the expression
\begin{equation*}
(-1)^n(n+2)\left(\sum_{k=0}^n\frac{1}{k+1}+
	\sum_{k=1}^{n+1}\frac{1}{k+1}\right)+
\sum_{k=1}^{n+1}\frac{\po{n+3}{k}\,\po{-n-1}{k}}{k!^2\, k}.
\end{equation*}
Replacing once more the term ${1}/{k}$ in the
rightmost sum by $\int_0^1x^{k-1}\dx$, we
obtain the expression
\begin{equation*}
(-1)^n(n+2)\left(\sum_{k=0}^n\frac{1}{k+1}+
	\sum_{k=1}^{n+1}\frac{1}{k+1}\right)+
\int_0^1\frac {1} {x}\left(-1+{\HypF{2}{1}{n+3,-n-1}{1}{x}}\right){\, dx}.
\end{equation*}
To the $_2F_1$-series we apply
the transformation formula \eqref{eq:T2106}. In the result, we
replace any occurrence of $x^m$ by $\lim _{\ep\to0^+}x^{m+\ep}$, so that
we arrive at the expression
\begin{multline*}
(-1)^n(n+2)\left(\sum_{k=0}^n\frac{1}{k+1}+
	\sum_{k=1}^{n+1}\frac{1}{k+1}\right)\\
+\int_0^1\lim_{\varepsilon\rightarrow0^+}
	\left(-x^{\varepsilon-1}+(-1)^{n+1}(n+2)
	\sum_{k=0}^{n+1}\frac{\po{-n-1}{k}^2}{k!\,(k+1)!}(x-1)^kx^{n-k+\varepsilon}
\right){\, dx}.
\end{multline*}
Again, we interchange limit and integration and apply the Euler beta
integral formula
\eqref{eq:beta-integral} once more. Writing the result in
hypergeometric notation, we obtain
\begin{multline*}
(-1)^n(n+2)\left(\sum_{k=0}^n\frac{1}{k+1}+
	\sum_{k=1}^{n+1}\frac{1}{k+1}\right)\\
+\lim_{\varepsilon\rightarrow0^+}
\left(
	-\frac{1}{\varepsilon}
        +(-1)^{n+1}\frac{1}{n+2}
	\left(
		1-
		\HypF{2}{1}{-n-2,-n-2}{-n-1-\varepsilon}{1}
	\right)
\right).
\end{multline*}
The hypergeometric series can be computed by 
Chu--Vandermonde summation \eqref{eq:Chu-Van}. Once more we make use
of de l'Hospital's rule for the limit and obtain after some simplifications
the expression
\begin{equation}
\label{eq:doublesumresult}
-\frac{1+(-1)^n\left(n^2+3n+3\right)}{n+2}-
(1-(-1)^n(n+2))\sum_{k=0}^{n}\frac{2}{k+1}.
\end{equation}

When adding together \eqref{eq:simplesumresult}, the result of our
computation for the sum in the first line of \eqref{eq:weird}, and
\eqref{eq:doublesumresult}, the result of our computation for the
double sum in the second line of \eqref{eq:weird},
the harmonic sums cancel, and it is easy to verify that, magically, 
the remaining
terms simplify to the right-hand side of \eqref{eq:weird}.
\end{proof}

\begin{rem}
Peter Paule demonstrated to us, that the identity \eqref{eq:weird}
can also be proved algorithmically. Clearly, the 
Gosper--Zeilberger algorithm \cite{PeWZAA,ZeilAM,ZeilAV} finds a
recurrence for the sum in the first line of \eqref{eq:weird}. Carsten
Schneider's extension of Karr's algorithm \cite{KarrAA}, implemented
by Schneider, finds a recurrence for the double sum in the second
line of \eqref{eq:weird}. Finally, Mallinger's {\sl Mathematica}
package {\tt GeneratingFunctions} \cite{MallAA} 
or Salvy and Zimmermann's {\sl
Maple} package {\tt gfun} \cite{SaZiAA} can be used to combine these two
recurrences into one, a recurrence of order 10. It is then routine to
check (preferably on the computer) that the right-hand side of
\eqref{eq:weird} satisfies
this same recurrence. However, in the present implementation, these algorithms
are not able to find the explicit evaluations, in terms of harmonic
numbers, of the sums in the first and second line of
\eqref{eq:weird}, given in \eqref{eq:simplesumresult} and
\eqref{eq:doublesumresult}, respectively.
\end{rem}

% Die Determinante, um die es geht:
% ATM[m_,n_,X_]:=ATM[m,n,X]=Sum[s^m,{s,-n,n-X-1}]
% ATM[k_,X_]:=ATM[k,X]=Table[ATM[i+j,n,X],{i,1,k},{j,1,k}]
% DATM[k_,X_]:=DATM[k,X]=Factor[Det[ATM[k,X]]]
%
% In der Notation (2.1) unseres Artikels:
% n=a,n-X=b,k=N
%
% TSF[k_,X_]:=TSF[k,X]=
% 	(n-X/2)^k*
% 	Product[
% 		((n-e)*(n-X+e))^(k-2*e+X),
% 		{e,(X+1)/2,(k-1+X)/2,1/2}
% 	]*
% 	Product[
% 		(n-e)*(n-X+e),
% 		{e,X+1,Floor[(k+X)/2]}
% 	]
% TCF[k_,X_]:=TCF[k,X]=Factor[DATM[k,X]/TSF[k,X]]
% DegRP[k_,X_]:=2*If[k<=X+1,k,Ceiling[(k+X)/2]]
% NofEval[k_,X_]:=
% 	2*(
% 		Floor[(k-If[EvenQ[X],1,0])/2] +
% 		If[Floor[(k +X)/2] > Max[X, (-1 + k +X)/2],1, 0]
% 	)
% 	+If[EvenQ[X],1,0]


%%%@@@ end input from lemma2


%%%@@@ input from problems
\section{Evaluations of Hankel determinants featuring Bernoulli
polynomials}
\label{sec:further}

There are several theorems hidden in the body of this paper. 
Among these are evaluations of Hankel determinants of
{\em Bernoulli polynomials} evaluated at special values. 
Recall that the $l$-th Bernoulli polynomial is defined by
\begin{equation*} 
B_l(x):=\sum_{k=0}^l\binom{l}{k}B_{l-k}x^k.
\end{equation*}
The Hankel determinants of Bernoulli polynomials of which we are talking
are special cases of the determinant $B(N;x)$ given by
\begin{equation} \label{eq:BPol}
B(N;x):=\det_{1\leq i,j\leq N}\left(B_{i+j}(x)\right).
\end{equation}
This is in a fundamental way different from the Hankel determinant
$$\det_{0\le i,j\le N}(B_{i+j}(x)),$$
which has been considered earlier (see \cite[Sec.~5]{AlCaAA}). 
(Note that the
difference is that, in the latter determinant, indices start
already with 0.) As is not difficult to see (cf\@.
\cite[p.~419]{MuirAD} or \cite[Lemma~15]{KratBN}), the latter 
determinant does in fact not depend on $x$
(i.e., the powers of $x$ cancel in the expansion of the
determinant), so that its value is equal to its value at $x=0$,
which, in turn,
is given by Theorem~\ref{thm:HankelBernoulli} with $a=b=1$, $c=d=0$.
This is in sharp contrast to the Hankel determinant \eqref{eq:BPol},
where the powers of $x$
do not cancel, so that \eqref{eq:BPol} is a nontrivial
polynomial in $x$. As such, the evaluation of the determinant
\eqref{eq:BPol} is much more difficult. Below, we provide evaluations
of \eqref{eq:BPol} for $x=-1$, $x=-1/2$ and $x=1/2$. Needless to say that 
the evaluation in the special case $x=0$ (and as well in the special
case $x=1$) is given by 
Theorem~\ref{thm:HankelBernoulli} with $a=b=c=d=1$.


First of all, in the proof of Lemma~\ref{lem:Det3}, we observed that
(in symbolic notation $B^k\equiv B_k$)
$$\det_{1\le i,j\le N}\big((B-1)^{i+j}\big)=
\det_{1\le i,j\le N}\big(B^{i+j-2}(B-1)^2\big).
$$
The determinant on the right-hand side was then evaluated in
Lemma~\ref{lem:extra-Bernoulli}. The linear combination $(B-1)^l$
of Bernoulli numbers is nothing else but $B_l(-1)$, the $l$-th
Bernoulli polynomial evaluated at $-1$. Thus, we obtain the following
corollary.

\begin{cor}
Let $N$ be a positive integer. Then there holds
\begin{multline}
B(N;-1)=
\det_{1\le i,j\le N}\left(B_{i+j}+(-1)^{i+j}(i+j)\right)
\\=
{(-1)^{{N(N-1)}/{2}}} {6^{-N}}(1 + 2N + 5N^2 + 4N^3 + N^4)
\prod_{i=1}^N\left(\frac{i(i+1)^4(i+2)}{(2i+1)(2i+2)^2(2i+3)}\right)^{N-i}.
\end{multline}
\end{cor}

Our next evaluation results from the determinant evaluation in
Lemma~\ref{lem:Det2}.

\begin{thm}
\label{prop:extrad-det}
Let $m$ be a positive integer. Then for
\begin{equation*}
B(N;-1/2)=
\det_{1\le i,j\le N}\left((2^{1-i-j}-1)B_{i+j}-(-1/2)^{i+j-1}(i+j)\right)
\end{equation*}
there hold
\begin{multline} \label{eq:B1}
B(2m-1;-1/2)=\frac{
	\left( -1 \right)^{m - 1}\,\left(2 m - 1\right)!^2
	\left( \prod_{i = 1}^{2 m - 1}i! \right)^4
	\left( \prod_{i = 1}^{m}\left(2 i - 1\right) ! \right)^4
}{
	2^{6 \left(m-1\right)}\,\left(m - 1\right)!^6
	\left( \prod_{i = 1}^{4 m - 1}i! \right)
}\\\times
\left( 3 + 8 m + 
	\frac{
		8 \left(m - 1\right)  \left(2 m + 1\right)
	}{
		3 m \left(m + 1\right)  \left(2 m - 3\right)  
      \left(2 m - 1\right)
	}
   \HypF{4}{3}{
		3,\frac{3}{2},2-m,\frac{3}{2}+m
	}{
		\frac{5}{2},2+m,\frac{5}{2}-m
	}{
		1
	}
\right)
\end{multline}
and
\begin{multline} \label{eq:B2}
B(2m;-1/2)=\frac{
	\left( -1 \right)^{m}
	\left( \prod_{i = 1}^{2 m}i! \right)^4
	\left( \prod_{i = 1}^{m + 1}\left(2 i - 1\right) ! \right)^4
}{
	2^{6 m}\,m!^6\,\left( \prod_{i = 1}^{4 m + 1}i! \right)
}\\\times
\left(
	1 + 8 m - 
	\frac{
		8 m \left(2 m + 3\right)
	}{
		3 \left(m + 1\right)  \left(m + 2\right)  
      \left(2 m - 1\right)  \left(2 m + 1\right)
	} 
	\HypF{4}{3}{
		{3,\frac{3}{2}},1- m,{\frac{5}{2}+m}
	}{
		{\frac{5}{2}},3+m,{\frac{3}{2}-m} 
	}{
		1
	}
\right).
\end{multline}
\end{thm}
\begin{proof} 
Consider the determinant $D(n,n-1,N)$ (see
\eqref{eq:gen-det} for definition). Factor
$2(n-1/2)$ out of each column of $D(n,n-1,N)$, and then
set $n=1/2$. By the appropriate variant of \eqref{eq:BernoulliP} and
de l'Hospital's rule,
this yields the Hankel determinant $B(N;-1/2)$. 
On the other
hand, we have evaluated $D(n,n-1,N)$ in Lemma~\ref{lem:Det2}.
Then by dividing the results in \eqref{eq:Det2odd} and
\eqref{eq:Det2even} by $2^N(n-1/2)^N$ and then setting $n=1/2$,
we obtain the expressions on the right-hand sides of \eqref{eq:B1}
and \eqref{eq:B2}.
\end{proof}

When solving the enumeration of rhombus tilings of a hexagon which
contain the central rhombus, the determinant $D(n,n+1,N)$ 
was explicitly \cite{HeGeAA} (see the proof of Lemma~10 and
Proposition~14 in \cite{HeGeAA}) or
implicitly \cite{CiucKratAB,FuKrAC} evaluated (compare
Proposition~\ref{thm:HeGe-det}). If one adapts the
preceding proof to this case, one obtains the following result.

\begin{thm}
\label{prop:extrad-det2}
Let $m$ be a positive integer. Then for
\begin{equation*}
B(N;1/2)=
\det_{1\le i,j\le N}\left((2^{1-i-j}-1)B_{i+j}\right)
\end{equation*}
there hold
\begin{multline} \label{eq:B3}
B(2m-1;1/2)=
{\frac {{{\left( 2 m \right) !}^2} 
   }
   {{2^{4 m-2}}\,{{\left( m-1 \right) !}^2}\,{{m!}^2}}}
     \bigg( \prod_{i = 1}^{2 m}
        {\frac {{{\left( i-1 \right) !}^5}}
          {\left( 2m+i-1 \right) !}} \bigg)
 \sum_{i = 0}^{m-1}
     \frac {{{\left( -1 \right) }^{m-i}}} {\left( 2m- 2 i-1 \right)}
        {\frac {  ({ \textstyle {\frac 1 2} - i}) _{2 i} }
          {  {{i!}^2}}} 
\end{multline}
and
\begin{equation} \label{eq:B4}
B(2m;1/2)=
{\frac {{{\left( 2m + 2  \right) !}^2} 
   }
   {{2^{4 m+2}}\,{{m!}^2}\,{{\left( m+1 \right) !}^2}}}
     \bigg( \prod_{i = 1}^{2 m+1}
        {\frac {{{\left( i-1 \right) !}^5}} {\left( 2 m+i \right) !}}
        \bigg)
 \sum_{i = 0}^{m}
     \frac {{{\left( -1 \right) }^{m-i}}} {\left( 2 m-2i+1 \right)}
        {\frac {  ({ \textstyle {\frac 1 2} - i}) _{2 i} }
          {  {{i!}^2}}} .
\end{equation}
\end{thm}
\begin{proof} 
Consider the determinant $D(n,n+1,N)$. Factor
$2(n+1/2)$ out of each column of $D(n,n+1,N)$, and then
set $n=-1/2$. By the appropriate variant of \eqref{eq:BernoulliP} and
de l'Hospital's rule,
this yields the Hankel determinant $B(N;1/2)$. 
On the other
hand, by Proposition~\ref{thm:HeGe-det} (with $N$ replaced by $N+1$,
$l=n$, and $M=2n-N$) we know that $D(n,n+1,N)$ counts, up to a
multiplicative constant, the number of rhombus tilings of a hexagon
with sides $N+1,2n-N,N+1,N+1,2n-N,N+1$, which contain the central
rhombus. This enumeration problem was solved in 
\cite{CiucKratAB,FuKrAC,HeGeAA}. If in the result we perform
the according manipulations and then set $n=-1/2$,
we obtain the expressions on the right-hand sides of \eqref{eq:B3}
and \eqref{eq:B4}.
\end{proof} 

\section{Concluding comments and open problems}
\label{sec:open}

We conclude this article by pointing to open questions
which are raised by this work.

\smallskip
(1)
In Corollary~\ref{thm:Propp} it was demonstrated that, for $M$ close
to $N$, the number of 
rhombus tilings of a hexagon with side lengths $N,M,N,N,M,N$,
which contain the rhombus above and next to the center of the hexagon, equals
$(\frac {1} {3}+r(N,M))\,T(N,M)$, where $T(N,M)$ is the total number of
rhombus tilings of the hexagon, and where $r(N,M)$ is a ``closed form"
expression. (Magically,
the value of $1/3$ which appears here is, 
according to \cite[Corollary~3]{CiucKratAB}, the {\em exact\/} 
proportion of the rhombus tilings that contain the
{\em central\/} rhombus in the total number of rhombus tilings of a
hexagon with side lengths $2n-1,2n,2n-1,2n-1,2n,2n-1$ or with side
lengths $2n,2n-1,2n,2n,2n-1,2n$.) 
As we mentioned in the Introduction, it is easy to derive
many more such results, also for the central rhombus and the other
two cases that were considered in
Theorems~\ref{thm:NEven}--\ref{thm:NOdd2}.
Our proof, given in Section~2, consisted of starting with
the expressions \eqref{eq:MEven} and \eqref{eq:MOdd} and applying
Zeilberger's algorithm to establish the simplification of 
the sum in these expressions when
$m$ and $n$ are close. This is, unfortunately, not conceptual (as it
just {\em verifies}, but does not {\em derive} the result), and
therefore does not explain why these simplifications take place.
The fact that apparently many more such results exist indicates
that there must be a hypergeometric transformation formula lurking in
the background, which we were, however, unable to discover. (It is obvious
that the sums in \eqref{eq:MEven}--\eqref{eq:NOdd2} can be written as
very-well-poised $_7F_6$-series --- see e.g\@.
\eqref{eq:very-well-poised} --- and, by means of Whipple's
transformation formula \eqref{eq:Whipple}, can therefore be transformed
into balanced $_4F_3$-series, to which, in turn, we could apply
Sears' $_4F_3$ transformation formulas. However, it seems that this
does not suffice to find the desired identity which would ``explain"
Corollary~\ref{thm:Propp}.)


\smallskip
(2)
Is it possible to find a uniform formula for the number of
rhombus tilings of a hexagon with side lengths $N,M,N,N,M,N$, which
contain an {\em arbitrary} (but fixed) rhombus on the ``vertical"
symmetry axis (i.e., the symmetry axis
which runs in parallel to the sides of length $M$)?
Recall that (as we mentioned already in the Introduction) 
in \cite{FuKrAC} such a formula was found for the ``horizontal" symmetry
axis (i.e., the symmetry axis which cuts through the sides of length
$M$). In contrast, here we encountered increasing difficulties in the
proofs of our enumerations the farther we moved the rhombus which is
contained in every tiling from the center. Recall that for solving
our enumeration problems we needed to compute the determinants
$D(n,n-t,N)$ (see \eqref{eq:gen-det} for definition) for $t=0,1,2$.
For the case of a rhombus which is even farther away from the center, we
would have to evaluate this determinant for even larger values of $t$. The
increasing difficulties in doing this arise in Step~5 (compare the
proof of Lemma~\ref{lem:Det1}) of the computation. The previous
steps, Steps~1--4, would prove that
$$D(n,n-t,N)=\big(\text {product of linear factors in $n$}\big)\cdot
S(n,N,t),$$
where $S(n,N,t)$ is a polynomial of degree $2\ceil{(N+t)/2}$ (compare
\eqref{eq:polydef}, \eqref{eq:polydef2}, and \eqref{eq:polydef3}).
Thus, in order to determine $S(n,N,t)$, the larger $t$ becomes, the
more evaluations of $S(n,N,t)$ at special values of $n$ 
(or other informations about $S(n,N,t)$) we need. (The computations in
\cite{FuKrAC} have exactly the opposite behaviour: The farther the
rhombus which is contained in every tiling is moved away from the
center, the {\em smaller} in degree becomes the irreducible
polynomial in the result.) Even worse, the larger $t$ becomes,
the more difficult it becomes to obtain these special values.
(Remember, for example, the difficulty of evaluation of $R(n,N)$ at
$n=1$ via Lemmas~\ref{lem:extra-Bernoulli} and \ref{lem:weird}.)

That the problem that we considered here is at a different level of
complexity than the problem in \cite{FuKrAC} is also indicated by the
(partially conjectural) form of the asymptotic behaviour of the
proportion of the rhombus tilings that contain this particular
rhombus in the total number of rhombus tilings. While the asymptotic
behaviour is totally smooth when the rhombus which is contained in
every tiling is moved along the ``horizontal" symmetry axis (see
\cite[Theorem~1.3]{FuKrAC}), the conjectured form
\cite[Conjecture~1]{CoLPAA} of the asymptotics when
the rhombus which is contained in
every tiling is moved along the ``vertical" symmetry axis behaves
nonsmoothly. It is increasing for some time when the rhombus is moved
away from the center, but at some point, when the rhombus enters the ``arctic
region" near the (top or bottom) corner, it becomes 1 and stays
1 from thereon. Thus, a formula for exact enumeration must,
somehow, reflect this nonsmooth asymptotic behaviour.

Is there a way to overcome these difficulties?

\smallskip
(3)
In Theorem~\ref{thm:HankelBernoulli} only $c$ or $d$ may be 0, but
not $a$ or $b$. In fact, Theorem~\ref{thm:HankelBernoulli} is wrong
if $a=0$ or $b=0$. But, apparently, not terribly wrong. 
Lemma~\ref{lem:extra-Bernoulli}
shows the evaluation of the determinant in \eqref{eq:HankelBernoulli}
with $a=b=0$, $c=d=2$. Remarkably, the result is almost identical
with the right-hand side in \eqref{eq:HankelBernoulli}, the only
difference being the polynomial in $n$ of fourth degree in
\eqref{eq:extra-Bernoulli}. In fact, computer experiments suggest
that a much more general result holds.

\begin{conj}
For positive integers integers $c,d$ there holds
\begin{multline} \label{eq:HankelBernoulli-extra1}
\det_{1\le i,j\le n}
\left(
	B^{i+j-2}\,\po{-B+1}{c-1}\,\po{-B+1}{d-1}
\right)\\
=(-1)^{n(n-1)/2}\left({\frac{\left( c -1\right) !\,\left( c - 1\right) !\,
     \left( d - 1\right) !\,\left( d - 1\right) !}{\left(
        c + d - 1\right) !}}\right)^n\kern6cm\\
\times\prod_{i=1}^{n-1}\Biggl({\frac{i\,\left( c + i - 1\right) \,
     \left( c + i - 1\right) \,\left( d + i - 1\right)\,
\left( d + i - 1\right) \,
     \left( c + d + i - 2\right) }
	  {\left( c + d + 
       2 i - 3\right) \,{{\left( c + d + 2 i - 2\right) }^2\,
     \left( c + d + 2 i - 1\right) }}}
\Biggr)^{n-i} P(n;c,d),
\end{multline}
where $P(n;c,d)$ is a certain polynomial in $n$ of degree
$2(c+d-2)$.

Furthermore, for positive integers integers $b,c,d$ there holds
\begin{multline} \label{eq:HankelBernoulli-extra2}
\det_{1\le i,j\le n}
\left(
	B^{i+j-1}\,\po{B+1}{b-1}\,\po{-B+1}{c-1}\,\po{-B+1}{d-1}
\right)\\
=(-1)^{n(n-1)/2}\left({\frac{\left( c -1\right) !\,\left(b + c - 1\right) !\,
     \left(d - 1\right) !\,\left(b + d - 1\right) !}{\left(
        b + c + d - 1\right) !}}\right)^n\\
\times\prod_{i=1}^{n-1}\Biggl({\frac{i\,\left( c + i - 1\right) \,
     \left(b + c + i - 1\right) \,\left( d + i - 1\right)}
	  {\left( b + c + d + 
       2 i - 3\right) \,{{\left( b + c + d + 2 i - 2\right) }^2}}}\\
\times{\frac{\left(b + d + i - 1\right) \,
     \left( b + c + d + i - 2\right) }{
     \left( b + c + d + 2 i - 1\right) }}
\Biggr)^{n-i} R(n;b,c,d),
\end{multline}
where $R(n;b,c,d)$ is a certain rational function in $n$, which can be
written with a
numerator of degree $c+d-2$ and a denominator of degree $b-1$.
\end{conj}

In principle, our approach of proving Lemma~\ref{lem:extra-Bernoulli}
(the special case $c=d=2$ of \eqref{eq:HankelBernoulli-extra1}),
which consisted of using linearity of the determinant in order to
break it into several pieces, to each of which we could either 
apply Theorem~\ref{thm:HankelBernoulli} or Theorem~\ref{thm:Leclerc},
should make a proof of the above conjecture possible. However,
serious difficulties have to be expected in actually doing the
calculations, in particular, when working through a generalized form
of Lemma~\ref{lem:weird}. We believe that, in view of the simplicity
of the result \eqref{eq:extra-Bernoulli} and of the conjectured
results \eqref{eq:HankelBernoulli-extra1} and
\eqref{eq:HankelBernoulli-extra2}, there must be a more elegant way
to attack these Hankel determinant evaluations, in particular, if one
also desires to obtain explicit forms for the polynomial $P(n;c,d)$
and the rational function $R(n;b,c,d)$.

%%%@@@ end input from problems

\bigskip

{\small
\noindent
{\sc Note.} \scriptsize
Since first versions of this article were distributed,
Ilse Fischer (``Enumeration of rhombus tilings of a hexagon which
contain a fixed rhombus in the centre", preprint,
{\tt math/9906102}) generalized
Theorems~\ref{thm:MEven} and \ref{thm:MOdd} to arbitrary semiregular
hexagons.
}

%%%@@@ input from bib
\ifx\undefined\bysame
\newcommand{\bysame}{\leavevmode\hbox to3em{\hrulefill}\,}
\fi







%%%@@@ end input from bib

\end{document}


